

No. **S1910194** Vancouver Registry

IN THE SUPREME COURT OF BRITISH COLUMBIA

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND

IN THE MATTER OF THE BUSINESS CORPORATIONS ACT, S.B.C. 2002, C. 57

AND

IN THE MATTER OF ENERGOLD DRILLING CORP., CROS-MAN DIRECT UNDERGROUND LTD., EGD SERVICES LTD., BERTRAM DRILLING CORP. AND OMNITERRA INTERNATIONAL DRILLING INC.

PETITIONERS

ORDER MADE AFTER APPLICATION (SANCTION ORDER)

BEFORE

) THE HONOURABLE)[™].JUSTICE MILMAN

January 17, 2020

ON THE APPLICATION of Extract Advisors LLC, in its capacity as administrative agent for noteholders (the "Agent") coming on for hearing at Vancouver, British Columbia on January 17, 2020 and on hearing Christopher J. Ramsay and Katie G. Mak and those other counsel listed on Schedule "A" hereto; and upon reading the material filed, and pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. c-36, as amended (the "CCAA"), the British Columbia Supreme Court Rules and the inherent jurisdiction of this Honourable Court;

THIS COURT ORDERS AND DECLARES THAT:

SERVICE

1. The time for service of the Notice of Application and supporting materials be and is hereby abridged such that this Application is properly returnable today and service thereof upon any interested party other than the parties on the service list maintained by the Monitor in this proceeding is hereby dispensed with.

DEFINITIONS

2. All capitalized terms not otherwise defined in this Order (the **"Sanction Order**") shall have the meanings given to them in the Amended Plan of Compromise and Arrangement dated January 13, 2020 (the **"Plan**") considered at the Creditors Meeting held in Vancouver, British Columbia on January 13, 2020, a copy of which is attached hereto as **Schedule "B"**.

THE CREDITORS' MEETING

3. There has been good and sufficient service, delivery and notice to all Affected Noteholders of the Information Package, and that the Creditors' Meeting was duly, called, convened, held and conducted all in conformity with the CCAA and all other Orders of this Court in the CCAA Proceedings (collectively, the "CCAA Orders").

SANCTION OF THE PLAN

4. Pursuant to the Meeting Order, the only class of creditors for the purposes of voting to approve the Plan is the Affected Noteholder Class.

5. The Plan has been voted on, agreed to and approved by the Required Majority, all in conformity with the CCAA and the terms of the Meeting Order.

6. The Petitioners and the Agent have complied with the provisions of the CCAA and adherence of the CCAA Orders.

7. The Petitioners and the Agent have not done or purported to do anything that is not authorized by the CCAA.

8. The Plan, together with the compromises, arrangements, transactions, releases, discharges, injunctions and results provided therein and effected thereby, is fair and reasonable.

9. The Plan is hereby sanctioned and approved pursuant to section 6 of the CCAA.

10. The Petitioners, the Agent and the US LP shall rely on this Order and section 3(a)(10) of the United States *Securities Act of 1933* (the "**U.S. Securities Act**") as a basis for an exemption from the registration requirement as set out in the U.S. Securities Act for the distributions of securities contemplated in connection with the Plan.

PLAN IMPLEMENTATION

11. The Plan and all associated steps, compromises, transactions, arrangements, releases and reorganizations effected thereby are hereby approved, shall be deemed to be implemented and shall be binding and effective as of the Effective Time in accordance with the terms of the Plan or at such other time, times or manner as may be set forth in the Plan in the sequence provided therein, and shall enure to the benefit of and be binding and effective upon all Persons and parties named or referred to in, affected by or subject to the Plan as provided for in the Plan or this Order, including for certainty the Agent, the Petitioners, the Affected Noteholders and the Third Party Released Parties.

12. Each of the Agent, the Petitioners and the Monitor is authorized and directed to take all steps and actions and to do all things, necessary or appropriate, to implement the Plan in accordance with its terms and to enter into, execute, deliver, complete, implement and consummate all of the steps, transactions, distributions, disbursements, payments, deliveries, allocations, instruments and agreements contemplated by, and subject to the terms and conditions of, the Plan, and all such steps and actions are hereby authorized, ratified and approved. None of the Agent, the Petitioners or the Monitor shall incur any liability as a result of acting in accordance with the terms of the Plan and this Order, other than any liability arising out of or in connection with the gross negligence or wilful misconduct of such parties.

13. Upon delivery of written notice from the Petitioners and the Agent (or counsel on their behalf) to the Monitor that the conditions precedent as set out in the Plan have been satisfied or waived, as applicable, in accordance with the terms of the Plan, the Monitor shall as soon as reasonably practicable following receipt of such written notice, deliver to the Petitioners and the Agent (or counsel on their behalf) a certificate signed by the Monitor substantially in the form attached hereto as **Schedule "C"** (the "**Monitor's Certificate**") certifying that all conditions precedent set out in the Plan have been satisfied or waived and that the Implementation Date has occurred and that the

Plan and the provisions of this Sanction Order which come into effect on the Implementation Date are effective in accordance with their respective terms. Following the delivery of the Monitor's Certificate to the Petitioners and the Agent, the Monitor shall file the Monitor's Certificate with the Court, and shall post a copy of same, once filed, on the Monitor's website and provide a copy to the Service List. Upon delivery of the Monitor's Certificate to the Petitioners and the Agent, all applicable parties shall take such steps as are required to implement the steps set out in Article 8 of the Plan.

COMPROMISE OF AFFECTED NOTEHOLDER CLAIMS, EFFECT OF PLAN AND RELEASES

14. The compromises, arrangements, releases, discharges and injunctions contemplated in Article 9 of the Plan are integral components thereof and that, effective on the Implementation Date, all such compromises, releases, discharges and injunctions contemplated in the Plan are effective, sanctioned, approved and given full force and effect.

15. On the Implementation Date, in accordance with Article 8 of the Plan, all Affected Noteholder Claims, all Intercompany Working Capital Obligations and all Released Claims shall be fully, finally, irrevocably and forever released, discharged, cancelled and barred and Energold shall thereupon have no further obligation whatsoever in respect of the Affected Noteholder Claims and the Released Claims, as applicable; provided that nothing herein releases Energold or any other Person from their obligations to make distributions in the manner and to the extent provided for in the Plan and provided further that such discharge and release of Energold shall be without prejudice to the right of an Affected Noteholder in respect of an Unaffected Noteholder Claim.

At the Effective Time, Energold shall be released and discharged from any and 16. all demands, claims, liabilities, causes of action, debts, accounts, covenants, damages, executions and other recoveries based in whole or in part on any act or omission, transaction, dealing or other occurrence existing or taking place on or prior to the Implementation Date, applications, counterclaims, suits, sums of money, judgments, orders, including for injunctive relief or specific performance and compliance orders, expenses, encumbrances and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any Affected Noteholder, to the extent of their Affected Noteholder Claim, may be entitled to assert, whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Effective Time arising out of or in connection with the Notes or the Plan, and all such Claims shall be forever waived and released (other than the right to enforce Energold's obligations under the Plan or any related document), all to the full extent permitted by Applicable Law, save and except any gross negligence or wilful misconduct, provided that nothing herein shall release or discharge any of the Energold Group from any Excluded Claims.

At the Effective Time, the Petitioners' legal counsel, the Financial Advisor, the 17. Monitor, the Agent and their respective subsidiaries and affiliates and each of their respective shareholders, partners, officers, directors, current and former employees, financial advisors, legal counsel and agents (being referred to individually as a "Third Party Released Party") are hereby released and discharged from any and all demands. claims, liabilities, causes of action, debts, accounts, covenants, damages, executions and other recoveries based in whole or in part on any act or omission, transaction, dealing or other occurrence existing or taking place on or prior to the Implementation Date, actions, applications, counterclaims, suits, sums of money, judgments, orders, including for injunctive relief or specific performance and compliance orders, expenses, encumbrances and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any Person may be entitled to assert, whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Implementation Date arising out of or in connection with the Affected Noteholder Claims or the Plan, and any Claims, and all claims arising out of such actions or omissions shall be forever waived and released (other than the right to enforce the Monitor's or the Agent's obligations under the Plan or any related document), all to the full extent permitted by Applicable Law save and except any gross negligence or wilful misconduct on the part of the Third Party Released Party, provided that nothing herein shall release or discharge any Third Party Released Party if such Third Party Released Party is judged by the express terms of a judgment rendered on a final determination on the merits to have committed criminal, fraudulent or other wilful misconduct.

- 18. Notwithstanding:
 - (a) the pendency of the CCAA Proceedings;
 - (b) any applications for a bankruptcy order in respect of the Petitioners now or hereafter made pursuant to the *Bankruptcy and Insolvency Act* and any bankruptcy order issued pursuant to any such applications; and
 - (c) any assignment in bankruptcy made by or in respect of the Petitioners,

the transactions, distributions, steps and releases or compromises made during the CCAA Proceedings or contemplated to be performed or effected pursuant to the Plan and this Sanction Order shall be binding on any trustee in bankruptcy that may be

appointed in respect of the Petitioners and shall not be void or voidable by creditors of the Petitioners, nor shall it constitute or be deemed to be a transfer at undervalue, fraudulent preference, assignment, fraudulent conveyance or other reviewable transaction under the *Bankruptcy and Insolvency Act* or any other applicable federal or provincial legislation, nor shall it constitute oppressive or unfairly prejudicial conduct pursuant to any applicable federal or provincial legislation.

19. Except to the extent expressly contemplated by the Plan or this Sanction Order, all obligations or agreements to which the Petitioners are a party to immediately prior to the Effective Time, will be and shall remain in full force and effect as at the Implementation Date, unamended except as they may have been amended by agreement of the parties to such agreement, and no Person who is a party to any such obligation or agreement shall, following the Implementation Date, accelerate, terminate, rescind, refuse to perform or otherwise repudiate its obligations thereunder, or enforce or exercise any right (including any right of set-off, option, dilution or other remedy) or make any demand under or in respect of any such obligation or agreement, by reason of (i) any defaults or events of default arising as a result of the insolvency of the Petitioners prior to the Implementation Date; (ii) any change of control of the Petitioners arising from the implementation of the Plan; (iii) the fact that the Petitioners have sought or obtained relief under the CCAA or that the Plan has been implemented; (iv) the effect on the Petitioners of the completion of any of the transactions contemplated by the Plan; (v) any compromises, arrangements, or reorganization effected pursuant to the Plan; or (vi) any other event(s) which occurred on or prior to the Implementation Date which would have entitled any Person to enforce rights and remedies subject to any express provisions to the contrary in any agreements entered into with the Petitioners after the Filing Date.

20. From and after the Implementation Date, all Persons shall be deemed to have waived any and all defaults of the Petitioners then existing or previously committed by the Petitioners, or caused by the Petitioners, directly or indirectly, or non-compliance with any covenant, warranty, representation, undertaking, positive or negative pledge, term, provision, condition or obligation, expressed or implied, in any contract, instrument, credit document, lease, guarantee, agreement for sale, deed, licence, permit or other agreement, written or oral, and any and all amendments or supplements thereto, existing between such Person and any of the Petitioners arising directly or indirectly from the filing by the Petitioners under the CCAA and the implementation of the Plan, and any and all notices of default and demands for payment or any step or proceeding taken or commenced in connection therewith under any such agreement shall be deemed to have been rescinded and of no further force or effect, provided that nothing herein shall be deemed to excuse the Petitioners from performing their

obligations under the Plan or be a waiver of defaults by the Petitioners under the Plan and the related documents.

RECORD OF US LP UNITS

On the Implementation Date or as soon as reasonably practicable thereafter, in 21. accordance with the Plan, the US LP shall lodge in its records the US LP Units to be distributed to the Affected Noteholders and an Affected Noteholder shall only be entitled to the rights associated with the US LP Units upon the delivery by the Affected Noteholder of its US LP Unitholder Information to the Agent or the US LP in accordance with the Plan. In the event that an Affected Noteholder fails to deliver its US LP Unitholder Information in accordance with the Plan on or before the date that is six months following the Implementation Date, the US LP Units otherwise issuable to such Affected Noteholder pursuant to the Plan shall not be reserved on the books and records of the US LP and the US LP shall have no further obligation to issue or deliver, and shall have no further obligation to reserve on its books and records, any LP Units otherwise issuable to such Affected Noteholder and the Affected Noteholder shall cease to have a claim to, or interest of any kind or nature against or in, Energold, the US LP or the Unissued US LP Units and the Agent shall be entitled to delete such Unissued US LP Units from the books and records of the US LP as maintained by the US LP.

POST-IMPLEMENTATION DATE TRANSACTIONS

22. After the Implementation Date, each of the Agent, the Petitioners or the Monitor shall, as applicable, take all steps necessary to implement the Plan, including:

- (a) calculating and paying any amounts required to be paid in accordance with section 8.3 of the Plan;
- (b) implementing the transaction steps in accordance with section 8.2 of the Plan; and
- (c) effecting the transfer of the Purchased Assets to the Agent, Canadian Holdcos and US LP, as applicable, pursuant to the Vesting Orders.

23. In addition to its prescribed rights and obligations under the CCAA Orders, the Monitor is granted the powers, duties and protections contemplated by and required under the Plan and that the Monitor be and is hereby authorized, entitled and empowered to perform its duties and fulfil its obligations under the Plan to facilitate the implementation thereof and to apply to this Court for any orders necessary or advisable to carry out its powers and obligations under any other CCAA Order.

24. The Petitioners shall be and are hereby directed to maintain the books and records of the Petitioners for purposes of implementing the Plan.

25. Matthew Freeman and Claudia Jordan are hereby directed and authorized to, on behalf of the Petitioners, execute any documents and give such instructions as may be necessary or desirable to fulfil the Petitioners' obligations under the Plan and facilitate the implementation thereof, which direction and authority shall survive and continue after the CCAA Proceedings have come to an end.

APPROVAL OF MONITOR'S FOURTH REPORT

26. The Fourth Report and the conduct and activities of the Monitor described therein be and are hereby approved.

GENERAL

27. Nothing in this Order nor in the Plan shall affect, in any way, any of the Residual Assets or any claims, security interests, encumbrances, or interests against same, until such time as a corresponding Vesting Order (the **"Residual Asset Vesting Order"**) concerning the Residual Assets has been granted and all rights and remedies of such creditors and stakeholders are hereby reserved until such time. Any provisions of this Order and the Plan which, in any way, affect the Residual Assets or any rights, security interests, claims, and interests therein, shall remain subject to the terms of the Residual Asset Vesting Order.

28. The Petitioners, Agent or the Monitor on behalf of the Petitioners, as the case may be, shall be authorized, in connection with the taking of any step or transaction or performance of any function under or in connection with the Plan, to apply to any Governmental Entity for any consent, authorization, certificate or approval in connection therewith.

29. The Agent, the Petitioners and the Monitor and any other interested parties are hereby granted leave to apply to this Court for such further advice, directions or assistance as may be necessary to give effect to the terms of the Plan and any other matters that pertain to the completion of the administration of the CCAA Proceedings.

30. Endorsement of this Order by counsel appearing on this application, other than counsel for the Agent, is hereby dispensed with.

THIS COURT HEREBY REQUESTS:

31. The aid and recognition of any court or any judicial, regulatory or administrative body having jurisdiction in Canada, the United States, or in any other foreign

jurisdiction, to recognize and give effect to the Plan and this Order, to confirm the Plan and this Order as binding and effective in any appropriate foreign jurisdiction, and to assist the Agent, the Petitioners, the Monitor and their respective agents in carrying out the terms of the Plan and this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Agent, the Petitioners, and the Monitor, as an officer of this Court, as may be necessary or desirable to give effect to this Order, to grant representative status to the Agent, the Petitioners and the Monitor in any foreign proceeding, or to assist the Agent, the Petitioners and the Monitor and their respective agents in carrying out the terms of this Order.

THE FOLLOWING PARTIES APPROVE THE FORM OF THIS ORDER AND CONSENT TO EACH OF THE ORDERS, IF ANY, THAT ARE INDICATED ABOVE AS BEING BY CONSENT:

Signature of Lawyer for Agent Lawyer: Christopher J. Ramsay / Katie G. Mak

BY THE COURT, Milman, J

Registrar

SCHEDULE "A"

LIST OF COUNSEL

Lisa Hiebort Many Butteny	Petitioners Monitor
Jonathan Ross	Export Queiconvent Canada-

SCHEDULE B

-1

PLAN

FLA

No. **S1910194** Vancouver Registry

IN THE SUPREME COURT OF BRITISH COLUMBIA

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND

IN THE MATTER OF THE BUSINESS CORPORATIONS ACT, S.B.C. 2002, C. 57

AND

IN THE MATTER OF ENERGOLD DRILLING CORP., CROS-MAN DIRECT UNDERGROUND LTD., EGD SERVICES LTD., BERTRAM DRILLING CORP. AND OMNITERRA INTERNATIONAL DRILLING INC.

PETITIONERS

AMENDED PLAN OF COMPROMISE AND ARRANGEMENT

PURSUANT TO THE COMPANIES' CREDITORS ARRANGEMENT ACT, S.B.C. 2002, c.57 and the BUSINESS CORPORATIONS ACT, S.B.C. 2002, c.57

concerning, affecting and involving

ENERGOLD DRILLING CORP., CROS-MAN DIRECT UNDERGROUND LTD., EGD SERVICES LTD., BERTRAM DRILLING CORP. AND OMNITERRA INTERNATIONAL DRILLING INC.

January 13, 2020

PLAN OF COMPROMISE AND ARRANGEMENT

- A. On September 13, 2019, the Supreme Court of British Columbia (the "Court") made an order (the "Initial Order") granting each of Energold Drilling Corp. ("Energold"), Cros-Man Direct Underground Ltd. ("Cros-Man"), EGD Services Ltd., Bertram Drilling Corp. ("Bertram") and Omniterra International Drilling Inc. ("Omniterra")(collectively, the "Petitioners"), protection from their creditors pursuant to the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended (the "CCAA").
- B. Pursuant to an order made by the Court on September 13, 2019 (the "Sale Process Order"), the Court approved a sale solicitation procedure (the "SSP") to be conducted by the Petitioners, with the assistance of FTI Consulting Canada Inc. ("FTI") as the monitor in the CCAA proceedings (in such capacity, the "Monitor") and Ernst & Young Orenda Corporate Finance Inc. as the financial advisor in the CCAA proceedings ("Financial Advisor"), for the solicitation of offers to acquire any of the Petitioners' direct and indirect wholly-owned subsidiaries, including the other Petitioners (collectively, the "Energold Group").
- C. The Financial Advisor and the Petitioners under the Sales Process Order and the SSP conducted a fair and reasonable marketing process that involved identifying a list of potential bidders and preparing a confidential information memorandum that was made available to interested parties and conducting an auction in respect of assets which received multiple qualified offers.
- D. Certain of the Petitioners are the registered and beneficial owners of the shares (the "**Purchased Shares**") in the capital of each issuer set out in **Schedule** "**A**" attached hereto, and each issuer is a member of the Energold Group and the Purchased Shares have been marketed for sale in connection with the Sale Process Order and the SSP.
- E. Cros-Man is the owner of the assets set out in **Schedule "B"** attached hereto (the "**Cros-Man Assets**"), which have also been marketed for sale in connection with the Sales Process Order and the SSP.
- F. Certain of the Petitioners are also the owners, both registered and beneficial, of the assets listed in **Schedule "C"** attached hereto (the "**Residual Assets**", and together with the Purchased Shares and the Cros-Man Assets, herein referred to as the "**Purchased Assets**").
- G. Extract Advisors LLC (the "Agent") is the administrative agent for Noteholders (as defined herein), who are secured creditors with a first ranking security interest over the majority of Energold Group's' assets, pursuant to the Note Purchase Agreement (as defined herein)
- H. Pursuant to the SSP and at the recommendation of the Financial Advisor, the Agent placed bids for the Purchased Shares and the Cros-Man Assets, which bids were deemed to be the stalking horse bids in the following amounts (each amount subject to certain working capital and other adjustments):
 - (a) for the Latin America Unit (as set out in Schedule "A"), a credit bid amount of \$6.8 million (subject to further working capital adjustments to be calculated in due course);
 - (b) for the EMEA Unit (as set out in Schedule "A"), a credit bid amount of \$2 million (subject to further working capital adjustments to be calculated in due course);

14

- (c) for the BDI Unit (as set out in Schedule "A"), a credit bid amount of \$1.5 million (subject to further working capital adjustments to be calculated in due course); and
- (d) for the Cros-Man Assets, a credit bid amount of \$3 million (subject to further working capital adjustments to be calculated in due course).
- I. As of December 16, 2019, the total debt owing to the Noteholders is \$25,701,919.58, with interest continuing to accrue after this date (the "Total Noteholder Claim"). There were no other qualified bids except for the Agent's stalking horse bids for the Latin America Unit, the BDI Unit and the Cros-Man Assets under the SSP. There was one other qualified bidder for the EMEA Unit, which resulted in an auction of the EMEA Unit with the final winning bid by the Agent of \$3.05 million (subject to further working capital adjustments to be calculated in due course). Following the conclusion of the SSP, the Agent, as purchaser, was therefore the successful bidder for the Purchased Shares and the Cros-Man Assets.
- J. Pursuant to the SSP, the Agent, as purchaser, entered into the following share purchase agreements (the "Share Purchase Agreements"):
 - (a) share purchase agreement with Energold and Omniterra, as vendors, dated October 11, 2019 for the purchase of the Purchased Shares in the Latin America Unit for a credit bid amount of \$4,160,000 (giving effect to \$2,640,000 working capital adjustment due to assumption of debt owed by Energold to the Economic Development Corporation of Canada (the "EDC"))(the "Latin America Credit Bid Amount") and a cash payment of \$2,000 (subject to further working capital adjustments to be calculated in due course);
 - (b) share purchase agreement with Energold, as vendor, dated October 31, 2019 for the purchase of the Purchased Shares in the EMEA Unit for a credit bid amount of \$3,050,000 (subject to further working capital adjustments to be calculated in due course) (the "EMEA Credit Bid Amount"); and
 - (c) share purchase agreement with Energold, as vendor, dated October 11, 2019, for the purchase of the Purchased Shares in the BDI Unit for a credit bid amount of \$1,500,000 (subject to further working capital adjustments to be calculated in due course)(the "BDI Credit Bid Amount", and collectively with the Latin America Credit Bid Amount and the EMEA Credit Bid Amount, the "Purchased Shares Credit Bid Amount").
- K. Pursuant to the SSP, the Agent, as purchaser, entered into an asset purchase agreement (the "Cros-Man Asset Purchase Agreement", and together with the Share Purchase Agreements, herein referred to as the "Purchase Agreements"), with Energold, as vendor, dated October 11, 2019 for the purchase of the Cros-Man Assets for a credit bid amount of \$3,000,000 (subject to further working capital adjustments to be calculated in due course)(the "Cros-Man Credit Bid Amount").
- L. The Agent, as purchaser, will also purchase the Residual Assets for a credit bid amount of \$3,869,755 (the "Residual Assets Credit Bid Amount").
- M. The total amount of the Agent's credit bid for the Purchased Assets is \$15,579,755 (subject to further working capital adjustments to be calculated in due course) (the "Credit Bid Amount"). The Credit Bid Amount for each of the Purchased Assets is for Plan (as defined herein) approval purposes only and is subject to the purchaser's reservation of rights regarding the fair market valuation of the assets for all other purposes.

CW14452101.5

- N. The Petitioners and the Agent wish to enter into an arrangement that will see the paydown of the debt owing by the Energold Group to the Noteholders by application of the Credit Bid to each of the Latin America Unit, the EMEA Unit, the BDI Unit, the Cros-Man Assets and the Residual Assets. The Noteholders in return will receive their pro rata share of units in a new limited partnership acquisition vehicle which will manage and run the business formerly owned by the Energold Group. The sale transactions pursuant to the Purchase Agreements will be concluded through implementation of this Plan, the Sanction Order and the Vesting Order (as defined herein) which will transfer the Energold Group's assets directly or indirectly to the US LP (as defined herein).
- O. Upon the implementation of the Plan:

- (a) the Affected Noteholders (as defined herein) will hold their Affected Noteholder Pro Rata Share (as defined herein) of the partnership units in the US LP (as defined herein).
- (b) the US LP will directly or indirectly hold the Purchased Assets which will be managed by the general partner of the US LP;
- (c) the outstanding obligations of Energold to the Affected Noteholders up to the Affected Noteholder Claim (as defined herein) shall be fully extinguished;
- (d) all Intercompany Working Capital Obligations (as defined herein) shall be extinguished; and
- (e) the Transferred Subsidiaries Guarantee (as defined herein) shall remain extant, enforceable and independent of the Note Purchase Agreement.
- P. On December 19, 2019, the Court made an order (the "Meeting Order") accepting for filing the form of plan of compromise and arrangement dated December 19, 2019 attached thereto. The Agent has made amendments to the plan of compromise and arrangement and hereby submits the Plan herein dated January 13, 2019 to be voted upon by the Affected Noteholders.

NOW THEREFORE the Agent hereby proposes and presents this consolidated plan of compromise and arrangement under the CCAA and the *Business Corporations Act*, S.B.C. 2002, c.57.

ARTICLE 2 INTERPRETATION

2.1 Certain Rules of Interpretation

For the purposes of this Plan:

- (a) In this Plan and the Recitals, unless otherwise stated or the subject matter or context otherwise requires, all terms defined herein have their meanings ascribed thereto in Schedule "D".
- (b) Any reference in this Plan to a contract, instrument, release, indenture or other agreement or document being in a particular form or on particular terms and conditions means such document shall be substantially in such form or substantially on such terms and conditions;

- (c) Unless otherwise expressly provided herein, any reference in this Plan to an instrument, agreement or an Order or an existing document or exhibit filed or to be filed means such instrument, agreement, Order, document or exhibit as it may have been or may be amended, modified, or supplemented in accordance with its terms;
- (d) The division of this Plan into articles and sections is for convenience of reference only and does not affect the construction or interpretation of this Plan, nor are the descriptive headings of articles and sections intended as complete or accurate descriptions of the content thereof;
- (e) The use of words in the singular or plural, or with a particular gender, including a definition, shall not limit the scope or exclude the application of any provision of this Plan to such Person (or Persons) or circumstances as the context otherwise permits;
- (f) The words "includes" and "including" and similar terms of inclusion shall not, unless expressly modified by the words "only" or "solely", be construed as terms of limitation, but rather shall mean "includes but is not limited to" and "including but not limited to", so that references to included matters shall be regarded as illustrative without being either characterizing or exhaustive;
- (g) Unless otherwise specified, all references to time herein and in any document issued pursuant hereto mean local time in Vancouver, British Columbia and any reference to an event occurring on a Business Day shall mean prior to 5:00 p.m. on such Business Day;
- (h) Unless otherwise specified, time periods within or following which any payment is to be made or act is to be done shall be calculated by excluding the day on which the period commences and including the day on which the period ends;
- (i) Unless otherwise provided, any reference to a statute or other enactment of parliament, a legislature or other Governmental Entity includes all regulations made thereunder, all amendments to or re-enactments of such statute or regulations in force from time to time, and, if applicable, any statute or regulation that supplements or supersedes such statute or regulation;
- (j) References to a specific Recital, Article or Section shall, unless something in the subject matter or context is inconsistent therewith, be construed as references to that specific Recital, Article or Section of this Plan, whereas the terms "this Plan", "hereof, "herein", "hereto", "hereunder" and similar expressions shall be deemed to refer generally to this Plan and not to any particular Recital, Article, Section or other portion of this Plan and include any documents supplemental hereto; and
- (k) The word "or" is not exclusive.

2.2 Governing Law

This Plan shall be governed by and construed in accordance with the laws of British Columbia and the federal laws of Canada applicable therein. All questions as to the interpretation or application of this Plan and all proceedings taken in connection with this Plan and its provisions shall be subject to the jurisdiction of the Court.

2.3 Currency

Unless otherwise stated, all references in this Plan to sums of money are expressed in, and all payments provided for herein shall be made in, Canadian dollars.

2.4 Date for Any Action

If the date on which any action is required to be taken hereunder by a Person is not a Business Day, such action shall be required to be taken on the next succeeding day which is a Business Day.

2.5 Time

Time shall be of the essence in this Plan.

ARTICLE 3 PURPOSE AND EFFECT OF THIS PLAN

3.1 Purpose

The purpose of this Plan is to

- (a) implement the transfer of the Purchased Assets through the Sanction Order and the Vesting Orders;
- (b) compromise the Affected Noteholder Claims and distribute to the Affected Noteholders their Affected Noteholder Pro Rata Share of units in the US LP;
- (c) extinguish all Intercompany Working Capital Obligations; and
- (d) release Energold of its obligations to the Affected Noteholders to the extent of their Affected Noteholder Claims; and
- (e) allow the Transferred Subsidiaries to continue operations under the US LP with the Affected Noteholders as holders of US LP Units pursuant to the US LP Agreement which will preserve operations and employee jobs;

in the expectation that Persons who have an economic interest in the Energold Group's operations, when considered as a whole, will derive a greater benefit from the implementation of the Plan than would result from a liquidation.

3.2 Effectiveness

Subject to the satisfaction, completion or waiver (to the extent permitted pursuant to section 10.5) of the conditions precedent set out herein, this Plan will become effective in the sequence described in Section 8.2 from and after the Effective Time and shall be binding on and enure to the benefit of the Petitioners, the Affected Noteholders, and all other Persons as provided for herein, or subject to, this Plan and their respective successors and assigns and their respective heirs, executors, administrators and other legal representatives, successors and assigns.

3.3 Persons Not Affected

This Plan does not affect Excluded Creditors to the extent of their Excluded Claims. Nothing in this Plan shall affect the Energold Group's rights and defences, both legal and equitable, with respect to any Excluded Claims, including, but not limited to, all rights with respect to legal and equitable defences or entitlements to set-offs or recoupments against such Excluded Claims. Nothing herein shall constitute a waiver of any rights of any of the Energold Group to dispute the quantum or validity of an Excluded Claim.

ARTICLE 4

CLASSIFICATION, AFFECTED NOTEHOLDER CLAIMS AND RELATED MATTERS

4.1 Classes

For the purposes of considering, voting on, and receiving distributions under this Plan, there shall be one class being the Affected Noteholder Class.

4.2 Claims of Affected Noteholders

Affected Noteholders shall be entitled to vote their Affected Noteholder Claims at the Creditors' Meeting in respect of this Plan and shall be entitled to receive distributions on account of their Affected Noteholder Claims as provided under and pursuant to this Plan.

4.3 Excluded Claims

Excluded Claims shall not be compromised under the Plan. No Excluded Creditor shall be:

- (a) entitled to vote or attend in respect of their Excluded Claims at any Creditors' Meeting to consider and approve this Plan; or
- (b) entitled to receive any distribution or consideration under this Plan in respect of such Excluded Claim.

4.4 Creditors' Meeting

- (a) The Creditors' Meeting shall be held in accordance with this Plan, the Meeting Order and any further Order in the CCAA Proceedings. Subject to the terms of any further Order in the CCAA Proceedings, the only Persons entitled to notice of, to attend or to speak at the Creditors' Meeting are the Agent, the Affected Noteholders (or their respective duly-appointed proxyholders), representatives of the Monitor, the Petitioners, all such parties' financial and legal advisors, the Chair, Secretary and Scrutineers (all as defined in the Meeting Order). Any other person may be admitted to the Creditors' Meeting only by invitation of the Petitioners or the Chair.
- (b) If this Plan is approved by the Required Majority, then this Plan shall be deemed to have been agreed to, accepted and approved by the Affected Noteholders and shall be binding upon all Affected Noteholders immediately upon the delivery of the Monitor's Certificate in accordance with section 10.6 hereof.

4.5 Payments to Employees

If not otherwise paid pursuant to this Plan the Petitioners will pay in full all employee-related payments required by subsection 6(5) of the CCAA, provided that this Section 4.5 shall not require payment of any employee-related amounts in advance of the normal payroll cycle applicable to employees.

ARTICLE 5 TREATMENT OF CLAIMS

5.1 Treatment of Affected Noteholder Claims

- (a) On the Implementation Date, in accordance with this Plan and in accordance with the steps and in the sequence set forth in Section 8.2, each Affected Noteholder shall be entitled to receive a distribution of its Affected Noteholder Pro Rata Share of the LP Units which shall, and shall be deemed to, be received in full and final settlement of its Affected Noteholder Claim.
- (b) On the Implementation Date, each Affected Noteholder shall be deemed to be a party to the US LP Agreement, each in its capacity as a holder of US LP Units.

5.2 Priority Claims

- (a) In accordance with the Sanction Order, the CCAA and with the steps and in the sequence set forth herein, Section 8.2 and 8.3, the Employee Priority Claims and the Crown Priority Claims, if any, shall be paid from the Priority Claim Reserve Account.
- (b) Subject to the Effective Time occurring:
 - all Crown Priority Claims that were outstanding as at the Filing Date shall be paid in full by the Monitor on behalf of the Petitioners, from the Priority Claim Reserve within six months after the Sanction Order, as required by subsection 6(3) of the CCAA; and
 - (ii) all Employee Priority Claims to the extent unpaid prior to the Implementation Date shall be paid by the Monitor, on behalf of the Petitioners, from the Priority Claim Reserve immediately after the Sanction Order as required by subsection 6(5) of the CCAA.

5.3 Excluded Claims

Excluded Creditors in respect to and to the extent of their Excluded Claims shall not receive any consideration under this Plan in respect of their Excluded Claims. Excluded Creditors shall not be entitled to vote on this Plan at the Creditors' Meeting in respect of their Excluded Claims.

5.4 Extinguishment of Claims

On the Implementation Date, in accordance with its terms and in the sequence set forth in Section 8.2 herein and in accordance with the provisions of the Sanction Order, the treatment of Affected Noteholder Claims and all Released Claims, in each case as set forth herein, shall be final and binding on the Petitioners, all Affected Noteholders (and their respective heirs, executors, administrators, legal personal representatives, successors and assigns) and any Person holding a Released Claim. All Affected

Noteholder Claims, all Intercompany Working Capital Obligations and all Released Claims shall be fully, finally, irrevocably and forever released, discharged, cancelled and barred, and Energold shall thereupon have no further obligation whatsoever in respect of the Affected Noteholder Claims and the Released Claims, as applicable; provided that nothing herein releases Energold or any other Person from their obligations to make distributions in the manner and to the extent provided for in the Plan and provided further that such discharge and release of Energold shall be without prejudice to the right of an Affected Noteholder In respect of an Unaffected Noteholder Claim.

5.5 Set-Off

The law of set-off applies to all Claims.

ARTICLE 6 CREATION OF POOL AND RESERVES

6.1 Creation of the Administrative Reserve

- (a) At least three Business Day prior to the Implementation Date, the Agent shall deliver to BLG, in trust, by way of wire transfer (in accordance with the wire transfer instructions provided in the Purchase Agreements), Cash in the amount necessary to establish the Administrative Reserve.
- (b) BLG shall hold the Administrative Reserve in the Administrative Reserve Account for the purpose of paying the Administrative Reserve Costs and shall distribute any remaining balance in the Administrative Reserve Account to the Agent, in accordance with section 8.3 of the Plan.

6.2 Creation of the Priority Claim Reserve

- (a) At least three Business Days prior to the Implementation Date, the Petitioners shall deliver to the Monitor by way of wire transfer (in accordance with the wire transfer instructions provided by the Monitor at least five (5) Business Days prior to the Implementation Date), Cash in the amount necessary to establish the Priority Claim Reserve.
- (b) The Monitor shall hold the Priority Claim Reserve in the Priority Claim Reserve Account for the purpose of paying the Priority Claims in accordance with this Plan and shall distribute any remaining balance in the Priority Claim Reserve Account to the Applicants, in accordance with section 8.3 of this Plan.

ARTICLE 7

PROVISIONS REGARDING DISTRIBUTIONS AND DISBURSEMENTS

7.1 Distributions and Disbursements Generally

- (a) All distributions and disbursements to be effected pursuant to the Plan shall be made pursuant to this Article 7 and shall occur in the manner set out below under the supervision of the Monitor.
- (b) All distributions and disbursements to be effected pursuant to this Plan on account of Affected Noteholder Claims shall be made by the Agent for the benefit of the Affected

Noteholders holding such Affected Noteholder Claims as at the Implementation Date and the Petitioners, the Monitor, the Agent and their agents shall have no obligation to deal with a transferee or assignee of such Affected Noteholder Claim after the Implementation Date in respect of any such matter. Affected Noteholders who assign their Affected Noteholder Claims after the Implementation Date shall be wholly responsible for ensuring that plan distributions intended to be included within such assignments are in fact delivered to the assignee and neither the Petitioners, the Monitor, the Agent, as applicable, shall have any liability in connection therewith.

7.2 Issuance and Delivery of US LP Units

Sec. Land

- (a) The US LP Units to be distributed under this Plan will be made by the Agent to the Affected Noteholders.
- (b) On the Implementation Date or as soon as reasonably practicable thereafter, the US LP, on account of Affected Noteholder Claims, shall lodge in its records the US LP Units to be distributed to the Affected Noteholders in the name of and to the address as recorded in the books and records of the Agent or as otherwise communicated to the Agent not less than three Business Days prior to the anticipated Implementation Date.
- Notwithstanding Section 7.2(b), no Affected Noteholder shall be entitled to the rights (c) associated with the US LP Units and all such US LP Units shall be reserved for issuance on the books and records of US LP until such time as it has delivered its US LP Unitholder Information to the Agent and/or US LP, as applicable. In the event that a Affected Noteholder fails to deliver its US LP Unitholder Information in accordance with this Section 7.2(c) on or before the date that is 6 months following the Implementation Date, the US LP shall have no further obligation to issue or deliver, and shall have no further obligation to reserve on its books and records, any LP Units otherwise issuable to Affected Noteholders (such equity, the "Unissued US LP Units") that have not delivered their US LP Unitholder Information accordance this Section 7.2(c) and all such Affected Noteholders shall cease to have a claim to, or interest of any kind or nature against or in, the Petitioners, the US LP or the Unissued US LP Units and the Agent shall delete such Unissued US LP Units from the books and records of the US LP as maintained by the Agent and/or US LP. Upon receipt by the Agent of the US LP Unitholder Information in accordance with this Section 7.2(c), as soon as reasonably practicable thereafter the Agent shall cause the US LP to lodge in the records of the US LP the Affected Noteholder as the holder of its Affected Noteholder Pro Rata Share.
- (d) No fractional US LP Units of US LP shall be allocated or issued under this Plan. Any legal, equitable, contractual and any other rights or claims (whether actual or contingent, and whether or not previously asserted) of any Person with respect to fractional US LP Units of US LP issued pursuant to this Plan shall be rounded down to the nearest whole number without compensation therefor.

7.3 Tax Matters

(a) Notwithstanding any provisions of the Plan, each Person that receives a distribution, disbursement or other payment pursuant to the Plan shall have sole and exclusive responsibility for the satisfaction and payment of any Tax obligations imposed on such

Person by any Taxing Authority on account of such distribution, disbursement or payment.

- (b) Any payor shall be entitled to deduct and withhold and remit from any distribution, payment or consideration otherwise payable to any Person pursuant to the Plan such amounts as are required (a "Withholding Obligation") to be deducted and withheld with respect to such payment under the ITA, or any provision of federal, provincial, territorial, state, local or foreign tax law, in each case, as amended or succeeded. All distributions made pursuant to the Plan shall be first in satisfaction of the portion of Affected Noteholder Claims that are not subject to any Withholding Obligation.
- (c) To the extent that amounts are withheld or deducted and paid over to the applicable Taxing Authority, such withheld or deducted amounts shall be treated for all purposes of the Plan as having been paid to such Person as the remainder of the payment in respect of which such withholding and deduction were made.
- (d) For the avoidance of doubt, it is expressly acknowledged and agreed that the Monitor and any Director or Officer will not hold any assets hereunder, including Cash, or make distributions, payments or disbursements, and no provision hereof shall be construed to have such effect.

ARTICLE 8 IMPLEMENTATION

8.1 Corporate Authorizations

The adoption, execution, delivery, implementation and consummation of all matters contemplated under this Plan involving corporate action of the Petitioners will occur and be effective as of the Implementation Date, and will be authorized and approved under this Plan and by the Court, where appropriate, as part of the Sanction Order, in all respects and for all purposes without any requirement of further action by shareholders, directors or officers of any of the Petitioners. All necessary approvals to take actions shall be deemed to have been obtained from the Directors or the shareholders of the Petitioners, as applicable, including resolution or special resolution with respect to any of the steps contemplated by this Plan shall be deemed to be effective.

8.2 Implementation Date Transactions

Commencing at the Effective Time, the following events or transactions will occur, or be deemed to have occurred and be taken and effected at the time, manner or order set out in this section (or in such other manner, order or at such other time or times as the Agent may determine in consultation with the Monitor and with the Petitioners), without any further act or formality required on the part of any Person, except as may be expressly provided herein.

- (a) The Agent shall cause US LP to be formed.
- (b) The Agent shall incorporate two new Canadian corporations, Energold Drilling (Canada) Inc. ("New Energold") and Cros-Man Direct Underground (Acq) Inc. ("New Cros-Man"), each of which on incorporation shall be a wholly owned subsidiary of Energold.
- (c) Cros-Man shall transfer and shall be deemed to transfer all the Cros-Man Assets to New Cros-Man in accordance with the Vesting Orders. In consideration for such transfer,

 ~ 4

New Cros-Man shall assume the obligations and indebtedness of Cros-Man under the Transferred Subsidiaries Guarantee to the extent of the Cros-Man Credit Bid Amount.

- (d) The applicable Petitioners shall transfer and shall be deemed to transfer all of those of the Purchased Shares that are issued by corporations incorporated in Canada and Residual Assets to New Energold in accordance with the Vesting Orders. In consideration, New Energold shall assume the obligations and indebtedness of such applicable Petitioners under the Transferred Subsidiaries Guarantee to the extent of the Purchased Shares Credit Bid Amount attributed to such corporations and the Residual Asset Credit Bid Amount.
- (e) Energold shall transfer to the Agent all of the Purchased Assets, all of the shares of New Energold and New Cros-Man and the benefit of the Note Purchase Agreement and Transferred Subsidiaries Guarantee, as applicable, in full and final settlement of the obligations of Energold under the Note Purchase Agreement up to the extent of the Affected Noteholder Claims.
- (f) The Agent shall transfer to US LP all of the Purchased Assets, all of the shares of New Energold and New Cros-Man and the benefit of the Note Purchase Agreement and Transferred Subsidiaries Guarantee, as applicable. In consideration, the US LP shall issue to the Agent the US LP Units for the benefit of the Affected Noteholders in accordance with the Plan.
- (g) The Administration Charge shall be deemed to be released as against the Purchased Assets and shall attach to the Administrative Charge Reserve pursuant to and in accordance with the Vesting Orders.
- (h) The Affected Noteholders shall be entitled to the treatment set out in section 5.1 hereof in full and final settlement of their Affected Noteholder Claims, and the Affected Noteholder Claims shall, and shall be deemed to be, irrevocably and finally extinguished and such Affected Noteholders shall have no further right, title or interest in and to its Affected Noteholder Claim; and
 - (i) each Affected Noteholder shall be deemed to be a party to the US LP Agreement, each in its capacity as a holder of US LP Units; and
 - (ii) the Financial Advisor Charge, the Interim Lender's Charge and the D&O Charge shall be deemed to be discharged as against the Purchased Assets pursuant to and in accordance with the Vesting Orders.
- (i) The releases and injunctions referred to in accordance with Section 5.4 and Article 9 hereof shall become effective.

8.3 **Post-Implementation Date Transactions**

(a) As soon as practicable, but in no event more than 10 days, following the Implementation Date, the Petitioners shall prepare, or cause to be prepared, and deliver to the Agent a statement setting forth in reasonable detail the final calculation of the Administrative Reserve Costs in accordance with the terms of the Purchase Agreements.

- (b) Upon final determination of the Administrative Reserve Costs in accordance with the terms of the Purchase Agreements, the Monitor shall instruct BLG to pay:
 - (i) the Administrative Reserve Costs from the Administrative Reserve Fund; and
 - (ii) the difference between the Estimated Administrative Reserve Costs less the Administrative Reserve Costs, if any, to the Agent.
- (c) In the event the Administrative Reserve Costs are greater than the Estimated Administrative Reserve Costs, the Agent shall pay to the Monitor the difference no later than five business days after the final determination of the Administrative Reserve Costs in accordance with the Purchase Agreements.

ARTICLE 9 RELEASES

9.1 Plan Releases

. !

- (a) At the Effective Time, Energold shall be released and discharged from any and all demands, claims, liabilities, causes of action, debts, accounts, covenants, damages, executions and other recoveries based in whole or in part on any act or omission, transaction, dealing or other occurrence existing or taking place on or prior to the Implementation Date, applications, counterclaims, suits, sums of money, judgments, orders, including for injunctive relief or specific performance and compliance orders, expenses, encumbrances and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any Affected Noteholder, to the extent of their Affected Noteholder Claim, may be entitled to assert, whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Effective Time arising out of or in connection with the Notes or the Plan, and all such Claims shall be forever waived and released (other than the right to enforce Energold's obligations under the Plan or any related document), all to the full extent permitted by Applicable Law, save and except any gross negligence or wilful misconduct, provided that nothing herein shall release or discharge any of the Energold Group from any Excluded Claims.
- (b) At the Effective Time, the Petitioners' legal counsel, the Financial Advisor, the Monitor, the Agent and their respective subsidiaries and affiliates and each of their respective shareholders, partners, officers, directors, current and former employees, financial advisors, legal counsel and agents (being referred to individually as a "Third Party Released Party") are hereby released and discharged from any and all demands, claims, liabilities, causes of action, debts, accounts, covenants, damages, executions and other recoveries based in whole or in part on any act or omission, transaction, dealing or other occurrence existing or taking place on or prior to the Implementation Date, actions, applications, counterclaims, suits, sums of money, judgments, orders, including for injunctive relief or specific performance and compliance orders, expenses, encumbrances and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any Person may be entitled to assert, whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or

hereafter arising, based in whole or in part on any omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Implementation Date arising out of or in connection with the Affected Noteholder Claims or the Plan, and any Claims, and all claims arising out of such actions or omissions shall be forever waived and released (other than the right to enforce the Monitor's or the Agent's obligations under the Plan or any related document), all to the full extent permitted by Applicable Law save and except any gross negligence or wilful misconduct on the part of the Third Party Released Party, provided that nothing herein shall release or discharge any Third Party Released Party if such Third Party Released Party is judged by the express terms of a judgment rendered on a final determination on the merits to have committed criminal, fraudulent or other wilful misconduct.

(c) The Sanction Order will enjoin the prosecution, whether directly, derivatively or otherwise, of any claim, obligation, suit, judgment, damage, demand, debt, right, cause of action, liability or interest released, discharged, compromised or terminated pursuant to the Plan.

9.2 Timing of Releases and Injunctions

All releases and injunctions set forth in this Article 9 shall become effective on the Implementation Date.

9.3 Knowledge of Claims

Each Person to which Section 9.1 hereof applies shall be deemed to have granted the releases set forth in Section 9.1 notwithstanding that it may hereafter discover facts in addition to, or different from, those which it now knows or believes to be true, and without regard to the subsequent discovery or existence of such different or additional facts, and such party expressly waives any and all rights that it may have under any applicable law which would limit the effect of such releases to those Claims or causes of action known or suspected to exist at the time of the granting of the release.

ARTICLE 10

COURT SANCTION, CONDITIONS PRECEDENT AND IMPLEMENTATION

10.1 Application for Sanction Order

If this Plan is approved by the Required Majority, the Agent shall apply for the Sanction Order on the date set out in the Meeting Order or such later date as the Court may set.

10.2 Vesting Orders

The Agent shall apply for the order(s) vesting the Purchased Assets in the Agent and/or Canadian Holdcos, as applicable (the "Vesting Orders") at the same time the application is made for the Sanction Order.

10.3 Conditions to the Implementation Date

The implementation of this Plan shall be conditional upon the fulfillment, satisfaction or waiver (to the extent permitted by Section 10.4 hereof) of the following conditions:

(a) the Plan shall have been approved by the Required Majority;

- (b) the DIP Loan shall be fully repaid;
- (c) the Court shall have granted the Sanction Order the operation and effect of which shall not have been stayed, reversed or amended and in the event of an appeal or application for leave to appeal, final determination shall have been made by the appellate court;
- (d) the Court shall have granted the Vesting Orders the operation and effect of which shall not have been stayed, reversed or amended and in the event of an appeal or application for leave to appeal, final determination shall have been made by the appellate court;
- (e) the Administrative Reserve shall have been funded by the Agent;
- (f) the Priority Claim Reserve shall have been funded by the Petitioners; and
- (g) the Implementation Date shall have occurred no later than the Outside Date.

10.4 Waiver of Conditions

18 14 M

The Agent and the Petitioners, in consultation with the Monitor, may agree at any time and from time to time waive the fulfillment or satisfaction, in whole or in part, of the conditions set out herein, to the extent and on such terms as such parties may agree to.

10.5 Implementation Provisions

If the conditions contained in Section 10.3 are not satisfied or waived (to the extent permitted under Section 10.4) by the Outside Date, unless the Petitioners, in consultation with the Monitor, and the Agent, agree in writing to extend such period, this Plan and the Sanction Order shall cease to have any further force or effect and will not be binding on any Person.

10.6 Monitor's Certificate of Plan Implementation

Upon written notice from the Petitioners and the Agent (or counsel on their behalf) to the Monitor that the conditions to Plan implementation set out in Section 10.3, have been satisfied or waived, the Monitor shall, as soon as possible following receipt of such written notice, deliver to the Petitioners and the Agent (or counsel on their behalf) and file with the Court, a certificate substantially in the form attached hereto as **Schedule "E"** (the "**Monitor's Certificate**") which states that all conditions precedent set out in Section 10.3 have been satisfied or waived and that Implementation Date (which shall be set out on the certificate) has occurred.

ARTICLE 11 GENERAL

11.1 Deeming Provisions

In this Plan, the deeming provisions are not rebuttable and are conclusive and irrevocable.

11.2 Non-Consummation

If the Implementation Date does not occur on or before the Outside Date (as the same may be extended in accordance with the terms hereof), or if this Plan is otherwise withdrawn in accordance with its terms:

- (a) this Plan shall be null and void in all respects; and
- (b) nothing contained in this Plan, and no acts taken in preparation for consummation of this Plan, shall:
 - (i) constitute or be deemed to constitute a waiver or release of any Claims by or against the Petitioners, their respective successors or any other Person;
 - (ii) prejudice in any manner the rights of the Petitioners, their respective successors or any other Person in any further proceedings involving the Petitioners or their respective successors; or
 - (iii) constitute an admission of any sort by the Petitioners, their respective successors or any other Person.

11.3 Modification of Plan

- (a) The Agent reserves the right to amend, restate, modify and/or supplement this Plan at any time and from time to time, provided that (except as provided in subsection (c) below) any such amendment, restatement, modification or supplement must be contained in a written document that is:
 - (i) filed with the Court and, if made following the Creditors' Meeting, approved by the Court; and
 - (ii) approved by the Monitor and communicated to the Affected Noteholders in the manner required by the Court (if so required):
 - (A) if made prior to or at the Creditors' Meeting:
 - the Chair (as defined in the Meeting Order) shall communicate the details of any such amendment, restatement, modification and/or supplement to Affected Noteholders and other Persons present at the Creditors' Meeting prior to any vote being taken at the Creditors' Meeting;
 - (II) the Agent shall provide notice to the service list of any such amendment, restatement, modification and/or supplement and shall file a copy thereof with the Court forthwith and in any event prior to the Court hearing in respect of the Sanction Order; and
 - the Monitor shall post an electronic copy of such amendment, restatement, modification and/or supplement on the Monitor's Website forthwith and in any event prior to the Court hearing in respect of the Sanction Order;
 - (B) if made following the Creditors' Meeting:

- (I) the Agent shall provide notice to the service list of any such amendment, restatement, modification and/or supplement and shall file a copy thereof with the Court;
 - the Monitor shall post an electronic copy of such amendment, restatement, modification and/or supplement on the Monitor's Website; and
 - (III) such amendment, restatement, modification and/or supplement shall require the approval of the Court following notice to the service list.
- (b) Any amendment, modification or supplement to this Plan may be proposed by the Petitioners with the consent of the Monitor and the Agent at any time prior to or at the Creditors' Meeting, with or without any prior notice or communication (other than as may be required under the Initial Order), and if so proposed and affected at the Creditors' Meeting, shall become part of this Plan for all purposes.
- (c) Any amendment, modification or supplement to this Plan may be made following the Creditors' Meeting by the Petitioners, with the consent of the Monitor, without requiring filing with, or approval of, the Court, provided that it concerns a matter which is of an administrative nature and is required to better give effect to the implementation of this Plan and is not adverse to the financial or economic interests of any of the Consenting Parties or any Affected Noteholders.

11.4 Severability of Plan Provisions

If, prior to the Effective Time, any term or provision of this Plan is held by the Court to be invalid, void or unenforceable, at the request of the Agent, the Court shall have the power to either:

- (a) sever such term or provision from the balance of this Plan and provide the Agent with the option to proceed with the implementation of the balance of this Plan as of and with effect from the Effective Time; or
- (b) alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision shall then be applicable as altered or interpreted.

Notwithstanding any such holding, alteration or interpretation, and provided that this Plan is implemented, the remainder of the terms and provisions of this Plan shall remain in full force and effect and shall in no way be affected, impaired or invalidated by such holding, alteration or interpretation.

11.5 Preservation of Rights of Action

Except as otherwise provided in this Plan, the Sanction Order, the Vesting Orders or any Order of the Court, or in any contract, instrument, release, indenture or other agreement entered into in connection with this Plan, following the Implementation Date, the Petitioners will retain and may enforce, sue on, settle, or compromise (or decline to do any of the foregoing) all claims, rights or causes of action, suits and proceedings, whether in law or in equity, whether known or unknown, that the Petitioners may hold against any Person or entity without further approval of the Court.

11.6 Responsibilities of Monitor

The Monitor is acting and will continue to act in all respects in its capacity as Monitor in the CCAA Proceedings with respect to the Petitioners and not in its personal or corporate capacity, and shall have no liability in connection with the implementation of this Plan, including without limitation with respect to making distributions pursuant to and in accordance with the Plan, the establishment and administration of the Administrative Reserve and the Priority Claim Reserve (and in each case, any adjustments with respect to same) or the timing or sequence of the plan transaction steps, in each case save and except for gross negligence and wilful misconduct. The Monitor shall not be responsible or liable whatsoever for any obligations of the Petitioners. The Monitor shall at all times have the powers and protections granted to it by the Plan, the CCAA, the Initial Order, the Meeting Order, and any other Order made in the CCAA Proceedings.

11.7 Different Capacities

Persons who are affected by the Plan may be affected in more than one capacity. Unless expressly provided herein to the contrary, a Person will be entitled to participate hereunder in each such capacity. Any action taken by a Person in one capacity will not affect such Person in any other capacity, unless expressly agreed by a Person in writing or unless its Claims overlap or are otherwise duplicative.

11.8 Notices

Any notice or other communication to be delivered hereunder must be in writing and refer to this Plan and may, as hereinafter provided, be made or given by personal delivery, ordinary mail, email or by facsimile addressed to each of the respective Parties as follows:

(a) the Petitioners

1100 – 543 Granville Street Vancouver, British Columbia V6C 1X8

Attention: Mark Berger Email: mberger@pppllc.com

with a required copy (which shall not be deemed notice) to:

Borden Ladner Gervais LLP 1200 Waterfront Centre 200 Burrard Street Vancouver, British Columbia V7X 1T2

Attention: Lisa Hiebert / Ryan Laity

Email: ihiebert@blg.com / rlaity@blg.com

(b) The Monitor

FTI Consulting Canada Inc. Suite 15-131 555 Burrard Street Vancouver, British Columbia V7X 1M8

Attention:Craig Munro / Tom PowellEmail:craig.munro@fticonsulting.com / tom.powell@fticonsulting.com

with a required copy (which shall not be deemed notice) to:

Cassels Brock & Blackwell LLP 2200-885 West Georgia Street Vancouver, British Columbia V6C 3E8

 Attention:
 Mary Buttery, Q.C. / Lance Williams

 Email:
 mbuttery@cassels.com / lwilliams@cassels.com

(c) If to the Agent

. 1

Extract Capital 379 West Broadway, Suite 423 New York, NY 10012

Attention: Darin Milmeister Email: darin@extractcapital.com

with a required copy (which shall not be deemed notice) to:

Clark Wilson LLP 900-885 West Georgia Street Vancouver, British Columbia V6C 3H1

 Attention:
 Christopher Ramsay / Katie Mak

 Email:
 cramsay@cwilson.com / kmak@cwilson.com

or to such other address as any Party may from time to time notify the others in accordance with this section. Any such communication so given or made shall be deemed to have been given or made and to have been received on the day of delivery if delivered, or on the day of faxing or emailing, provided that such day in either event is a Business Day and the communication is so delivered, faxed or emailed before 5:00 p.m. on such day. Otherwise, such communication shall be deemed to have been given and made and to have been received on the next following Business Day.

11.9 Paramountcy

From and after the Effective Time, any conflict between:

- (a) this Plan; and
- (b) the covenants, warranties, representations, terms, conditions, provisions or obligations, expressed or implied, of any contract, mortgage, security agreement, indenture, trust

indenture, note, loan agreement, commitment letter, agreement for sale, lease or other agreement, written or oral and any and all amendments or supplements thereto existing between any Person and the Petitioners and/or the Transferred Subsidiaries as at the Implementation Date, will be deemed to be governed by the terms, conditions and provisions of this Plan and the Sanction Order, which shall take precedence and priority.

11.10 Further Assurances

. .÷ E

Notwithstanding that the transactions and events set out herein will occur and be deemed to occur in the order set out in this Plan without any further act or formality, each of the Persons named or referred to in, or subject to, this Plan will make, do and execute, or cause to be made, done and executed, all such further acts, deeds, agreements, transfers, assurances, instruments or documents as may reasonably be required by any of them to carry out the full intent and meaning of this Plan and to give effect to the transactions contemplated herein.

Dated this 13th day of January, 2020.

SCHEDULE A

PURCHASED SHARES

LATIN AMERICA UNIT

and the second second

,

Issuer / Transferred Subsidiary	Jurisdiction of Incorporation or Formation	Applicable Vendor	Number and Class of Shares	Certificate No. (if applicable)	Percentage of Total Issued and Outstanding Shares of Issuer
Omniterra International Drilling Inc.	British Columbia	Energold Drilling Corp.	50 Class B Common Shares	B-1 (under the Issuer's former corporate name, Kluane International Drilling Inc.)	
E. Drilling de Nicaragua S.A.	Nicaragua	Energold Drilling Corp	98 Acciones Comunes y Nominativas	1	
Energold Argentina S.A.	Buenos Aires, Argentina	Energold Drilling Corp.	1,998 Acciones Ordinarias	1	
Energold de Colombia S.A.S.	Bogotá, Columbia	Energold Drilling Corp.	3,308,875 Acciones Ordinarias	DEFINITIVO No. 001	
Energold de Colombia S.A.S.	Bogotá, Columbia	Energold Drilling Corp.	428,125 Acciones Ordinarias	DEFINITIVO No. 002	
Energold de Colombia S.A.S.	Bogotá, Columbia	Energold Drilling Corp.	5,000 Acciones Ordinarias	DEFINITIVO No. 003	
Energold de México, S.A. de C.V.	México	Energold Drilling Corp.	49,999 acciones comunes, ordinarias, nominativas, Serie "A"	No. 1 C.F. "A"	
Energold de México, S.A. de C.V.	México	Energold Drilling Corp.	1,596,102 acciones comunes, ordinarias, nominativas, Serie B"	No. 1 C.V. "B"	
OroEnergy S.A.	Chile	Energold Drilling Corp.	99 Common Shares		
Energold Drilling Dominicana S.R.L.	Dominican Republic	Energold Drilling Corp.	49,999 Cuotas		· · · · · · · · · · · · · · · · · · ·

BDI UNIT

Issuer / Jurisdiction of Transferred Incorporation Subsidiary or Formation	Applicable Vendor	Number and Class of Shares	Certificate No. (if applicable)	Percentage of Total Issued and Outstanding Shares of
--	----------------------	-------------------------------	------------------------------------	--

						Issuer
Bertram Inc.	Drilling,	Wyoming	Bertram Drilling Corp.	500 shares without par value	5	
Bertram Inc.	Drilling,	Wyoming	Bertram Drilling Corp.	190 shares without par value	6	
Bertram Inc.	Drilling,	Wyoming	Bertram Drilling Corp.	1,000 shares of non-cumulative non-voting preferred stock without par value	9	
Bertram Inc.	Drilling	Wyoming	Bertram Drilling Corp.	381 shares of non-cumulative non-voting preferred stock without par value	10	

EMEA UNIT

and the second second

lssuer / Transferred Subsidiary	Jurisdiction of Incorporation or Formation	Applicable Vendor	Number and Class of Shares	Certificate No. (if applicable)	Percentage of Total Issued and Outstanding Shares of Issuer
E Global Drilling Corp.	Barbados	Energold Drilling Corp.	100 Common Shares	1	
E Global Drilling Corp.	Barbados	Energold Drilling Corp.	1,500,000 Common Shares	2	
E Global Drilling Corp.	Barbados	Energold Drilling Corp.	10,080,000 Common Shares	3	
E Global Drilling Corp.	Barbados	Energold Drilling Corp.	5,813,313 Common Shares	4	
E Global Drilling Corp.	Barbados	Energold Drilling Corp.	2,000,000 Common Shares	5	

SCHEDULE B

CROS-MAN ASSETS

Contracts

Building Lease

1. Lease with 5698295 Manitoba Ltd.

Equipment Leases

- 2. Lease for a 2016 Caterpillar Hydraulic Excavator with Caterpillar Financial Services Limited
- 3. Lease for a 2015 Caterpillar Backhoe Loader with Caterpillar Financial Services Limited
- 4. Lease for a Vermeer D6x6 Directional Drill with De Lage Landen Financial Services Canada Inc.
- 5. Lease for a 2018 Rebel / 2019 Kenworth hydro-vac system with Paccar Financial

Vehicle Lease

6. Lease for a 2018 Dodge Ram with Meridian OneCap Credit Corp.

- 24 -

s cruist

<u>Equipment</u>

AS SET OUT IN THE ATTACHED SCHEDULE

.

B. Hole The Doment of A point Part P	-	Unit	Description	Oceaning	Adultions Discostals	Cloring	Rule	Openting	Accumulated Depresiation Additions Disposals	Contract	UCC	Rain
DDD Tell Submits data (DATA) NUMBOR TODAD TODAD <thtodad< th=""> <thtodad< th=""> <thtodad< <="" th=""><th></th><th></th><th></th><th></th><th></th><th>Citoring</th><th></th><th>Орентин</th><th>Automotion Disposein</th><th>Closing</th><th></th><th>Rain</th></thtodad<></thtodad<></thtodad<>						Citoring		Орентин	Automotion Disposein	Closing		Rain
Docs If Control of the Section State Transmission State Section State Sect		0-03	TR40 Directional Drit (Parts)	10 000 00		10,000,00	0.70	4 000 57	600 W	6 744 44		
CD-S 2010 Declaring to see 12000 W1, W102722, W1000 W1, W10272, W1000 W10000 W1000 W10												
DD-DE DD-DE <th< td=""><td></td><td>00-05</td><td>2012 Directioner One Mediat D36K50 Vibl. 1VR4230D4C1000989</td><td></td><td></td><td>270,000,00</td><td></td><td></td><td></td><td></td><td></td><td></td></th<>		00-05	2012 Directioner One Mediat D36K50 Vibl. 1VR4230D4C1000989			270,000,00						
Part D Col Verme (EF grade masked maske						300,000,00	0.20	140,000,00		142 000 00		
DD-3 270 Example data data data data data data data dat												
DO-00 211 Bandwin August 2000000000000000000000000000000000000												
Mr.M. DT/ Margin GE 2012 MIL MARINGE- (1972) Table										142,000.00		
Apr-1 Anthe April 10 27 (Transmer of Linear J. 1992) Linear J. 1992 (Heiner J.												
Description UNINE PROFILE											55,009.00	e
Di Langener, 1997 Di Langener,	AUE-10		Autom departy for the 2017 Vermon and	4,482,00	· · · · · · · · · · · · · · · · · · ·		0.20				3,563,19	
Mill B 2000 C/ Mode SM Bach Lin With C/MESCHAMPT Bigg B Status 11111 B (S) B Status 1122 11111 B (S) B Status 1122 11111 B (S) B 1122 1122 11111 B (S) B 1122	or Females	- 41420		1,074,062,00 1		1,074,002,00		e0,194,02	T1.391.23	\$34,523.15	540,398.43	
M M 20 DSC (11 Mod 00 Mass 11, Mod 00 Marg) 2013 Marg) M (20, 00 Marg) T (20, 00 Marg) <tht (20,="" 00="" marg)<="" th=""> T (20, 00 Marg)</tht>			TONA CAT Mandai ANNE Bank Mana MBA CATMANOCOVIAMINTOS	1 mmm		61 mm m						
The distance interm Trippe interm <thtrippe interm<="" th=""> Trippe interm Tri</thtrippe>												
Hole Status										23,800.00		
n = 6 Example 1000 100000 100000 100000											42,441,44	
Part Holl Example data of ESSAL 507 Part (2000) EXAMPLE											6,565.56	
PACH Merry Lipsky 100 PACH 201							0.20	3,733,33	497.78	4.231.11	3,768.89	
No.14 Display Source Source<	p.270167	M-01	Excervetor Mulcher - APE RDM34; SN: 73497 ECO 33L D Series HF	\$2,970.24		32,870,24	0 20	13,627 70	2,258.63	15.084.33	17,065.91	
No.2 Display Source Source </td <td></td>												
Write 197 BLC Apple No. That Wr. 1007/1002/0012 3500.00 X,000.0 0.20.12.0 2,177.2 41,111 41,44.4 Write XD Construction State State State XD Construction State State State XD Construction State State XD Construction State				224,070.24		324,870,24		148,894,37	20,421.52	170,319.86	154,650.38	
HP 2 202 Mean Num. Num. Num. Num. Num. Num. Num. Num.				1								
HP 209 Mean Number			1999 GMC Hydra Vac Truck VIN-1GDP7H1C7KI509682	35,000.00		36,000.00	0.20	16,333.33	2,177.76	18,511 11	10,400 80	
Hird 198 Weak Product State Mayber for Trol M. PRODUCTION PROVIDED AND ADDRESS 69,000 100,000 60,000 <	•	fV-02		185,000.00							67 165.56	
HT12 DBS frequency for the UNE SUBJECTION CONSTRUCT BUDDE 0 BUDDE 0 SUBJECT 0 <		rv-04	1998 Western Star Hudro Vic Truck VIII-2WLPCDKJOWC952223								56 533 33	
HT-02 X00 Lemmetanisation Line XM-2000LABCD1015 5000000 5000000 5000000 50000000 50000000 50000000 500000000 5000000000 50000000000 500000000000 5000000000000000000000000000000000000												
Hird NOB MACE And That WE JUGGARD (2010) TO CODE TO CODE Col Light of the structure o												
Int Prime Inter Name Inter Na												
UB-1 2000 Market Start Sta											32,977.76	
y 2016 11 (2010) 2010 (1012) 11 (2010) <th< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>4,711.11</td><td></td></th<>											4,711.11	
19718 197111 19718 19718 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>5,288.09</td><td>4,711,11</td><td></td></t<>										5,288.09	4,711,11	
9 2016 8624 9 2016 8623 9 2016						11,640,00	0.20	4,910.40	613.12	5,723.52	6,156.48	
Map - 6 2018 Mad / 2019 Launch Aging Launch Lager Str. (2018) 937, 460.00 931, 469.01 2.00 931, 568.00	7 2016 A	45-02	2016 Vermeet Mud System, MO240, 1Vh6120PpGF005485	23,760.00		23,760.00	0.20	9,620,60	1.626.24		12 312 95	
Con-16 2016 DBM Was / Res Vacuum Frank 177, 850.00 177, 850.00 2.00 2.00 0.00	day-18		2018 Aubel / 2019 Kennarth by the vec system: HUR-686-04-18 / 19800/AEX1/08997517	515,468.00			0.70	51 545 80			409,786,11	
Phote UND Dig Dig <thdig< th=""> <thdig< td="" th<=""><td>Dec 18</td><td></td><td>2018 Dilph Witch FX85 Vector/n Extended Trailer</td><td></td><td></td><td></td><td></td><td>12 785 00</td><td>13 424 25</td><td></td><td>101.640.75</td><td></td></thdig<></thdig<>	Dec 18		2018 Dilph Witch FX85 Vector/n Extended Trailer					12 785 00	13 424 25		101.640.75	
1,212,842,87 0,39 1,212,842,87 0,30 1,212,842,87 0,31,20,10 131,20,10 <t< td=""><td>2018</td><td></td><td>1993 IH Water Insta SAN 2NSFHA4R2PC071112-INVOICE CANCELLED AME 25, 2019</td><td></td><td></td><td></td><td></td><td></td><td></td><td>a</td><td>0.00</td><td></td></t<>	2018		1993 IH Water Insta SAN 2NSFHA4R2PC071112-INVOICE CANCELLED AME 25, 2019							a	0.00	
Mathe Color Status Figure 1 Figure 2 Status Figure 2				1								
Making 01 of 2000 Dedge hue, Wes MTMF12579173599 1 4000 00 0.78 9.533.3 97.11 7.04.44 6.544 01 of 2000 Dedge hue, Wes MTMF1257917341 30.000 00 30.000 00 20.000				1,218,848,60	. 0.00	1,216,846,80		331,042,00 (103.584.70	414,648,70	784,299,30	
OF 01 2020 Longer Law, Wei 307 MULECARGON 2021 Y (000 0) 16 000 0 2.7 (00.27 mm 1.5 mm	clet											
07-01 2004 holge has weis 207 Audit Audit 200 PTE 14,000 0 14,000 0 2.00 7,000 7 65,000 7 7,000 7 65,000 7 7,000 7		07-02	2000 Dodge Rass V84-18764F3367Y/178589	14,000,00		14,000.00 \$	0.20	0 533.33	471.11	7 404 44	6.585.55	
07:05 011 Delige Law, Vell-3077002.566176352 34:00:00 34:00:00 0.20 15:866 /r 21:53:8 71:02:20 40:07 17:04 2012 Delige Law, Vell-X707005012113 40:00:00 10:00:00 20:00:00 </td <td></td> <td>7.537.74</td> <td></td>											7.537.74	
Off-64 2012 Market Law, VW-SCHWOTTPOEL21915 C2,00.00 2,00.00 19,00.00 2,01.33 2,21.33 10,001 11 G4 2007 Ore-Structure VW-SCHWOTTPOEL21915 2,000.00 25,000.0 0.20 (4,666 str (4,666 str <t< td=""><td></td><td>77-05</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>		77-05										
11-14 1207 Ore Skurske VM-SOCKUMP7138324 100000 100000 0.20 66887 622.2 12.888.6 67.11 11-00 2011 GAC Line VM-SOCKUMP7138324 50.00.00 23.00.00 0.20 17.73.33 21.11 12.884.6 47.11 11-00 2014 GAC Sinv VM-SOCKUMP71262345129 50.00.00 36.00.00 0.20 17.73.33 21.11 22.644.64 27.95.13 21.11 22.644.64 27.95.13 21.11 22.644.64 27.97.13 21.97.17 17.73.13 21.11 22.644.64 27.97.17 17.97.1												
LT_G2 2010 Max Lines Weij-ETT/2007/BE3552 20000 0 21 0000 0 0.0												
LF-0P 2014 GMC STV VPH-STUTVOCCENSULT9 50.00.00 50.00.00 30.00 22.33.33 3.1111 22.444 21.352 LF-0P 2014 GMC STV VPH-STUTVOCCENSULT9 100.00 32.00.00 0.20 17.23.33 3.1111 22.444.44 20.07.73 3.1111 22.444.44 20.07.73 3.2111 22.444.44 20.07.73 3.2111 22.444.44 20.07.73 3.2111 22.444.44 20.07.73 3.2111 22.444.44 20.07.73 3.2111 22.444.44 20.07.73 3.2111 22.444.44 20.07.73 3.2111 22.444.44 20.07.73 3.2111 22.444.44 10.07.73 10.00 40.00 0.20 17.243.33 1.141.44 10.07.74 10.00 40.00 0.20 114.985.27 17.442.85 17.443.34 1.057.74 46.22 47.44 2.257.33 114.44 1.057.74 46.22 47.44 2.257.33 114.44 1.057.74 46.23 14.442.24 1.444.4 2.257.33 114.44 2.257.33 114.442.24 1.444.4 2.244.44 2.244.44												
Life 2014 ABC: Surg. Var. SETUTOPERSENTERS 350000 350000 17,723.33 2,244.4 20,077.0 17,720.34 Aver.15 VID.000 1,240.00 1,240.00 1,240.00 2,280.00 2,280.00 2,280.00 2,280.00 2,280.00 2,280.00 2,280.00 2,280.00 2,280.00 2,280.00 2,280.00 2,280.00 2,280.00 2,280.00 2,280.00 2,280.00 1,580.27 1,468.47 1,468.47											10.635.56	
Hyper-TB YF-02											23,556.56	
Non-11 200 Dodge MAR 500 VMI IRC/VMMIT SECSS464 77,99.9 17,99.9 0.20 7,99.9 1,97.9 1,90.20 17,99.9				38,600.00		38,000.00 (0.20	17,733.33	2,384.44	20.097.78	17,902.22	
Here: 11 20.0 Dodge NAE 5500 VM MC XMMIR Sec51:464 77,960.3 mg 7,960.3 mg 77,960.3 mg 15,620.2 mg	Apr-18 V	WT-01	1994 Ford LT9000 s/H: 1FDYU90TBA27545	12,980.00		12,980,00	0.20	1,296.00	1,360.00	2 656.60	10,303,20	
Inter PTO Am. 34 PTO Am. 34 </td <td>im-12</td> <td></td> <td>2018 Dodge RAM 5500 VIN IBCZWINFLMG354684</td> <td>77,669,38</td> <td></td> <td>77.009.30</td> <td>0 20</td> <td>7,708,94</td> <td>8 157 38</td> <td></td> <td>61,763.00</td> <td></td>	im-12		2018 Dodge RAM 5500 VIN IBCZWINFLMG354684	77,669,38		77.009.30	0 20	7,708,94	8 157 38		61,763.00	
Har Fib328 C (* Trader Humstmader 2,000,00 0.20 C13.31 134.4 1,057.78 6/22 C1-62 X010 Medians Car Tader VP3-MIX7623A0017029 5,000,00 3,000,00 0.20 2,283.33 311.11 2,644.44 2,864.45 C1-63 X010 Medians Car Tader VP3-MIX7623A0017029 5,000,00 3,000,00 0.20 4,030.33 311.11 2,644.44 2,864. C1-64 X010 Medians Car Tader VP3-MIX7623A0017029 6,000,00 14,000,00 0.20 4,633.33 871.11 1,464.44 2,864. F1-64 1989 Marcine Medians Trader VP3-MIX7622A0017022 6,000,00 2,000,00 0.20 4,533.33 871.11 7,464.44 2,265. F1-64 1997 (digt Press Trader VP3-F17X050104144 2,000,00 2,000,00 0.20 2,283.33 7,511.11 2,464.44 2,265. F1-65 1997 (digt Press Trader VP3-F17X050104144 2,000,00 2,000,00 2,000,00 2,000,00 2,000,00 2,000,00 2,000,00 2,000,00 2,000,00 2,000,00 2,000,00 1,000,01 2,000,0				317.649.34							178,006,40	
C1-01 Car Todia Humanuda 2,000,00 2,000,00 9,000 9,020 9,233,3 134.4 1,057,78 9,000 C1-02 2,000,00 3,000,00 3,000,00 3,000,00 3,000,00 1,000,00 2,000,00 2,000,00 2,000,00 1,000,00 1,000,00 1,000,00 2,000,00 2,000,00 2,000,00 2,000,00 1,000,00 1,000,00 1,000,00 2,000,00 2,000,00 2,000,00 1,000,00 1,000,00 1,000,00 1,000,00 1,000,00 1,000,00 1,000,00 1,000,00 1,000,00 1,000,00 1,000,00 1,000,00 1,000,00 <td>ere - #14</td> <td>364</td> <td></td>	ere - #14	364										
CT-G2 XDD Preference Gr Tealer, VM-24702782324037029 5 g00.00 5 g00.00 <td></td> <td></td> <td>Car Trailer Heersterrade</td> <td>2,000 m</td> <td></td> <td>2000 00</td> <td>0.20</td> <td>80.71</td> <td>124 44</td> <td>1 /057 74</td> <td>4G.22</td> <td></td>			Car Trailer Heersterrade	2,000 m		2000 00	0.20	80.71	124 44	1 /057 74	4G.22	
C1-03 2010 toward Cur Tude VM-24004 50218021811 3,000.00 3,000.00 1400.00												
Fride 1000 deniver New Table of Laboratory NEW MERGEOFTICO 77311 1,4000 0.20 6.4533 (2011) 111 1,40000 2,330000 2,33000 2,33000												
PT-02 1999 Mean Unity 1990 Aux Class Table VM 2990 ALCS 2580 (772) 1,000 0,000 2,000 0,000 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>												
IF-06 1984 News Trader Vel-32/02/035 7,000,00 7,000,00 0,20 2,200,00 2,200,											6,595.56	
F1-64 1001 Preas Trade: Presentative VMP 775457 \$ \$000 00 \$ \$ \$000 00 \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$											3.749 49	
IP-06 1997 (age Pros Timler VPL SH2270257V02141) 4.000.00 2.000.00 2.00 2.000.00									435.56	3,702.22	3,297.78	
IPFO 1997 (gate Phots Timler 'WF 1962/X05/YN21441 4 (000 4 (000 0.20 1,084,67 2,44.84 2,115.58 1,084,7 IF-06 3001 (Finite Pred Timler 'WF 1962/X05/YN21441 20,000.00 200 1,080,67 2,48.87 2,44.84 2,115.58 11,224,44 IF-07 Lead Timler 'WF 1962/X05/YN21440 4000.00 4,000,00 0.20 1,080,67 2,48.87 2,41.83 11,224,44 IF-07 Lead Timler 'WF 1962/X05/YN2144004 7,500,00 3,000,00 2,000 1,680,67 2,48.87 2,41.84 1,51.31 IF-07 Lead Timler 'WF 1962/X05/YN2144004 7,500,00 3,000,00 2,000 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00 16,000,00 2,200,00<									311.11	2,644,44	2,355.56	
P1-06 A015 Finder Tweis Trades Weis STRAID STR		T-05	1997 Eagle Pints Trailer VIN-1ENE27025V3021661	4,000.00		4,000,00	0.20	1,885,67			1.004.44	
P1-07 Lead Training Yinds Training Yinds (2002) (2004) (2004) 4 000 on 1 844 P1-02 2000 Willings (Web Winds) Training Yinds (2004) 7,500 a0 7,500 a0 7,500 a0 3,000 on 2,000 on 1,000 on 1,0		7-06	JOLS Failing Pints Trailer VIII-SETTA305603043048	26,000.00		28,000.00	0 20	12 133,33	1.617.78		12 248 80	
If 702 2000 Minimum Fight Multiple Trader Wei 24072278444149704 7 500.00 7,000.00 0.20 3,000.00 448.07 1,008.07 1,503.1 IF 702 1307 Logo Munor Fish Wei 12043278444149704 2,000.00 3,000.00 0.20 16,000.00 22,000.00 15,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 14,003.00 14,003.00 14,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.00 16,003.	,	T-07									1 844.44	
If Col 1997 User binner Phile Weind Trivier Vein 110(53)4415500502 2000 do 36 000 0 20 00 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 16 000 0 22 000 16 000 0 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 16 000 0 22 000 0 16 000 0 22 000 0 16 000 0 22 000 0 16 000 0 22 000 0 16 000 0 22 000 0 16 000 0 22 000 0 16 000 0 22 000 0 16 000 0 22 000 0 16 000 0 22 000 0 16 000 0 22 000 0 16 000 0 22 000 0 16 000 0 22 000 0 16 000 0 22 00 0 22 00 0		7-01	2005 Willindon Fifth Whend Trader Wet-2009725291644140704	7 500 m								
1761 2002 Self Tendy Veix-CR5423/26/20023/26/20025/26 2,000,00 2,000,00 0,200 1,002,00												
HT-01 Red Trade Managada 3.000 0 5.00.00 0 0.20 1 (rest), Marcing Managada 1.411 2 LT-01 2017 (rest), Marcing Managada 3.000 0 6.000 0 0.20 1 (rest), Marcing Marc												
ETG 1 2012 (rest Rew Endered Trader Weight(PL)25CW49970 6.000.00 2.000.00 0.000.00<												
Pb-G1 2013 T-Max Trivler VM-SAVGRUL2305391228 7,500 c0 7,500 c0 7,500 c0 2,500 c0 40,000 f 5,533 co M2 Extension												
Stationer, E12559											3,768.00	
14 Combinence 1 # 55259 Die Jaarg Kant loompersteine Lade 1 \$4931 Wer Toje 1* 15* hongenstel kond zwe Scale # 30025400 with mersel product stands 2,700,00 2,700,00 0.20 1,200,00 168,00 1,220,00 12,721	•	10-01	2013 T-New Troller VIN-16V08262105391326	7,500.00		7,500.00	0.20	3,500.00	486.67	3,969.67	3,533 33	
M. Zambanardi. #15259 Die Janag Kank beserperation Laden 154931 8,000,00 8,000,00 0.20 3,733,33 497.71 4,227.11 3,748.1 War Train V* 55* Insegnated land saw Social # 20023400 with mental product stands 2,700,00 0.20 1,200,00 142.00 1,274.11 1,748.10				L				L			I	
De hange field temperaterie hange field 1200 0.00 0.00 0.00 0.00 0.00 0.00 0.00			· · · · · · · · · · · · · · · · · · ·	147,000.00		142,000,00		04,204,67	8,435.64	75,102.22	84,807.76	
De hange field temperaterie hange field 1200 0.00 0.00 0.00 0.00 0.00 0.00 0.00	وبالمالة	went - 814256		1								
Way Truin 9" 3 5" hongonital band sear Sanial 5 3002 5400 with mana product stands 2,700,00 2,700,00 1,222				8,000.00		6,000,00	0.20	3,733,33	497.70	4,237,111	3,761.00	
			Way Train 9" = 36" horizontal band saw Serial # 30025400 with metal product stands	2,700.00		2,700.00	0.20		100.00		1,272.00	
											471.11	
				, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		·/~~···	V.20	1 40.07	V6 24		•n.n [
H:\NRPortb\Client\KGM\14151988_1.xlsx\JULY												

.1

Hins -Lonato Bichadul RULY 31, 2019 7

Lid Fisned An at

. |
 8 cm - 44 m		

H:\NRPortbl\Client\KGM\14151988_1.xlsx\JULY

Herefore for the State State State 100 0 Using State 0.00 400 0	Link	Descriptions	Opening	Additions	Discondu	Ciceina	Rate	Opening	Accumulated	Disponsie	Closing	UCC
Mail: Material (37 March 2014 2014) Mail: 19720 4.80.0 4.80.0 3.80.0		Hypertherm Power Maz 1000 63 Series Serial # 043189							8277			471.11
Mar. Mit. Mar. Mar. 1979 (1600) (100) <td></td> <td>Miller Millermark: \$50P Weiker Serial # MEDIO118N Sheck # 907300</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>2,120,00</td>		Miller Millermark: \$50P Weiker Serial # MEDIO118N Sheck # 907300										2,120,00
ACC 00 to higher band 1.000<		Miller Millermutic 350° Webler Serial # MD-400957% Stock # 907300	4,500,00									2,120,00
2 Carbon space 10.00		ACCO 60 ton Hydraulic Metal Press										1,177,78
Understand Use of the form		2 Cut Prempt Model 650										1,413,33
Central Cale A profile generative M3 11 (Public C 1920 and 1200 1 1000 1		Wilgen Tech AC Inventor 3000 West #20072	1,500.00									708.67
Milling Long Look on S / 1 1980 / 2000 / 2		Contractor Grade Pertable generation with 11 HO Honda GX 5500 Water										708.67
2010 Constrained for Market Market Strategy 1.500 mm 2.500 mm 0.50 1.1100 mm 1.500 mm												1, 177,78
Dies Constitution für 1400 1.500 0 1.500 0 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>1,177.79</td></td<>												1,177.79
DD1 Tex: Tex: Tex: Section 1.00000 1.0000 1.00000 <td></td>												
Die Lawer Anstall er Werter Machan 1,1000 1,0												1,177 76
198 Lands Marchalling marker with calasing 1.200 B					3 100 00	12,000 00						5,663.33
2006 66 Proc. of Start for Start (ST2) 1,56,50 0.20 7,70,50 7,50,50				•	3,100.00					- 1,595,69 -		0.00
2010 CX G81 LF metabolic MaTMe1 2.0000 3.0000												505.33
2000 Speer Sp												706.67
12* Second Pland 4,000 0 - 6,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 2,000 0 - 3,000 0 - 3,000 0 - 4,000 0 - 2,000 0 - 4,000 0												
Mr. Name Radi Constrain Constrain <thconstrain< th=""></thconstrain<>												1,546.00
27 Reame Read 7,000,0 0.00,0 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>2,120.00</td></td<>												2,120.00
27 Fearse Flast 1000 m 1000 m <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>2.626.67</td></t<>												2.626.67
27* Immunification 1.000 mm 0.000 mm <td></td> <td>3,297 78</td>												3,297 78
Augl Stade 1500.00 950.00 1200.00 1500.00 1000.00												3,768,89
00° Conception many data, paid on paid of applies. 11000.00 1700.00 0.00 7200.00 7200.00 7200.00 7200.00 7200.00 7200.00 7200.00 7200.00 7200.00 7200.00 7200.00 7200.00 </td <td></td> <td>4,240.00</td>												4,240.00
eff Conduct and Andhy for Ange of Mar. Luggen 110000 200 25000 2500												4,240.00
aff Contact in the Learning Link (Learning Systems (Learning											-,1	4,008.89
City Res. Meet That Rest 96 49 Available (1-44 Loss) (specific		and second and present present of the second of the second se										6,565.58
Call by any Ender CVERDERS CVERDERS 13, 823 (d) 13, 823 (d) <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>3,533.33</td></t<>												3,533.33
Under 2015 01 01 000 01 000 01 000 01 000 000 0												3,392.00
DD1 7 Dech Wech SDD2: VDI DWY-DDD2000000007 122 mil												68,774.54
Live broker Har No 27/073, trausmitter Let STL, III alkalve kerney only, 12 land Live Autor												70,293.90
Control Control <t< td=""><td></td><td></td><td>18,440.00</td><td></td><td></td><td></td><td></td><td>5,443,20</td><td></td><td></td><td></td><td>12,363.04</td></t<>			18,440.00					5,443,20				12,363.04
Jant Parsan Jant Janta Janta (Janta) Janta) Janta (Janta) Janta) Janta (Janta) Janta) Janta (Janta) Janta)		Their excents, error ward, 1/1/10,2' (calculational) (16, 32-12' are entities of excents A react 17. House	1 -	4,519,26		4,519.28	0.20	-	263.62		263.62	4,255.64
Entry Respiration 16488 19400 18280												
Fair-18 Little Little <thlittle< th=""> <thlittle< th=""> <thlittle< t<="" th=""><th></th><th></th><th>ROU Vehicles</th><th>ROU Equipment</th><th>Dette</th><th></th><th></th><th></th><th></th><th>Links Fauin</th><th></th><th></th></thlittle<></thlittle<></thlittle<>			ROU Vehicles	ROU Equipment	Dette					Links Fauin		
Fabriel V0,165.46 2.917.85 14,791.10 3,377.52 1,282.22 4,189.03 34, 34,192.22 4,189.03 34, 34,192.22 4,189.04 34, 34,192.22 3,807.01 3,377.52 1,282.22 4,189.01 3,377.52 1,282.22 4,189.01 3,377.52 1,282.22 4,189.01 3,377.52 1,282.22 4,189.01 3,377.52 1,282.22 4,189.01 3,377.52 1,282.22 4,189.01 3,377.52 1,282.22 4,180.07 377.52 1,282.22 4,180.07 377.52 1,282.22 4,180.07 377.52 1,282.22 4,180.07 377.52 1,282.22 4,180.07 377.52 1,282.22 4,180.07 397.72.52 1,282.22 4,180.17 397.72.52 1,282.22 1,382.24 101,171.02 22.01.24 397.72.52 1,282.22 1,382.84 101,171.02 22.01.24 397.72.52 1,282.22 1,382.84 101,171.02 22.01.24 397.72.22 1,382.84 101,171.02 22.01.24 397.72.22 1,382.84 101,171.02 22.01.24 22.01.24 22.01.24 <th< td=""><td></td><td>Entry Regelevel:</td><td></td><td></td><td></td><td>Hanny Squip.</td><td>Trucing</td><td>Vehiclas</td><td>Trailers</td><td></td><td>Dateit</td><td>Debit</td></th<>		Entry Regelevel:				Hanny Squip.	Trucing	Vehiclas	Trailers		Dateit	Debit
Mar-19 Link 10 Link 10 <thlink 10<="" th=""> <thlink 10<="" th=""> <thli< td=""><td></td><td></td><td></td><td></td><td>10140</td><td>Hinawy Equip. 16280</td><td>Tructos 18360</td><td>Vahicina 16380</td><td>Trailers 16380</td><td>16250</td><td>Dashit Bibliogi</td><td>Debit 99992</td></thli<></thlink></thlink>					10140	Hinawy Equip. 16280	Tructos 18360	Vahicina 16380	Trailers 16380	16250	Dashit Bibliogi	Debit 99992
Apr-18 1,1151-00 1,1152-00 1		Jan-18			10140 10,185.46	Hasky Equip. 14200 2,917.93	Truction 14:390 14:798.10	Vehicing 16309 1,377 52	Trailers 16380 1,262,22	4, 159.03	Dashit Bibliogi	Debit 199032 30,710.27
Lange 10		Jan-18 Fair-19			10140 10,105.46 10,105.46	Hindky Equip. 18280 2,917.93 2,917.93	Tracina 16300 14,798.10 14,798.10	Vehicins 16309 3,377.52 3,377.52	Trailans 16380 1,262,22 1,262,22	14250 4, 159.03 4, 159.03	Dashit Bibliogi	Debit 99992
Jahr 19 0,542,04 78,722,20 3,877,46 2,717,40 2,217,21 1,542,54 1,542,22 1,542,54 101,177,10 22,2 July 19 8,157,36 8,157,36 8,157,35 5,144,46 2,217,216 1,342,22 1,542,54 101,177,10 22, Auge 19 8,157,36 8,157,36 8,157,35 5,144,46 2,217,216 1,342,22 1,542,54 101,177,10 22, Sap-19 0,05,16 8,157,36 8,102,172 6,162,46 2,917,55 5,144,46 2,212,16 1,342,22 1,542,54 101,177,10 22, Sap-19 0,05,16 1,052,54 2,917,25 5,144,46 2,212,16 1,342,22 1,542,54 101,177,10 22, Sap-18 0,05,16 10,05,17 101,172,16 101,172,16 11,112,11 111,112,11 111,112,11 111,112,11 111,112,11 111,112,11 111,112,11 111,112,11 111,112,11 111,112,11 111,112,11 111,112,11 111,112,11 111,112,11 111,112,11 111,112,11		Jan-19 Fab-19 Mar-19			10140 10,105.46 10,105.46	Hindky Equip. 18280 2,917.93 2,917.93	Tracina 16300 14,798.10 14,798.10	Vehicins 16309 3,377.52 3,377.52	Trailans 16380 1,262,22 1,262,22	14250 4, 159.03 4, 159.03	Dabit 96601 -	Debit 199032 30,710.27
Jan 19 0,777.46 1,777.46 1,777.46 1,777.45 <		Jan-19 Fab-19 Mar-19			10140 10, 185, 46 10, 185, 46 10, 195, 46	Hanny Equip. 14240 2,917.93 2,917.93 2,917.83	Tracks 18390 14,798,10 14,796,10 14,796,10	Vehicina 16389 1,377 52 2,377 52 3,377 52 3,377 52	Trailens 16380 1,262.22 1,262.22 1,262.22	4, 159.03 4, 159.03 4, 159.03 4, 159.04	Dabit 68801 -	Debit 99900 36,710.27 36,710.28
Au-19 List 2 all 19/2000 List 2 all 19/2000 <thlist 19="" 2="" 20000<="" all="" th=""> List 2 all 19/20000<</thlist>		Jan-18 File 10 Jan-18 Age-18			10140 10, 195.45 10, 195.45 10, 195.46 10, 195.46	Histoy Equip. 16280 2,917.63 2,917.63 2,917.63 2,917.63 2,917.63	Tructas 14,798.10 14,796.10 14,796.10 14,798.10 15,396.10	Vehicins 16309 3,377 52 3,377 52 3,377 52 3,377 52	Trailan 16360 1,262,22 1,262,22 1,262,22 1,262,22	4, 159,03 4, 159,03 4, 159,03 4, 159,04 4, 309,67	Dabit 68001 - - -	Dubit 99990 36,710.27 36,710.28 36,710.27 37,490,91
Aug-19 E-g-19 D-g-19		Jan-19 File-19 Jan-19 Age-19 Mag-13	165ee		10140 10, 115, 45 10, 195, 45 10, 195, 45 10, 115, 45 10, 115, 45	Hanny Equip. 16200 2,017.03 2,017.03 2,017.03 2,017.03 2,017.03 2,017.03	Triectas 18340 14,798.10 14,798.10 14,798.10 15,396.10 14,048.10	Vehiclas 18209 3,377.52 3,377.52 1,377.52 3,377.52 3,377.52 3,377.52	Trailens 16380 1,262,22 1,262,22 1,262,22 1,262,22 1,262,22	11250 4, 159.03 4, 159.03 4, 159.03 4, 159.04 4, 509.67 4, 169.14	Dubit 68801 - - - - -	Debit 99900 36,710.27 36,710.26 36,710.27 37,460.91 36,870.37
5ap-10 Oc-16 Non-19 Dimpenden Total 2019 Jampen forward Jameer J. 2019 Strand		Jun-16 Fak-18 Jag-18 Ag-18 Jag-18 Jag-18		78,732.80	19140 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 3, 677, 45	Heatry Equip. 162400 2,917,50 2,917,85 2,917,85 2,917,85 2,917,85 2,917,85	Trucks 14390 14,798.10 14,798.10 14,798.10 15,386.10 14,048.10 (43,649.62)	Vehicins 1436 3,377 52 3,377 52 3,377 52 3,377 52 3,377 52 (3,614 53)	Trailan 14380 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22	4,159,03 4,159,03 4,159,03 4,159,04 4,509,67 4,169,14 (11,548,64)	Dabit 86801 - - -	Debit 99900 36,710.27 36,710.26 36,710.27 37,460.91 36,870.37
Sap-10 Dispension Dispension<		Jun-16 Fak-18 Jag-18 Ag-18 Jag-18 Jag-18		78,732.80	19140 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 3, 677, 45	Heatry Equip. 162400 2,917,50 2,917,85 2,917,85 2,917,85 2,917,85 2,917,85	Trucks 14390 14,798.10 14,798.10 14,798.10 15,386.10 14,048.10 (43,649.62)	Vehicins 1436 3,377 52 3,377 52 3,377 52 3,377 52 3,377 52 (3,614 53)	Trailan 14380 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22	4,159,03 4,159,03 4,159,03 4,159,04 4,509,67 4,169,14 (11,548,64)	Dabit 86801 - - -	Debit 99900 36,710.27 36,710.26 36,710.27 37,460.91 36,870.37
Out-18 Main-19 Data-18 Dispension Total 2019 Dispension Total 2019 15, 44, 42 Mainton Removed Journal 1, 2010 7,766.64 Patagrams Removed Journal 1, 2010 7,766.64 Dispension 7,766.64 Patagrams Removed Journal 1, 2010 7,766.64 Response 7,766.64 Response 1,66,657.08 Response 1,567.69 Response 1,567.69 Response 1,567.69 Response 1,567.69 Response 1,567.69 Response 1,567.69 Response Zon 10 1,557.69 Response Zon 10 1,557.64 Response Zon 10,71 1,557.69		Jan-18 File-18 Jan-18 Jan-19 Jan-18 Jan-18 Jan-18		78,732.80	19140 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 3, 677, 45	Heatry Equip. 162400 2,917,50 2,917,85 2,917,85 2,917,85 2,917,85 2,917,85	Trucks 14390 14,798.10 14,798.10 14,798.10 15,386.10 14,048.10 (43,649.62)	Vehicins 1436 3,377 52 3,377 52 3,377 52 3,377 52 3,377 52 (3,614 53)	Trailan 14380 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22	4,159,03 4,159,03 4,159,03 4,159,04 4,509,67 4,169,14 (11,548,64)	Dabit 86801 - - -	Debit 99900 30,710.27 35,710.20 35,710.27 37,400.91 36,870.37 - 50,754.58
Mar-19 De-19 De-19 Telul 2019 Telul 2019 Manyon Remove Function (num-PFAE) PresugNatio Deprecisions ROU Deprecisions Route Precisions ROU Deprecisions ROU Deprecisi		Jan-19 File-19 Jan-19 Agr-19 Jan-19 Jan-19 Jan-19 Agg-19		78,732.80	19140 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 3, 677, 45	Heatry Equip. 162400 2,917,50 2,917,85 2,917,85 2,917,85 2,917,85 2,917,85	Trucks 14390 14,798.10 14,798.10 14,798.10 15,386.10 14,048.10 (43,649.62)	Vehicins 1436 3,377 52 3,377 52 3,377 52 3,377 52 3,377 52 (3,614 53)	Trailan 14380 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22	4,159,03 4,159,03 4,159,03 4,159,04 4,509,67 4,169,14 (11,548,64)	Dabit 86801 - - -	Debit 99900 30,710.27 35,710.20 35,710.27 37,400.91 36,870.37 - 50,754.58
Date:18 Disponse Dete:19 Disponse Total 2010 15,148,42 112,764,62 53,487,23 20,425,53 76,720,52 4,459,34 11,199,21 147,464,44 166, Internit Honores 27,764,64 156,557,08 456,154,123 149,284,35 266,171,143 107,220,32 85,350,64 75,789,65 1.558,69 ROU Deprecision 1,558,69 1,558,69 1.558,69 1.558,69 1.558,69 Routine Deprecision 22,916,34 328,311,70 620,152,16 170,316,47 152,754,4 152,962,34 94,650,77 947,264,54 159		Jan-16 Fai-16 Jan-18 Jan-18 Jan-18 Jan-18 Jan-18 Jan-18 Jan-18 Jan-19 Jan-19 Jan-10		78,732.80	19140 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 3, 677, 45	Heatry Equip. 162400 2,917,50 2,917,85 2,917,85 2,917,85 2,917,85 2,917,85	Trucks 14390 14,798.10 14,798.10 14,798.10 15,386.10 14,048.10 (43,649.62)	Vehicins 1436 3,377 52 3,377 52 3,377 52 3,377 52 3,377 52 (3,614 53)	Trailan 14380 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22	4,159,03 4,159,03 4,159,03 4,159,04 4,509,67 4,169,14 (11,548,64)	Dabit 86801 - - -	Debit 99900 30,710.27 35,710.20 35,710.27 37,400.91 36,870.37 - 50,754.58
Dispenses Total 2019 15, 148, 42 172, 744,82 53, 487, 20 20, 425, 52 19, 529, 32 13, 488, 46 11, 199, 21 137, 944, 24 184, Insurant Innex Encode 7, 786, 64 156, 567, 08 450, 134, 62 149, 204, 25 286, 731, 14 107, 220, 32 05, 300, 64 75, 790, 85 1, 559, 80 Disponsition 11, 199, 244, 15 149, 244, 25 286, 731, 14 107, 220, 32 05, 300, 64 75, 790, 85 Results Deprecision 1, 559, 80 1, 559, 80 1, 559, 80 1, 559, 80 Results Deprecision 22, 914, 34 228, 311, 70 620, 132, 14 170, 314, 67 132, 716, 54 132, 714, 54 152, 192, 24 66, 450, 77 167, 204, 54 156, 90, 126 156, 90, 126		Jan-16 Fai-16 Jan-18 Jan-18 Jan-18 Jan-18 Jan-18 Jan-18 Jan-18 Jan-19 Jan-19 Jan-10		78,732.80	19140 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 3, 677, 45	Heatry Equip. 162400 2,917,50 2,917,85 2,917,85 2,917,85 2,917,85 2,917,85	Trucks 14390 14,798.10 14,798.10 14,798.10 15,386.10 14,048.10 (43,649.62)	Vehicins 1436 3,377 52 3,377 52 3,377 52 3,377 52 3,377 52 (3,614 53)	Trailan 14380 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22	4,159,03 4,159,03 4,159,03 4,159,04 4,509,67 4,169,14 (11,548,64)	Dabit 86801 - - -	Debit 99900 30,710.27 35,710.20 35,710.27 37,400.91 36,870.37 - 50,754.58
Disponde 15, 148, 42 172,744,82 52,877,23 20,425,53 19,575,34 11,198,24 11,198,24 154, Insurants Removed Journal 1, 2010 7,756,64 155,567,08 456,151,143 149,264,35 286,731,143 107,220,32 05,300,64 75,700,85 1,556,80 ROU Departicities 1,556,80 1,556,80 1,556,80 1,556,80 1,556,80 Routines Street Law 22,814,34 228,311,70 620,132,18 197,276,84 152,194,46 75,190,20		Jan-18 File-18 Jan-18 Jan-18 Jan-18 Jan-18 Jan-18 Jan-18 Gap-10 Oco-18		78,732.80	19140 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 3, 677, 45	Heatry Equip. 162400 2,917,50 2,917,85 2,917,85 2,917,85 2,917,85 2,917,85	Trucks 14390 14,798.10 14,798.10 14,798.10 15,386.10 14,048.10 (43,649.62)	Vehicins 1436 3,377 52 3,377 52 3,377 52 3,377 52 3,377 52 (3,614 53)	Trailan 14380 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22	11250 4, 159.03 4, 159.03 4, 159.03 4, 159.04 4, 159.04 4, 159.04 4, 169.14 (11, 548.04)	Dabit 86801 - - -	Debit 99900 30,710.27 35,710.20 35,710.27 37,400.91 36,870.37 - 50,754.58
Total 2019 15,146,42 172,744,42 12,47,23 20,425,53 76,993.56 77,746,45 148,991.56 17,746,45 148,991.56 77,746,45 148,991.56 77,746,45 148,991.56 77,746,45 148,991.56 77,746,45 148,991.56 77,746,45 78,751.16 107,220.32 87,250.56 77,746,45 78,751.16 107,220.32 87,250.56 77,746,45 75,740,45 75,92,24 96,900,77 967,900,454 760,716,71 967,900,454 760,716,71 967,900,454 760,716,71 967,900,454 760,716,71 967,900,454 760,716,71 967,900,454 760,716,71 967,900,454 760,716,71 967,900,454 760,716,71 967,900,454 760,716,71		Jun-16 Fain-16 Jain-18 Jain-18 Jain-18 Jain-18 Jain-18 Jain-19 Say-10 Oct-18 Jain-19		78,732.80	19140 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 3, 677, 45	Heatry Equip. 162400 2,917,50 2,917,85 2,917,85 2,917,85 2,917,85 2,917,85	Trucks 14390 14,798.10 14,798.10 14,798.10 15,386.10 14,048.10 (43,649.62)	Vehicins 1436 3,377 52 3,377 52 3,377 52 3,377 52 3,377 52 (3,614 53)	Trailan 14380 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22	11250 4, 159.03 4, 159.03 4, 159.03 4, 159.04 4, 159.04 4, 159.04 4, 169.14 (11, 548.04)	Dabit 86801 - - -	Debit 99900 30,710.27 35,710.20 35,710.27 37,400.91 36,870.37 - 50,754.58
Balancin fermeri J. 2019 7,784.04 155,557.08 455,134.03 149,84-35 246,731.14 107,220.32 85,205.64 75,785.55 Dayroods ROL Depresation (am-PP&E) Instandato Depresation Balances 2019 22,515,25 22,217,124 107,220.32 85,205.64 75,785.55 1,559,80 Rolesce 2019 22,515,257 282,217,70 220,122.19 170,3154,127,1554 122,745,45 75,902.24 95,600.77 167,206.64 156,		Jan-19 Fai-10 Jan-10 Jan-10 Jan-10 Jan-10 Gan-10 Gan-10 Jan-10 Dan-10		78,732.80	19140 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 3, 677, 45	Heatry Equip. 162400 2,917,50 2,917,85 2,917,85 2,917,85 2,917,85 2,917,85	Trucks 14390 14,798.10 14,798.10 14,798.10 15,386.10 14,048.10 (43,649.62)	Vehicins 1436 3,377 52 3,377 52 3,377 52 3,377 52 3,377 52 (3,614 53)	Trailan 14380 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22	11250 4, 159.03 4, 159.03 4, 159.03 4, 159.04 4, 159.04 4, 159.04 4, 169.14 (11, 548.04)	Dabit 86801 - - -	Debit 99900 30,710.27 35,710.20 35,710.27 37,400.91 36,870.37 - 50,754.58
Biological Strenger J (2019 7,756 64 156,557 68 456,131 82 149,864.35 266,721 1.6 107,220.32 80,200.66 71,766,65 Darponitic ROU Operation Strengeline 1,756,657 68 456,131 82 149,864.35 266,721 1.6 107,220.32 80,200.66 71,766,65 ROU Operation Strengeline 1,559,692 1,559,692 1,559,692 1,559,692 Rougest 2019 22,915,34 229,912,19 179,316,47 152,716,34 152,902,24 66,690,77 167,206,64 156,		Jan-19 Fai-10 Jan-10 Jan-10 Jan-10 Jan-10 Gan-10 Gan-10 Jan-10 Dan-10		78,732.80	19140 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 10, 195, 45 3, 677, 45	Heatry Equip. 162400 2,917,50 2,917,85 2,917,85 2,917,85 2,917,85 2,917,85	Trucks 14390 14,798.10 14,798.10 14,798.10 15,386.10 14,048.10 (43,649.62)	Vehicins 1436 3,377 52 3,377 52 3,377 52 3,377 52 3,377 52 (3,614 53)	Trailan 14380 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22 1,262.22	11250 4, 159.03 4, 159.03 4, 159.03 4, 159.04 4, 159.04 4, 159.04 4, 169.14 (11, 548.04)	Dabit 86801 - - -	Debit 99900 30,710.27 35,710.20 35,710.27 37,400.91 36,870.37 - 50,754.58
Disponsion RPU Disponsion (sen-PPEE) Insangtive Depressione Balances 2019 22,014.34 228,311.70 220,122.19 170,314,47 342,776.54 123,714.54 35,102.34 86,403,77 547,204.04 155		Jan-18 Far 10 Jan-18 Ag-19 Ag-19 Ag-19 Ag-19 Sag-10 Co-10 Sag-10 Co-10 Sag-10 Co-10 Sag-10 Co-10 Sag	16590 8.962.04 8.157.36	78,732,80 63,621,72	19980 10, 185 46 10, 185 46 10, 185 46 10, 185 46 10, 185 46 10, 955 46 3, 877 49 8, 142 46	Heavy Equip. 14200 2,917 60 2,917 60 2,917 63 2,917 63 2,917 63 2,917 63 2,917 53	Tautas 19360 14,798.10 14,798.10 15,386.10 14,948.10 (43,649.62) 5,148.48	Versian 1839 1,377 52 1,377 52 1,377 52 3,377 52 3,377 52 2,874 53) 2,2712 18	Traban 18980 1,282,22 1,282,22 1,282,22 1,282,22 1,282,22 1,282,22 1,282,22 1,282,22	4159.03 4,159.03 4,159.04 4,590.04 4,590.04 4,500.67 4,169.14 (11,548.04) 1,542.04	Dabit 99901 - - - - - - - - - - - - -	Debe 59902 36,710,27 36,710,27 36,710,27 37,400,81 39,870,37 50,754,58 22,206,22 - -
Disposalis 1,569,80 RGU Depresident (mn-PPEE) Interglike Depresident Balappes 2019 22,01.34 228,311.70 620,122.19 170,314,47 342,776,54 122,716,54 152,716,54 152,716,54 153,757,542 153,717,54 153,717,56 150,71		Jan-18 Fini 10 Jan-18 Ag-18 Mg-19 Jan-18 Jan-18 Ag-19 Ag-19 Ag-19 Cos-18 Mg-19 Dos-18 Mg-19 Dos-18 Mg-19 Dos-19 Dispensed	16590 8.962.04 8.157.36	78,732,80 63,621,72	19980 10, 185 46 10, 185 46 10, 185 46 10, 185 46 10, 185 46 10, 955 46 3, 877 49 8, 142 46	Heavy Equip. 14200 2,917 60 2,917 60 2,917 63 2,917 63 2,917 63 2,917 63 2,917 53	Tautas 19360 14,798.10 14,798.10 15,386.10 14,948.10 (43,649.62) 5,148.48	Versian 1839 1,377 52 1,377 52 1,377 52 3,377 52 3,377 52 2,874 53) 2,2712 18	Traban 18980 1,282,22 1,282,22 1,282,22 1,282,22 1,282,22 1,282,22 1,282,22 1,282,22	4159.03 4,159.03 4,159.04 4,590.04 4,590.04 4,500.67 4,169.14 (11,548.04) 1,542.04	Dabit 99901 - - - - - - - - - - - - -	Debit 99900 30,710.27 35,710.29 35,710.27 37,400.91 36,870.37 - 50,754.58
ROU Depresation (mm-PP4E) Integrated Depresition Balance 2010 22,014,36 220,311,70 620,132,10 170,314,47 342,776,54 112,744,56 75,192,24 95,493,77 147,264,54 116		Jan-16 Fai-19 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Ga-10 Dim-10 Dim-10 Dim-10 Dim-10 Dim-10 Dim-10 Dim-10 Dim-10 Dim-10 Dim-10 Dim-10	16390 8.962.04 8.157.35 15.462.42	16865 78,732.80 63,621.72 172,754.62	1940 10, 165 45 10, 165 45 10, 165 45 10, 165 45 10, 165 45 0, 165 45 0, 165 45 0, 162 45 0, 162 45	Henry Equip. 12353 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 53	Teacha 19390 14,798,10 14,798,10 14,798,10 14,798,10 15,286,10 15,286,10 (43,494,02 5,148,40	Versian 1430 1,377 52 1,377 52 1,377 52 3,377 52 3,377 52 (8,614 53) 2,2712 18	Trabum 14580 1.262 22 1.262 22 1.262 22 1.262 22 1.262 22 1.262 22 1.262 22	4220 4, 159,03 4, 159,03 4, 159,04 4, 159,04 4, 159,04 4, 159,04 (11,546,04) 1,542,04 11,542,04	Dabit 99901 - - - - - - - - - - - - -	Debe 59902 36,710,27 36,710,27 36,710,27 37,400,81 39,870,37 50,754,58 22,206,22 - -
Balance 2019 22,015.36 228,011.70 220,122,10 170,316.07 M0,778.54 112,715.46 15,102.24 06,000.77 107,206.64 150		Jun-16 Fish-10 Jun-10 Jun-10 Jun-10 Jun-10 Jun-10 Jun-10 Sep-10 Oc-10 Sep-10 Oc-10 Sep-10 Oc-10 Sep-10 Oc-10 Sep-10 Oc-10 Sep-10 Oc-10 Sep-10 Oc-10 Sep-10 Disponds	16390 8.962.04 8.157.35 15.462.42	16865 78,732.80 63,621.72 172,754.62	1940 10, 165 45 10, 165 45 10, 165 45 10, 165 45 10, 165 45 0, 165 45 0, 165 45 0, 162 45 0, 162 45	Henry Equip. 12353 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 53	Teacha 19390 14,798,10 14,798,10 14,798,10 14,798,10 15,286,10 15,286,10 (43,494,02 5,148,40	Versian 1430 1,377 52 1,377 52 1,377 52 3,377 52 3,377 52 (8,614 53) 2,2712 18	Traken 16360 1.262.22 1.262.22 1.262.22 1.262.22 1.262.22 1.262.22 1.262.22 1.262.22	4220 4.159.03 4.159.03 4.159.04 4.300.67 4.160.14 (11,340.04) 1.582.04 11,160.61	Dabit 99901 - - - - - - - - - - - - -	Debe 59902 36,710,27 36,710,27 36,710,27 37,400,81 39,870,37 50,754,58 22,206,22 - -
		Jan-16 Fai-19 Ag-19 Ag-19 Jan-19	16390 8.962.04 8.157.35 15.462.42	16865 78,732.80 63,621.72 172,754.62	1940 10, 165 45 10, 165 45 10, 165 45 10, 165 45 10, 165 45 0, 165 45 0, 165 45 0, 162 45 0, 162 45	Henry Equip. 12353 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 53	Teacha 19390 14,798,10 14,798,10 14,798,10 14,798,10 15,286,10 15,286,10 (43,494,02 5,148,40	Versian 1430 1,377 52 1,377 52 1,377 52 3,377 52 3,377 52 (8,614 53) 2,2712 18	Traken 16360 1.262.22 1.262.22 1.262.22 1.262.22 1.262.22 1.262.22 1.262.22 1.262.22	4220 4.159.03 4.159.03 4.159.04 4.300.67 4.160.14 (11,340.04) 1.582.04 11,160.61	Dabit 99901 - - - - - - - - - - - - -	Debe 59902 36,710,27 36,710,27 36,710,27 37,400,81 39,870,37 50,754,58 22,206,22 - -
		Jan-16 Fai-19 Ag-19 Ag-19 Jan-19	16390 8.962.04 8.157.35 15.462.42	16865 78,732.80 63,621.72 172,754.62	1940 10, 165 45 10, 165 45 10, 165 45 10, 165 45 10, 165 45 0, 165 45 0, 165 45 0, 162 45 0, 162 45	Henry Equip. 12353 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 63 2,917 53	Teacha 19390 14,798,10 14,798,10 14,798,10 14,798,10 15,286,10 15,286,10 (43,494,02 5,148,40	Versian 1430 1,377 52 1,377 52 1,377 52 3,377 52 3,377 52 (8,614 53) 2,2712 18	Traken 16360 1.262.22 1.262.22 1.262.22 1.262.22 1.262.22 1.262.22 1.262.22 1.262.22	4220 4.159.03 4.159.03 4.159.04 4.300.67 4.160.14 (11,340.04) 1.582.04 11,160.61	Dabit 99901 - - - - - - - - - - - - -	Debe 59902 36,710,27 36,710,27 36,710,27 37,400,81 39,870,37 50,754,58 22,206,22 - -
		Jan-16 Fai-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Da-10 Man-10 Da-10 Man-10 Da-10 Man-10 Da-10 Man-10 Da-10 Man-10 Da-10 Man-10 Da-10 Man-10 Da-10 Man-10 Da-10 Man-10 Da-10 Man-10 Da-10 Man-10 Da-10 Man-10 Da-10 Man-10 Da-10 Man-10 Ma	16200 8.862,04 8,157.36 15,148,42 7,788,64	78,732,80 83,021,72 172,754,83 155,557 08	1990 0, 195 46 10, 195 46 10, 195 46 10, 195 46 3, 977 49 6, 142 46 43, 497 23 456, 134 83	Henry Equip. 12353 2,917 85 2,917 85	Tautas (1990) 14,788,10 14,788,10 14,788,10 14,788,10 14,788,10 14,548,10 (43,649,62) 5,148,48 2,148,48 2,148,48 2,148,48 2,148,48	Vendas 1839 1,377 52 1,377 52 1,377 52 3,377 52 3,377 52 3,377 52 (4,614,53) 2,212 16 10,212 16	Traken 14300 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22	4220 4.159.03 4.159.03 4.159.04 4.300.07 4.169.14 (11,346.04) 1,542.04 11,106.41 1,542.04	Dabit 66071 661724.04 10117210 -	Desiel 99900 38,710,27 38,710,27 37,480,81 39,470,37 59,754,58 22,295,52 4 484,872,71
Total Accus, Dependetion		Jun-16 Fai-16 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Dispersion Sep-10 Co-16 Men-19 Dispersion Total 3919 Meaning Server(January 1, 2010 Dispersion Meaning Server(January 1, 2010 Meaning Server(January	16200 8.862,04 8,157.36 15,148,42 7,788,64	78,732,80 83,021,72 172,754,83 155,557 08	1990 0, 195 46 10, 195 46 10, 195 46 10, 195 46 3, 977 49 6, 142 46 43, 497 23 456, 134 83	Henry Equip. 12353 2,917 85 2,917 85	Tautas (1990) 14,788,10 14,788,10 14,788,10 14,788,10 14,788,10 14,548,10 (43,649,62) 5,148,48 2,148,48 2,148,48 2,148,48 2,148,48	Vendas 1839 1,377 52 1,377 52 1,377 52 3,377 52 3,377 52 3,377 52 (4,614,53) 2,212 16 10,212 16	Traken 14300 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22	4220 4.159.03 4.159.03 4.159.04 4.300.07 4.169.14 (11,346.04) 1,542.04 11,106.41 1,542.04	Dabit 66071 661724.04 10117210 -	Debe 59902 36,710,27 36,710,27 36,710,27 37,400,81 39,870,37 50,754,58 22,206,22 - -
1,540,842 ps		Jun-16 Fai-16 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Jan-10 Dispersion Sep-10 Co-16 Men-19 Dispersion Total 3919 Meaning Server(January 1, 2010 Dispersion Meaning Server(January 1, 2010 Meaning Server(January	16200 8.862,04 8,157.36 15,148,42 7,788,64	78,732,80 83,021,72 172,754,83 155,557 08	1990 0, 195 46 10, 195 46 10, 195 46 10, 195 46 3, 977 49 6, 142 46 43, 497 23 456, 134 83	Henry Equip. 12353 2,917 85 2,917 85	Tautas (1990) 14,788,10 14,788,10 14,788,10 14,788,10 14,788,10 14,548,10 (43,649,62) 5,148,48 2,148,48 2,148,48 2,148,48 2,148,48	Vendas 1839 1,377 52 1,377 52 1,377 52 3,377 52 3,377 52 3,377 52 (4,614,53) 2,212 16 10,212 16	Traken 14300 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22	4220 4.159.03 4.159.03 4.159.04 4.300.07 4.169.14 (11,346.04) 1,542.04 11,106.41 1,542.04	Dabit 66071 661724.04 10117210 -	Desiel 99900 38,710,27 38,710,27 37,480,81 39,470,37 59,754,58 22,295,52 4 4 84,872,71
		Jun-16 Fai-10 Jag-10 Jan-10 Jan-10 Jan-10 Jan-10 Jag-10 Sag-10 Oc-10 Sag-10 Oc-10 Sag-10 Oc-10 Sag-10 Despends Total 2919 Islamint forward January 1, 2910 Disponsition Total 2019 Islamint forward January 1, 2910 Disponsition (Int-PPLE) Islamint forward January 1, 2910 Disponsition (Int-PPLE) Islamint forward January 1, 2910 Disponsition Saturd Saturd January 1, 2910 Disponsition	16200 8.862,04 8,157.36 15,148,42 7,788,64	78,732,80 83,021,72 172,754,83 155,557 08	1990 0, 195 46 10, 195 46 10, 195 46 10, 195 46 3, 977 49 6, 142 46 43, 497 23 456, 134 83	Henry Equip. 12353 2,917 85 2,917 85	Tautas (1990) 14,788,10 14,788,10 14,788,10 14,788,10 14,788,10 14,548,10 (43,649,62) 5,148,48 2,148,48 2,148,48 2,148,48 2,148,48	Vendas 1839 1,377 52 1,377 52 1,377 52 3,377 52 3,377 52 3,377 52 (4,614,53) 2,212 16 10,212 16	Traken 14300 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22 1,202 22	4220 4.159.03 4.159.03 4.159.04 4.50.07 4.159.04 (11,59.04 (11,59.04) 1.592.04 11,199.41 75,769.45 7,526.69	Dabit 66071 661724.04 10117210 -	Desiel 99900 38,710,27 38,710,27 37,480,81 39,470,37 59,754,58 22,295,52 4 4 84,872,71

7

. 1

1.64

Cros-man Direct Un Fixed Americ Schedule Ar at JULY 31, 2019

Page 2 of 2

SCHEDULE C

RESIDUAL ASSETS

- Cash balance held by, anticipated to be received by and payable to Energold Drilling Corp. and Bertram Drilling Corp. at the Effective Date (subject to RBC's security interest)
- Accounts Receivable from third parties held by, anticipated to be received by and payable to Bertram Drilling Corp. at the Effective Date (subject to RBC's security interest)
 - Publicly listed shares of the following:
 - o Impact Silver
 - o Avrupa Minerals
 - o Everton Resources
 - o Pinestar Gold
 - o Voyageur Minerals
 - o Nortec

٠

- o Candente Copper
- Warrants of the following:
 - o Candente Copper
 - o Nortec
 - o Voyageur Minerals
 - o Avrupa Minerals
- Note with a principal amount of 3.1 million British pounds by Dando Drilling International
- Inventory and PP&E held by Energold Drilling Corp.
- Trademark, IP, rights, and licences held by Energold Drilling Corp.
- Inventory held by Energold Drilling Corp
- Inventory and PP&E held by EGD Services Ltd.
- Real estate held by Bertram Drilling Corp.
- Shares held by Energold in Bertram Drilling Corp.
- Choses in action and other intangibles of Energold, Cros-Man and Bertram Drilling Corp.
- Secured Rescue Note from Energold de Mexico, S.A. de C.V. to Energold
- Shares in Energold Drilling Peru held by Energold Drilling Corp.
- Any and all other assets of any kind held by the debtors and guarantors of the Note Purchase Agreement

SCHEDULE D

DEFINITIONS

-E

"Administration Charge" has the meaning ascribed to that term in the Initial Order;

"Administrative Reserve" means a Cash reserve, in the amount of the Estimated Administrative Reserve Costs to be deposited by the Agent into the Administrative Reserve Account for the purpose of paying the Administrative Reserve Costs;

"Administrative Reserve Account" means a segregated interest-bearing trust account established by BLG to hold the Administrative Reserve;

"Administrative Reserve Costs" means the reasonable fees and disbursements of the Monitor, the CRO, counsel to the Monitor and counsel to the Petitioners which are related to the CCAA Proceedings as secured by the Administration Charge;

"Affected Noteholder Claim" means the proportionate amount of the Claim of a Noteholder to the Credit Bid Amount;

"Affected Noteholder Class" means the creditors holding Affected Noteholder Claims;

"Affected Noteholder Pro Rata Share" means, in respect of any Affected Noteholder Claim, the proportionate share of the Affected Noteholder Claim held by it of all Affected Noteholder Claims held by all Affected Noholders;

"Agent" has the meaning ascribed to that term in the Recitals;

"Applicable Law" means, with respect to any Person, property, transaction, event or other matter, any Law relating or applicable to such Person, property, transaction, event or other matter, including, where appropriate, any interpretation of the Law (or any part thereof) by any Person, court or tribunal having jurisdiction over it, or charged with its administration or interpretation;

"BDI Credit Bid Amount" has the meaning ascribed to that term in the Recitals;

"Bertram" has the meaning ascribed to that term in the Recitals;

"BLG" means Borden Ladner Gervais LLP, legal counsel to the Petitioners and the Energold Group;

"Business Day" means any day, other than a Saturday, or a Sunday or a statutory or civic holiday, on which banks are generally open for business in Vancouver, British Columbia;

"Cash" means cash, certificates of deposit, bank deposits, commercial paper, treasury bills and other cash equivalents;

"Canadian Holdcos" means the Canadian corporations the formation of which is described in Section 8.2;

"CCAA" has the meaning ascribed to that term in the Recitals;

"CCAA Proceedings" means the proceedings commenced by the Petitioners under the CCAA as contemplated by the Initial Order;

"Charges" has the meaning ascribed to that term in the Initial Order;

...]

"Claim" means:

- (a) any right or claim, including any Tax Claim, of any Person that may be asserted or made in whole or in part against any of the Petitioners, in any capacity, whether or not asserted or made, in connection with any indebtedness, liability or obligation of any kind whatsoever of any of the Petitioners, and any interest accrued thereon or costs payable in respect thereof, in existence on the Filing Date, or which is based on an event, fact, act or omission which occurred in whole or in part prior to the Filing Date, whether at law or in equity, including by reason of the commission of a tort (intentional or unintentional), by reason of any breach of contract or other agreement (oral or written), by reason of any breach of duty (including, any legal, statutory, equitable or fiduciary duty) or by reason of any equity interest, right of ownership of or title to property or assets or right to a trust or deemed trust (statutory, express, implied, resulting, constructive or otherwise), and together with any security enforcement costs or legal costs associated with any such claim, and whether or not any indebtedness, liability or obligation is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured, unsecured, perfected, unperfected, present or future, known or unknown, by guarantee, warranty, surety or otherwise, and whether or not any right or claim is executory or anticipatory in nature, including any claim arising from or caused by the termination, disclaimer, resiliation, assignment or repudiation by any of the Petitioners of any contract, lease or other agreement, whether written or oral, any claim made or asserted against any of the Petitioners through any affiliate, subsidiary, associated or related person, or any right or ability of any Person to advance a claim for an accounting, reconciliation, contribution, indemnity, restitution or otherwise with respect to any matter, grievance, action (including any class action or proceeding before an administrative tribunal), cause or chose in action, whether existing at present or commenced in the future, and including any other claims that would have been claims provable in bankruptcy had any of the Petitioners become bankrupt on the Filing Date; and
- (b) any right or claim of any Person against any of the Petitioners in connection with any indebtedness, liability or obligation of any kind whatsoever owed by any of the Petitioners to such Person arising out of the restructuring, disclaimer, resiliation, termination or beach by any of the Petitioners on or after the Filing Date of any contract, lease, warranty obligation or other agreement whether written or oral;

"Court" has the meaning ascribed to that term in the Recitals;

"Credit Bid" means the winning bid of the Agent on behalf of the Noteholders pursuant to the SSP.

"Credit Bid Amount" means has the meaning ascribed to that term in the Recitals;

"Creditors' Meeting" means the meeting or meetings of the Affected Noteholders called for the purpose of considering and voting in respect of this Plan as described in the Meeting Order;

"CRO" has the meaning ascribed to that term in the Initial Order;

"Cros-Man" has the meaning ascribed to that term in the Recitals;

"Cros-Man Assets" has the meaning ascribed to that term in the Recitals;

"Cros-Man Asset Purchase Agreement" has the meaning ascribed to that term in the Recitals;

"Cros-Man Credit Bid Amount" has the meaning ascribed to that term in the Recitals;

"Crown" means Her Majesty in right of Canada or a province of Canada;

"Crown Priority Claim" means any Claim of the Crown, for all amounts that were outstanding at the Filing Date and are of a kind that could be subject to a demand under:

- (a) subsection 224(1.2) of the ITA;
- (b) any provision of the Canada Pension Plan or of the Employment Insurance Act that refers to subsection 224(1.2) of the ITA and provides for the collection of a contribution, as defined in the Canada Pension Plan, an employee's premium, or employer's premium, as defined in the Employment Insurance Act, or a premium under Part VII.1 of that Act and of any related interest, penalties or other amounts;
- (c) any provision of provincial legislation that has a similar purpose to subsection 224(1.2) of the ITA, or that refers to that subsection, to the extent that it provides for the collection of a sum, and of any related interest, penalties or other amounts, where the sum:
 - (i) has been withheld or deducted by a person from a payment to another person and is in respect of a tax similar in nature to the income tax imposed on individuals under the ITA; or
 - (ii) is of the same nature as a contribution under the Canada Pension Plan if the province is a "province providing a comprehensive pension plan" as defined in subsection 3(1) of the Canada Pension Plan and the provincial legislation establishes a "provincial pension plan" as defined in that subsection;

"D&O Charge" has the meaning ascribed to it in the initial Order;

"DIP Loan" means the amounts advanced and outstanding under the credit facility from Energold DIP Lender, LLC to the Petitioners pursuant to the Initial Order;

"Director" means anyone who is or was, or may be deemed to be or have been, whether by statute, operation of law or otherwise, a director or de facto director of any of the Petitioners;

"EDC" has the meaning ascribed to that term in the Recitals;

"Effective Time" means 12:01 a.m. on the Implementation Date (or such other time as the Petitioners, the Monitor and the Agent may agree);

"EMEA Credit Bid Amount" has the meaning ascribed to that term in the Recitals;

"Employee Priority Claims" means, with respect to current or former employees of the Petitioners, the following claims:

- (a) Claims of the Petitioners' employees and former employees equal to the amounts that such employees and former employees would have been qualified to receive under paragraph 136(1)(d) of the *Bankruptcy and Insolvency Act* (Canada) if the Petitioners had become bankrupt on the Filing Date; and
- (b) Claims of the Petitioners' employees and former employees for wages, salaries, commissions or compensation for services rendered by them after the Filing Date and on or before the date of the Sanction Order, together with, in the case of travelling salespersons, disbursements properly incurred by them in and about the Petitioners' business during the same period;

"Energoid" has the meaning ascribed to that term in the Recitals;

"Energold Group" has the meaning ascribed to that term in the Recitals;

"Estimated Administrative Reserve Costs" means the Petitioners' good faith best estimate, based on a detailed calculation of the Administrative Reserve Costs in accordance with the terms of the Purchase Agreements.

"Excise Tax Act" means the Excise Tax Act, R.S.C. 1985, c.E-15, as amended and any regulations thereunder;

"Excluded Claim"

- (a) any Claims secured by any of the Charges;
- (b) any Claims that cannot be compromised pursuant to subsection 19(2) of the CCAA;
- (c) all Secured Priority Claims;
- (d) all Unaffected Noteholder Claims;
- (e) all unsecured Claims against the Petitioners;
- (f) any Priority Claims;
- (g) any Post-Filing Claims; and
- (h) any Claim entitled to the benefit of any applicable insurance policy, excluding any such Claim or portion thereof that is directly recoverable as against an Applicant;

"Excluded Creditor" means a Person who has an Excluded Claim, but only in respect of and to the extent of such Excluded Claim;

"Filing Date" means September 13, 2019;

"Financial Advisor" has the meaning ascribed to that term in the Recitals;

"Financial Advisor Charge" has the meaning ascribed to that term in the Initial Order;

- 29 -

- 30 -

"FTI" has the meaning ascribed to that term in the Recitals;

"Governmental Entity" means any government, regulatory authority, governmental department, agency, commission, bureau, official, minister, Crown corporation, court, board, tribunal or dispute settlement panel or other law, rule or regulation-making organization or entity:

- (a) having or purporting to have jurisdiction on behalf of any nation, province, territory or state or any other geographic or political subdivision of any of them; or
- (b) exercising, or entitled or purporting to exercise any administrative, executive, judicial, legislative, policy, regulatory or taxing authority or power;

"Implementation Date" means the Business Day on which this Plan becomes effective, which shall be the day indicated on the certificate which the Monitor has filed with the Court contemplated in Section 10.7 hereof;

"Initial Order" has the meaning ascribed to that term in the Recitals;

"Intercompany Working Capital Obligations" means any obligations by any of the Energold Group against any of the other Energold Group in relation to accounts receivable and accounts payable listed in the respective company's books and records as of the date of the Initial Order, exclusive of the obligations by Energold against Energold de Mexico, S.A. de C.V. pursuant to the Secured Rescue Note.

"Interim Lender's Charge" has the meaning ascribed to that term in the Initial Order;

"ITA" means the *Income Tax Act*, R.S.C. 1985, c.1 (5th Supp.), as amended and any regulations thereunder;

"Latin America Credit Bid Amount" has the meaning ascribed to that term in the Recitals;

"Law" means any law, statute, order, decree, consent decree, judgment, rule regulation, ordinance or other pronouncement having the effect of law whether in Canada or any other country, or any domestic or foreign state, county, province, city or other political subdivision or of any Governmental Entity;

"Meeting Order" means the Order of the Court dated Decemer 19, 2019 in connection with the CCAA Proceedings;

"Monitor" has the meaning ascribed to that term in the Recitals;

"Monitor's Certificate" has the meaning ascribed to that term in section 10.6 hereof;

"Monitor's Website" means http://cfcanada.fticonsulting.com/energold/;

"New Cros-Man" means the Canadian corpraction incorporated pursuant to section 8.2 hereof;

"New Energoid" means the Canadian corporation incorporated pursuant to section 8.2 hereof;

"Notes" means the convertible secured notes issued pursuant to the Note Purchase Agreement, which are each, for the avoidance of doubt, severable and constitute its own Claim;

"Noteholder" means a holder of Notes, including any beneficial or entitlement holder of Notes holding such Notes in a securities account with a depository participant or other securities intermediary;

"Note Purchase Agreement" means the note purchase agreement dated as of June 15, 2017 among the Noteholders, Energold, as issuer and certain other entities of the Energold Group, as guarantors;

"Officer" means anyone who is or was, or may be deemed to be or have been, whether by statute, operation of law or otherwise, an officer or de facto officer of any of the Petitioners;

"Omniterra" has the meaning ascribed to that term in the Recitals;

"Order" means any order of the Court in the CCAA Proceedings;

"Outside Date" means February 28, 2020 (or such other date as the Agent, Petitioners and the Monitor may agree);

"Person" is to be broadly interpreted and includes any individual, firm, corporation, limited or unlimited liability company, general or limited partnership, association, trust, unincorporated organization, joint venture, Governmental Entity or any agency, officer or instrumentality thereof or any other entity, wherever situate or domiciled, and whether or not having legal status;

"Petitioners" has the meaning ascribed to that term in the Recitals;

"Plan" means this Amended Plan of Compromise and Arrangement and any amendments, restatements, modifications or supplements hereto made from time to time in accordance with the terms hereof or made at the direction of the Court in the Sanction Order or otherwise;

"Post-Filing Claim" means any claims against any of the Petitioners that arose from the provision of authorized goods and services provided or otherwise incurred on or after the Filing Date in the ordinary course of business;

"Priority Claim" means a Crown Priority Claim or an Employee Priority Claim;

"Priority Claim Reserve" means a Cash reserve, equal to the amount of the Priority Claims, to be deposited by the Petitioners into the Priority Claim Reserve Account for the purpose of paying the Priority Claims;

"Priority Claim Reserve Account" means a segregated interest-bearing trust account established by the Monitor to hold the Priority Claim Reserve;

"Purchase Agreements" has the meaning ascribed to that term in the Recitals;

"Purchased Assets" has the meaning ascribed to that term in the Recitals;

"Purchased Shares" has the meaning ascribed to that term in the Recitals;

"Purchased Shares Credit Bid Amount" has the meaning ascribed to that term in the Recitals;

"Released Claims" means the matters that are subject to release and discharge pursuant to section 9.1;

"Required Majority" means a majority in number of Affected Noteholders representing at least two thirds in dollar value of the Affected Noteholder Claims of Affected Noteholders who actually vote (in person or by Proxy) at the Creditors' Meeting;

"Residual Assets" has the meaning ascribed to that term in the Recitals;

"Resdiual Assets Credit Bid Amount" has the meaning ascribed to that term in the Recitals;

"Sales Process Order" has the meaning ascribed to that term in the Recitals;

"Sanction Order" has the meaning ascribed to that term in section 10.2;

"Section 5.1(2) Director/Officer Claims" means any Director/Officer Claims that may not be compromised pursuant to section 5.1(2) of the CCAA;

"Secured Priority Claims" means the secured Claims against the Energold Group, or any of them, that have priority over the Claim of the Noteholders and/or the Agent;

"Secured Rescue Note" means the secured note from Energold de Mexico, S.A. de C.V. to Energold;

"Share Purchase Agreements" has the meaning ascribed to that term in the Recitals;

"SSP" has the meaning ascribed to that term in the Recitals;

"Tax" or "Taxes" means any and all taxes, duties, fees, premiums, assessments, imposts, levies and other charges of any kind whatsoever, including all interest, penalties, fines, additions to tax or other additional amounts in respect thereof, and including those levied on, or measured by, or referred to as, income, gross receipts, profits, capital, transfer, land transfer, sates, goods and services, harmonized sales, use, value-added, excise, stamp, withholding, business, franchising, property, development, occupancy, employer health, payroll, employment, health, social services, education and social security taxes, all surtaxes, all customs duties and import and export taxes, countervail and anti-dumping, all licence, franchise and registration fees and all employment insurance, health insurance and Canada, Quebec and other government pension plan premiums or contributions;

"Tax Claim" means any Claim by a Taxing Authority against the Petitioners regarding any Taxes in respect of any taxation year or period;

"Taxing Authority" means any of Her Majesty the Queen in right of Canada, Her Majesty the Queen in right of any province or territory of Canada, any municipality of Canada, the Canada Revenue Agency, the Canada Border Services Agency, any similar revenue or taxing authority of Canada and each and every province or territory of Canada and any political subdivision thereof and any Canadian or foreign government, regulatory authority, government department, agency, commission, bureau, minister, court, tribunal or body or regulation making entity exercising taxing authority or power;

"Third Party Released Parties" has the meaning ascribed to that term in section 9.1(a);

"Total Affected Noteholder Claims" has the meaning ascribed to that term in the Recitals;

"Transferred Subsidiaries" means Energold's direct or indirect wholly-owned subsidiaries that are being vested in the Agent in accordance with the Vesting Orders;

"Transferred Subsidiaries Guarantee" means the guarantees given by the Transferred Subsidiaries pursuant to the Note Purchase Agreement;

"Unaffected Affected Noteholder Claims" means the Claims of the Noteholders for the balance owing to them by the Petitioners and the Transferred Subsidiaires under the Note Purchase Agreement and

the Transferred Subsidiaries Guarantee. after applying a credit of the Credit Bid Amount, less the Estimated Priority Payment Amount.

"Unissued US LP Units" has the meaning ascribed to that term in Section 7.2(c);

"US LP" means the limited partnership organized under the laws of the State of Delaware within the United State of America which partnership units shall be held by the Affected Noteholders;

"US LP Agreement" means the US LP Agreement made between and among the partners, as the case may be, of US LP on the Implementation Date, substantially in the form attached hereto as Schedule "F";

"US LP Unitholder Information" means such information and documentation as the Agent and/or US LP may require from recipients of the US LP Units in order to comply with any anti-money laundering, know your client, proceeds of crime and other Laws applicable to the Agent and US LP, respectively, which shall be communicated to the Affected Noteholders by the Agent and/or the US LP.

"US LP Units" means the limited partnership units of US LP, each of which shall have the rights and restrictions attached as set out in the US LP Agreement;

"Vesting Orders" has the meaning ascribed to that term in section 10.2; and

"Withholding Obligation" means the amounts that any payor shall be entitled to deduct and withhold and remit from any distribution, payment or consideration otherwise payable to any Person pursuant to the Plan.

SCHEDULE E

FORM OF MONITOR'S PLAN IMPLEMENTATION DATE CERTIFICATE

No. S1910194 Vancouver Registry

IN THE SUPREME COURT OF BRITISH COLUMBIA

٠

PETITIONERS

(the "Petitioners")

MONITOR'S CERTIFICATE (Plan Implementation)

All capitalized terms not otherwise defined herein shall have the meanings ascribed thereto in the Order of the Honourable \blacklozenge made in these proceedings on \blacklozenge (the "Sanction Order").

Pursuant to paragraph \blacklozenge of the Sanction Order, FTI Consulting Canada Inc. in its capacity as Court-appointed Monitor of the Petitioners (the "Monitor") delivers to the Petitioners this certificate and hereby certifies that it has been informed in writing by the Petitioners and the Agent that all of the conditions precedent set out in the Plan have been satisfied or waived, and that the Implementation Date has occurred and the Plan and the provisions of the Sanction Order which come into effect on the Implementation Date are effective in accordance with their respective terms. This Certificate will be filed with the Court and posted on the website maintained by the Monitor.

DATED at the City of Vancouver, in the Province of British Columbia, this \blacklozenge day of \blacklozenge at 10:00 a.m.

FTI CONSULTING CANADA INC., in its capacity as Court-appointed Monitor of the Petitioners and not in its personal or corporate capacity

By:

Name: Title:

SCHEDULE F US LP AGREEMENT

1.1

Draft 12/18/19 Confidential

,

LIMITED PARTNERSHIP AGREEMENT

OF

ENERGOLD DRILLING ACQUISITION, L.P. a Delaware Limited Partnership

Dated [•], 2020

Error! Unknown document property name.

1. .

TABLE OF CONTENTS

		Page
Article I. THE PART	NERSHIP; DEFINITIONS	2
Section 1.1	Formation of the Partnership	
Section 1.2	Name of the Partnership	
Section 1.3	Purpose of the Partnership	
Section 1.4	Principal Place of Business; Registered Office and Agent	
Section 1.5	Certificate of Limited Partnership	
Section 1.6	Term of Partnership	
Section 1.7	Certain Definitions	
Section 1.8	Interpretation	
	RS; PARTNERSHIP INTERESTS AND units	
Section 2.1	General Partner	
Section 2.2	Limited Partners	
Section 2.2 Section 2.3	Removal and Withdrawal of General Partner	
Section 2.3	Effect of Removal or Withdrawal General Partner	
Section 2.4		
Section 2.5 Section 2.6	Appointment of New General Partner Interests and Units	
		10
	CONTRIBUTIONS; CAPITAL ACCOUNTS; PREEMPTIVE	
RIGHTS.		
Section 3.1	Initial Capital Contributions	10
Section 3.2	Additional Capital Contributions	
Section 3.3	Failure to Make Contributions	
Section 3.4	Partner Loans	
Section 3.5	Withdrawal and Return of Capital to Partners	
Section 3.6	Title to Partnership Property	
Section 3.7	Maintenance of Capital Accounts	
Section 3.8	No Obligation to Restore Negative Balances in Capital Accounts	
Section 3.9	Liability of Partners	
Section 3.10	Preemptive Rights	12
Article IV. ALLOCA	TIONS	
Section 4.1	Allocation of Net Income and Net Loss	
Section 4.2	Allocations to Assigned Interests	
Article V DISTRIBI	JTIONS	
Section 5.1	Distributions	
Section 5.2	Tax Distributions	
Section 5.2 Section 5.3	Restrictions on Distributions	
	EMENT	
Section 6.1	Control and Management.	
Section 6.2	Restrictions on Authority of General Partner	16
Section 6.3	Compensation of Partners and Partner Affiliates; Expenses of	1~
Section 6.4	General Partner	
Section 0.4	Meetings and Votes Without a Meeting	1/

e de la construcción de la const

. . . 1

Section 6.5	Other Activities	
Section 6.6	Indemnification of Partners	18
Article VII. ACCOU	JNTING; REPORTING; TAX MATTERS	
Section 7.1	Partnership Accounting Practices and Tax Elections	
Section 7.2	Access to Records by Limited Partners	
Section 7.3	Income Tax Information	
Section 7.4	Partnership Tax or Information Returns	
Section 7.5	Banking	20
Section 7.6	Financial Reports	20
Section 7.7	Fiscal Year	
Section 7.8	Election of Adjusted Basis	20
Section 7.9	Tax Matters Partner	21
Article VIII. TRAN	SFERS	. 21
Section 8.1	Prohibited Transfers; Transfers in General	
Section 8.2	Right of First Refusal	
Section 8.3	Drag-Along Rights	
Section 8.4	Tag-Along Rights	
Section 8.5	Limited Partners Prohibition on Assignments Violating Securities	
	Laws	
Section 8.6	Transfers in Violation of Restrictions Void	
Section 8.7	Substituted Partners	
Article IX. DISSOL	UTION AND LIQUIDATION	28
Section 9.1	Dissolution and Winding Up	
Section 9.2	Responsibility for Winding Up	
Section 9.3	Liquidation and Distribution	
Section 9.4	Requirements On Liquidation	
Section 9.5	No Recourse Against Partners for Return of Capital	
Section 9.6	Cancellation of Certificate of Limited Partnership	
Article X. REPRES	ENTATIONS AND WARRANTIES	
Section 10.1	Representations, Warranties, and Acknowledgments by Partners	
Article XI MISCEI	LANEOUS	
Section 11.1		
	Complete Agreement; Amendments and Modifications; Partial	
Section 11.2	Invalidity; Waivers	21
Section 11.3	Third-Party Beneficiary; Successors and Assigns	
Section 11.4	Further Assurances	
Section 11.5	Interpretation and Construction	
Section 11.6		
Section 11.7	Waiver of Jury Trial	
Section 11.8	Equitable Remedies	
Section 11.9	Ownership of Partnership Interest; Right of Partition	
	0 Confidentiality	
	-	

and a second second

LIMITED PARTNERSHIP AGREEMENT OF ENERGOLD DRILLING ACQUISITION, L.P.

This Limited Partnership Agreement (as amended, restated, amended and restated, modified or otherwise supplemented from time to time, this "Agreement") of Energold Drilling Acquisition, L.P., a Delaware limited partnership (the "Partnership"), dated as of $[\bullet]$, 2020 (the "Effective Date") is made and entered into by and among the Partnership, Energold Drilling Acquisition G.P., LLC, a Delaware limited liability company, as the initial General Partner (as defined below), and the Limited Partners (as defined below).

RECITALS

WHEREAS, the Partnership was formed on $[\bullet]$, 2019 pursuant to the filing of a Certificate of Limited Partnership (the "*Certificate*") with the Secretary of State of the State of Delaware in accordance with Delaware Revised Uniform Limited Partnership Act, Del. Code Ann. tit. 6, ch. 17 (as amended from time to time, the "*Act*").

WHEREAS, immediately prior to the Effective Date, each of the Initial Limited Partners (as defined below) was a holder of senior secured convertible notes issued by Energold Drilling Corp. ("*Energold*") pursuant to a convertible note agreement, dated June 15, 2017 among Energold, the noteholders party thereto and Extract Advisors LLC, as agent for the noteholders (as amended, restated, amended and restated, modified or otherwise supplemented from time to time, the "*Convertible Note Agreement*").

WHEREAS, on or about September 13, 2019, Energold and its wholly-owned Canadian subsidiaries (collectively, the "*Energold Group*") filed for protection in Canada under the Companies' Creditors Arrangement Act ("*CCAA*") pursuant to *In the Matter of Energold Drilling Corp. and others, SCBC, Vancouver Registry, No. S1910194* (the "*Energold Action*").

WHEREAS, pursuant to the plan of arrangement (the "*Plan of Arrangement*") approved by the Supreme Court of British Columbia (the "*Court*") on $[\bullet]$, 2020, in full satisfaction of the obligations of the Energold Group under the Convertible Note Agreement, (i) the Court vested ownership of all or substantially all of the assets of Energold (including the equity interests in each of its wholly-owned subsidiaries) to the Partnership and (ii) each of the Initial Limited Partners was deemed to have contributed its respective interests under the Convertible Note Agreement to the Partnership in exchange for the interests in the Partnership, all as described in further detail under the Plan of Arrangement.

WHEREAS, the Partnership, the General Partner and the Initial Limited Partners desire to enter into this Agreement in order to (i) reflect the admission of the Initial Limited Partners as limited partners of the Partnership and (ii) set forth the rights and obligations of the Partners (as defined below), as well as the terms of the operation and management of the Partnership, as set forth herein.

NOW, THEREFORE, in consideration of the foregoing and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

AGREEMENT

ARTICLE I.

THE PARTNERSHIP; DEFINITIONS

Section 1.1 Formation of the Partnership. $[\bullet]$, as an authorized person within the meaning of the Act, has executed, delivered and filed the Certificate in the office of the Secretary of State of Delaware. Upon the filing of the Certificate with the Secretary of State of the State of Delaware, [his/her] powers as an authorized person ceased. The rights, powers, duties, obligations, and liabilities of the Partners shall be determined under the Act and this Agreement. To the extent that the rights, powers, duties, obligations, and liabilities of any Partner are different by reason of any provision of this Agreement than they would be under the Act in the absence of such provision, this Agreement shall, to the extent permitted by the Act, control.

Section 1.2 Name of the Partnership. The name of the Partnership is "Energold Drilling Acquisition, L.P." or such other name as may from time to time be designated by the General Partner. The business of the Partnership shall be conducted under that name or such other name as the General Partner may from time to time deem appropriate.

Section 1.3 **Purpose of the Partnership**. The Partnership shall engage solely in the business of, directly or indirectly, through one or more entities: (a) acquiring, owning, operating, managing, holding for investment, exchanging, selling, re-investing, and disposing of the assets (including any Securities of any member of the Energold Group) acquired by the Partnership pursuant to the Plan of Arrangement; and (b) such other activities as are related to or incidental to the foregoing.

Section 1.4 Principal Place of Business; Registered Office and Agent.

(a) The initial principal place of business of the Partnership shall be at c/o Downtown Capital Partners, 360 Hamilton Ave, Suite 1110, White Plains, NY 10601. The Partnership may locate its principal place of business at any other place or places, within or outside of the State of Delaware as shall be designated, from time to time, by the General Partner. The Partnership may have such other offices, within or outside of, the State of Delaware as shall be designated, from time to time, by the General Partner.

(b) The registered office and statutory agent for service of process for the Partnership in the State of Delaware shall be: [•]. The General Partner (as hereinafter defined) may change the registered office or agent for service of process for the Partnership from time to time as permitted under the Act.

Section 1.5 Certificate of Limited Partnership. The General Partner shall execute and cause to be filed certificates of amendment to the Certificate whenever required by the Act or this Agreement, together with any other documents required for qualification of the Partnership to do business where required. Section 1.6 **Term of Partnership**. The term of the Partnership commenced on the date of the filing of the Certificate with the Secretary of State of Delaware and shall continue in perpetuity until the Partnership is dissolved in accordance with this Agreement or by law.

Section 1.7 Certain Definitions. Capitalized terms used herein shall have the following meanings:

"Acceptance Period" shall have the meaning set forth in Section 8.2(c).

"Accepting Limited Partner" shall have the meaning set forth in Section 3.10(b).

"Act" shall have the meaning set forth in the Recitals.

"Additional Capital Contributions" shall have the meaning set forth in Section 3.2(a).

"Affiliate" of any particular Person means any other Person Controlling, Controlled by or under common Control with such particular Person. Notwithstanding the foregoing, (i) neither the Partnership nor any Person Controlled by the Partnership shall be deemed to be an Affiliate of any Partnership or of any Affiliate of any Partner and (ii) no Partner or any Affiliate thereof shall be deemed to be an Affiliate of the Partnership, any Person Controlled by the Partnership or any other Partner solely by virtue of its interest as a Partner.

"Agreement" shall have the meaning set forth in the preamble.

"Available Cash" means cash on hand at the Partnership, after deduction for Reserves.

"BBA" shall have the meaning set forth in Section 7.9.

"Business Day" means any day other than Saturday, Sunday or another other day in which banks in the state of New York or Canada are authorize or required by law to remain closed.

"Capital Account" shall have the meaning set forth in Section 3.7.

"*Capital Contribution*" means, with respect to any Partner, any cash, cash equivalents or the value of property which such Partner contributes to the Partnership pursuant to <u>Section 3.1</u> or <u>Section 3.2</u>.

"CCAA" shall have the meaning set forth in the Recitals.

"Certificate" shall have the meaning set forth in the Recitals.

"Class A Unit" means a limited partnership interest in the Partnership, the holders of which shall have such powers, preferences, rights, qualifications, limitations and restrictions as provided in the Act and as specified in this Agreement with respect to Class A Units.

"*Code*" means the Internal Revenue Code of 1986, as amended from time to time (or any corresponding provisions of succeeding law).

"Confidential Information" means confidential or proprietary information, data or knowhow (regardless of whether disclosed orally or in writing, and whether or not identified as such), whether of a technical, financial, commercial or other nature, of the Partnership or any of its Subsidiaries, which any party receives or has received in connection with the ownership, management and operation of the Partnership and its Subsidiaries.

"Contribution Loan" shall have the meaning set forth in Section 3.3(b).

"Contributing Limited Partner" shall have the meaning set forth in Section 3.3(a).

"Convertible Note Agreement" shall have the meaning set forth in the Recitals.

"Court" shall have the meaning set forth in the Recitals.

"Drag-Along Partners" shall have the meaning set forth in Section 8.3(a).

"Drag-Along Sellers" shall have the meaning set forth in Section 8.3(a).

"Drag-Along Transaction" shall have the meaning set forth in Section 8.3(a).

"Drag-Along Transaction Notice" shall have the meaning set forth in Section 8.3(a).

"Dragging Limited Partners" shall have the meaning set forth in Section 8.3(a).

"*Effective Date*" shall have the meaning set forth in the preamble.

"Energold" shall have the meaning set forth in the Recitals.

"Energold Action" shall have the meaning set forth in the Recitals.

"Energold Group" shall have the meaning set forth in the Recitals.

"Excluded Issuance" means (i) any issuance of New Securities in connection with any pro rata (based on the Partners' respective then-current Percentage Interests) split, dividend or other division, exchange or combination of Units, (ii) any issuance of New Securities as direct consideration in connection with an acquisition or business combination (whether through a merger, recapitalization, joint venture or otherwise) by or involving the Partnership, (iii) any issuance of New Securities upon conversion or exercise of warrants, options and convertible securities issued in compliance with the terms of this Agreement, (iv) any issuance of New Securities to banks, lessors, financial institutions and licensors in connection with the incurrence of indebtedness of the Partnership in accordance with this Agreement, and (v) any issuance of New Securities pursuant to any bona fide equity-based management incentive plan.

"First Acceptance Period" shall have the meaning set forth in Section 8.2(b).

"First Preemptive Offer Period" shall have the meaning set forth in Section 3.10(b).

"First Preemptive Rights Notice" shall have the meaning set forth in Section 3.10(a).

"GAAP" means United States generally accepted accounting principles, consistently applied.

"General Partner" shall have the meaning set forth in Section 2.1.

see and

"Governmental Authority" means any international, national, federal, state, provincial or local governmental, regulatory or administrative authority, agency, commission, court, tribunal, arbitral body or self-regulated entity, whether domestic or foreign.

"Indemnified Party" means (i) the General Partner, (ii) any Limited Partner (or any Affiliate thereof), (iii) any officer, director, manager, shareholder, member, partner, member, employee, representative, advisor or agent of the Partnership, any Partner or any Affiliate thereof, in each case, in their capacity as such, or (iv) any Person who was serving at the request of General Partner, the Partnership, any of its Subsidiaries or any of their respective predecessors as a member, director, manager, officer, partner, venturer, proprietor, trustee, employee, agent, or similar functionary of another foreign or domestic Person or other enterprise; *provided*, that, any Person who was acting as an officer, director or manager of the Partnership, any of its Subsidiaries or any of their respective Date shall not be an "Indemnified Party."

"Individual Matters" shall have the meaning set forth in Section 8.3(d)(ii).

"Interest" means the entire interest of a Limited Partner as a limited partner in the Partnership, all as provided in this Agreement, including the Limited Partner's (i) Capital Account, (ii) Units and the rights attendant to such Units, (iii) interest in the Partnership's profits and losses, (iv) interest in gains, losses, deductions and credits for tax purposes, (v) interest in distributions, and (vi) right to participate as a Limited Partner.

"*Limited Partner*" means each Initial Limited Partner and any other Person admitted to the Partnership as a limited partner in accordance with this Agreement but only so long as such Person owns Units.

"Major Decisions" shall have the meaning set forth in Section 6.2.

"New Securities" means any Units representing Interests in the Partnership (other than non-convertible debt securities), including any "phantom" equity, equity appreciation rights or similar rights, contractual or otherwise and any rights, options or warrants to purchase any Units representing limited partnership interests in the Partnership of any type whatsoever, including any such rights that may become convertible into or exchangeable or exercisable for any of the foregoing.

"Non-Contributing Limited Partner" shall have the meaning set forth in Section 3.3(a).

"Offer Price" shall have the meaning set forth in Section 8.2(a).

"Other Indemnitors" shall have the meaning set forth in Section 6.6(g).

"Ownership Ledger" shall have the meaning set forth in Section 2.2.

"Participation Offer" shall have the meaning set forth in Section 8.4(a).

"Partner Acceptance Notice" shall have the meaning set forth in Section 8.2(c).

"Partners" means, collectively, the General Partner and the Limited Partners.

"Partnership" shall have the meaning set forth in the preamble.

"Partnership Acceptance Notice" shall have the meaning set forth in Section 8.2(b).

"Percentage Interests" means, as of any date of determination, with respect to any Limited Partner or the Units, the percentage determined by dividing (i) the number of Units owned by such Limited Partner as of such date by (ii) the aggregate number of Units owned by all of the Limited Partners as of such date.

"Permitted Transferee" means (a) with respect to any Limited Partner, any Affiliate of such Limited Partner, and (b) with respect to any Limited Partner who is a natural person, (i) the Relatives or Affiliates of such Limited Partner, (ii) a trust under which the distribution of Interests may be made only to such Limited Partner and/or Relatives of such Limited Partner, (iii) a charitable remainder trust, the income from which will be paid to such Limited Partner during his or her life, (iv) any trust, corporation, partnership or limited liability company controlled by such Limited Partner and/or Relatives of such Limited Partner established and maintained solely for charitable purposes or in connection with bona fide estate planning purposes.

"*Person*" means an individual, partnership, corporation, limited liability company, association, joint stock company, trust, joint venture, unincorporated organization or Governmental Authority or any department, agency or political subdivision thereof, including a court.

"Plan of Arrangement" shall have the meaning set forth in the Recitals.

"Preemptive Acceptance Notice" shall have the meaning set forth in Section 3.10(b).

"Preemptive Offer" shall have the meaning set forth in Section 3.10(a).

"Preemptive Rights Sale" shall have the meaning set forth in Section 3.10(a).

"Proportionate Share" shall have the meaning set forth in Section 8.2(c).

"*Relatives*" means, with respect to any Limited Partner who is a natural person, (i) such individual's spouse or domestic partner, (ii) any lineal descendant, parent, grandparent or great grandparent (in each case whether natural or by legal adoption), and (iii) the spouse of an individual described in clause (ii) of this definition

"Removal Date" shall have the meaning set forth in Section 2.4.

"Reserves" means unpaid costs and expenses of the Partnership incurred as of the date of determination plus reserves for future operating and liquidity needs of the Partnership as determined by the General Partner, including expenditures, repayment of debt, contingent liabilities, working capital needs, capital expenditures and other capital requirements and contingencies. The determination of Reserves by the General Partner as of any date of determination will be final and conclusive on the Partners.

"ROFR Notice" shall have the meaning set forth in Section 8.2(a).

"ROFR Units" shall have the meaning set forth in Section 8.2(a).

"Second Acceptance Period" shall have the meaning set forth in Section 8.2(c).

"Second Preemptive Offer" shall have the meaning set forth in Section 3.10(c).

"Second Preemptive Rights Notice" shall have the meaning set forth in Section 3.10(c).

"Securities" means the Units and any other equity securities (or rights, options, warrants or other securities convertible into or exercisable or exchangeable for Units or other Equity Securities) of the Partnership or any of its Subsidiaries.

"Securities Act" shall have the meaning set forth in Section 10.1(b).

"Selling Limited Partner" shall have the meaning set forth in Section 8.4(a).

"Subsidiary" means, with respect to any Person, any corporation, limited liability company, partnership, association or other business entity Controlled by such Person directly or indirectly through any other Subsidiary of such Person or in which such Person owns directly or indirectly through any other Subsidiary of such Person more than ten percent (10%) of the outstanding common stock or other outstanding equity securities ordinarily entitled to vote for the board of directors or board of managers in such Person. Unless otherwise indicated, the term "Subsidiary" refers to a Subsidiary of the Partnership.

"Substituted Partner" shall have the meaning set forth in Section 8.7.

"Tag-Along Partner" shall have the meaning set forth in Section 8.4(a).

"*Tag Sale*" shall have the meaning set forth in <u>Section 8.4(a)</u>.

"Tax Matters Partner" shall have the meaning set forth in Section 7.9.

"*Third Party Purchaser*" means any Person who, immediately prior to the contemplated transaction: (a) does not directly or indirectly own or have the right to acquire any outstanding Units and (b) would not constitute a Permitted Transferee of any Person who directly or indirectly owns or has the right to acquire any Units.

"Transfer" shall have the meaning set forth in <u>Section 8.1</u>.

"Transferring Limited Partner" shall have the meaning set forth in Section 8.2(a).

"Units" means the Class A Units and any other class or series of ownership interests of the Partnership issued from time to time in accordance with this Agreement, in each case owned by the Limited Partners and representing the respective proportionate interests attendant to the Interests in the Partnership.

Section 1.8 **Interpretation.** The definitions in this Agreement shall apply equally to both the singular and plural forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms and words in the singular form shall be construed to include the plural and vice versa. All references herein to Articles, Sections, Annexes, Exhibits and Schedules shall be deemed to be references to Articles and Sections of, and Annexes, Exhibits and Schedules to, this Agreement, unless the context shall otherwise require. All Annexes, Exhibits and Schedules attached hereto shall be deemed incorporated herein as if set forth in full herein and, unless otherwise defined therein, all terms used in any Annex, Exhibit or Schedule shall have the meaning ascribed to such term in this Agreement. The words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation." All accounting terms not defined in this Agreement shall have the meanings determined by GAAP. The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement. Unless otherwise expressly provided herein, any agreement, instrument or statute defined or referred to herein or in any agreement or instrument that is referred to herein means such agreement, instrument or statute as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes) by succession of comparable successor statutes and references to all attachments thereto and instruments incorporated therein. The use of the words "or," "either" and "any" shall not be exclusive. Whenever the term "day" is used in this Agreement, it shall refer to a calendar day unless "Business Days" are specified. Unless otherwise expressly provided herein, whenever the General Partner is granted the right of approval, consent, acceptance, satisfaction, election, determination, discretion, waiver or other action or decision in this Agreement, the General Partner shall made such decision in its sole and absolute discretion, unless otherwise stated to the contrary in this Agreement. This Agreement shall be construed without regard to any presumption or rule requiring construction or interpretation against the party drafting an instrument or cause any instrument to be drafted.

ARTICLE II.

PARTNERS; PARTNERSHIP INTERESTS AND UNITS

Section 2.1 General Partner. Any Person designated from time to time by the Limited Partners holding more than 50% of the issued and outstanding Class A Units with the responsibility of managing the Partnership shall be the "General Partner." Energoid Drilling Acquisition G.P., LLC is hereby admitted to the Partnership as of the Effective Date and appointed as the initial General Partner by the Limited Partners.

Section 2.2 Limited Partners. Each Person listed on the Ownership Ledger as a Limited Partner on the Effective Date is hereby admitted as a limited partner of the Partnership (each, an "*Initial Limited Partner*"). The name and address of each Limited Partner, and the number of each class of Units and Percentage Interest of each Limited Partner, as of the

Effective Date, are set forth on the Partnership's Ownership Ledger (the "Ownership Ledger") attached hereto as <u>Exhibit A</u>. The Ownership Ledger may be amended from time to time by the General Partner to reflect the admission of new Limited Partners, withdrawals or resignations of Limited Partners, additional issuances of Units to Limited Partners, and the issuance and Transfer of Class A Units, each as permitted by the terms and conditions of this Agreement.

Section 2.3 Removal and Withdrawal of General Partner.

(a) The General Partner may be removed at any time, with or without cause, upon the vote of the Limited Partners holding at least 66.66% of the issued and outstanding Class A Units.

(b) The General Partner may withdraw as General Partner by giving written notice of its withdrawal to the Partnership and Limited Partners; *provided*, that such withdrawal shall not be effective until a replacement General Partner has been appointed in accordance with <u>Section</u> <u>2.5</u>.

Section 2.4 Effect of Removal or Withdrawal General Partner. Effective on the effective date of removal or withdrawal of the General Partner (the "*Removal Date*"):

(a) Such General Partner shall no longer transact any business on behalf of the Partnership and shall no longer be the General Partner of the Partnership;

(b) All contracts, agreements or arrangements between the Partnership and such General Partner and/or any of its Affiliates shall immediately terminate as of the Removal Date, without penalty or termination fees;

(c) Such General Partner shall take such action as may be necessary, or as the replacement General Partner may direct, for the transfer, protection, and/or preservation of partnership assets that is in the possession or control of the General Partner and in which the Partnership have or may acquire an interest;

(d) The General Partner shall cooperate fully with the replacement General Partner and the Partnership to effect an orderly transition of management responsibilities to the new general partner; and

(e) The General Partner shall be released of all duties, obligations, rights, and powers as general partner of the Partnership first arising or occurring after the Removal Date; *provided*, *however*, that the General Partner shall not be released from any duties or obligations arising before the Removal Date.

Section 2.5 Appointment of New General Partner. No removal or withdrawal of the General Partner shall be effective unless and until a replacement General Partner has been appointed by the Limited Partners holding at least 50% of the issued and outstanding Class A Units. Such substitute General Partner shall be elected or admitted to the Partnership and the business of the Partnership shall be carried on by such new General Partner in accordance with this Agreement.

Section 2.6 Interests and Units. The Interests in the Partnership held by the Limited Partners shall be represented by Units. The Partnership is initially authorized to issue one class of Units designated as the "Class A Units". The Class A Units shall be subject to the terms and conditions set forth in this Agreement. The Class A Units shall have one vote per Class A Unit, in person or proxy, on all matters upon which the Limited Partners have the right to vote as set forth in this Agreement or the Act. Unless and until the General Partner shall determine otherwise, the Class A Units shall be uncertificated and recorded on the Ownership Ledger in the books and records of the Partnership. If at any time the General Partner shall determine to certificate the Class A Units, such certificates will contain such legends as the General Partner determines are necessary or advisable. The Partnership shall be permitted to create such other classes of Units as determined by the General Partner from time to time in accordance with the provisions of this Agreement.

ARTICLE III.

CAPITAL CONTRIBUTIONS; CAPITAL ACCOUNTS; PREEMPTIVE RIGHTS

Section 3.1 Initial Capital Contributions. As of the Effective Date, each of the Initial Limited Partners has made or is deemed to have made the aggregate initial Capital Contributions in exchange for the number of Class A Units as set forth in the Ownership Register set forth on Exhibit A attached hereto.¹

Section 3.2 Additional Capital Contributions.

(a) In addition to the initial Capital Contributions, each Limited Partner shall make such additional Capital Contributions to the Partnership as the General Partner (with the approval of Limited Partners holding more than 50% of the issued and outstanding Class A Units) may require from time to time (the "Additional Capital Contributions").

(b) The General Partner shall deliver written notice of any Additional Capital Contribution to the Limited Partners specifying the aggregate amount of such Additional Capital Contribution, such Limited Partner's share thereof in accordance with the terms of this Agreement (which shall be made pro rata by each Limited Partner in accordance with their Percentage Interests), and the purpose therefor. Each Limited Partner shall fund its proportionate share of any Additional Capital Contribution within five (5) Business Days after written notice thereof is given.

Section 3.3 Failure to Make Contributions.

(a) If any Limited Partner fails to make any Additional Capital Contribution under <u>Section 3.2</u> as and when due, then such Limited Partner shall be referred to as a "*Non-Contributing Limited Partner*" and each of the Limited Partners contributing their full proportionate share of such Additional Capital Contribution shall be referred to herein as a "*Contributing Limited Partner*."

¹ <u>NTD</u>: Exhibit A will set forth the principal amount of the convertible notes that each Limited Partner will be deemed to have contributed to the Partnership as set forth in the Plan of Arrangement.

(b) If any Limited Partner fails to make any Additional Capital Contribution under <u>Section 3.2</u>, as and when due, then the General Partner shall have the right, but not the obligation, on two (2) Business Days' prior written notice to the Non-Contributing Limited Partner and each Contributing Limited Partner to lend to the Partnership an amount up to the amount of such Non-Contributing Limited Partner's proportionate share (based on such Limited Partner's Percentage Interest) of such Additional Capital Contribution (a "*Contribution Loan*").

(c) Each Contribution Loan shall bear interest at an annual rate equal to the lesser of: (i) eighteen percent (18%) per annum (compounded annually); and (ii) the maximum rate of interest permitted by applicable law, and shall be repaid through distributions by the Partnership pursuant to <u>Section 5.1</u>. In addition, the General Partner may request repayment of any Contribution Loan and all accrued and unpaid interest thereon at any time.

(d)Each Contributing Limited Partner for such Additional Capital Contribution shall have the right to participate in any Contribution Loan related to such Additional Capital Contribution in an amount equal to such Contributing Limited Partner's proportionate share (based on the aggregate Percentage Interests of all Contributing Limited Partners who wish to make the loan contemplated by this Section 3.3) of the Contribution Loan funded by the General Each Contributing Limited Partner shall exercise its right to participate in a Partner. Contribution Loan by delivering written notice to the General Partner and funding its proportionate share within five (5) Business Days after receipt of notice that the General Partner is making such Contribution Loan. Each Contributing Limited Partner making participating in a Contribution Loan shall be deemed to have appointed the General Partner as its agent with the power and authority to make all decisions of such Contributing Limited Partner in respect of such Contribution Loan. All Contribution Loans made in response to the same request for an Additional Capital Contribution shall be deemed, for purposes of Section 5.1(a)(i), to have the same priority.

(e) In addition to the General Partner's and Contributing Limited Partners' right to make a Contribution Loan, the Partnership shall issue additional Class A Units to each Contributing Limited Partner such that the Percentage Interest of the Non-Contributing Limited Partner shall be diluted by 200% of the dilution that otherwise would have resulted from the failure to fund such Additional Capital Contribution.

(f) Without limiting the generality of the foregoing, the sole remedy available to the Partners and the Partnership if one or more Limited Partners fails to make an Additional Capital Contribution under <u>Section 3.2</u> shall be the ability to make a Contribution Loan to the Partnership and dilution under <u>Section 3.3(e)</u>. Except as otherwise provided in this Agreement, no Limited Partner shall have any liability for any failure to make all or any portion of any Additional Capital Contribution required to be made by such Limited Partner under <u>Section 3.2</u>.

Section 3.4 **Partner Loans.** In addition to Contribution Loans, the Partners shall have the right to make loans to the Partnership from time to time on such terms approved by the General Partner and the Limited Partners holding more than 50% of the issued and outstanding Class A Units. Each Limited Partner shall have the right to participate in any such loan in an amount equal to such Limited Partner's proportionate share of the principal amount of such loan (based on the aggregate Percentage Interests of all Limited Partners who wish to make the loan contemplated by this <u>Section 3.4</u>).

NUT SUCCESSION

Section 3.5 Withdrawal and Return of Capital to Partners. Except as otherwise provided in this Agreement: (a) no Partner may withdraw any portion of the capital of the Partnership; (b) no Partner shall be entitled to the return of its Capital Contribution; (c) under circumstances requiring a return of any Capital Contributions, no Partner shall have the right to receive property other than cash; and (d) no interest shall be paid on any Capital Contribution to the Partnership.

Section 3.6 **Title to Partnership Property**. All Partnership assets shall be owned by the Partnership as an entity and, insofar as permitted by applicable law, no Partner shall have any ownership interest in the Partnership assets in its individual name or right, and each Partner's ownership interest in the Partnership (including its Units) shall be personal property for all purposes.

Section 3.7 Maintenance of Capital Accounts.² [The Partnership shall establish and maintain for each Partner a separate capital account (a "*Capital Account*") reflecting the Capital Contributions made by such Partner, the profits and losses allocated to such Partner and the distributions made to such Partner.]

Section 3.8 No Obligation to Restore Negative Balances in Capital Accounts.³ [No Limited Partner shall have an obligation, at any time during the term of the Partnership or on its liquidation, to pay to the Partnership or any other Partner or third party an amount equal to the negative balance in such Limited Partner's Capital Account.]

Section 3.9 Liability of Partners. Except as otherwise expressly provided in the Act, the debts, obligations, and liabilities of the Partnership, whether arising in contract, tort, or otherwise, shall be solely the debts, obligations, and liabilities of the Partnership, and no Limited Partner shall be obligated personally for any such debt, obligation, or liability in its capacity as, and solely by reason of being, a Limited Partner. Except as otherwise expressly provided in the Act, the liability of each Limited Partner in its capacity as a Limited Partner shall be limited to the amount of Capital Contributions required to be made or deemed to have been made by such Limited Partner in accordance with the provisions of this Agreement, but only when and to the extent the same shall become due under the provisions of this Agreement. The provisions of this Section shall not be deemed to modify, waive, or limit the personal obligations and liabilities (if any) of the Limited Partners to each other as specifically provided in this Agreement (including, without limitation, in Section 3.1 and 3.2).

Section 3.10 Preemptive Rights.

(a) If, at any time after the Effective Date, the Partnership proposes to issue any New Securities other than Excluded Issuances (a "*Preemptive Rights Sale*"), subject to <u>Section</u> <u>3.10(e)</u>, the Partnership or such Subsidiary shall first offer in writing at least fifteen (15)

² <u>NTD</u>: Tax provisions subject to review.

³ <u>NTD</u>: Tax provisions subject to review.

Business Days prior to the proposed issuance date of such New Securities (the "*First Preemptive Rights Notice*") to sell to each Limited Partner (a "*Preemptive Offer*") a pro rata share of such offered New Securities equal to its Percentage Interest. The First Preemptive Rights Notice shall include the price(s) and the material terms (in reasonable detail) upon which the Partnership proposes to issue the New Securities.

(b) Each Limited Partner shall have ten (10) Business Days from the date of receipt of a First Preemptive Rights Notice (the "First Preemptive Offer Period") to irrevocably agree to purchase up to that number of New Securities equal to such Limited Partner's then-current Percentage Interest at the price and on the terms set forth in the First Preemptive Rights Notice by giving written notice to the Partnership and the General Partner and stating therein the quantity of New Securities that such Limited Partner desires to purchase. In order to timely accept a Preemptive Offer, in whole or in part, such Limited Partner shall be required to deliver written notice of acceptance (a "Preemptive Acceptance Notice") to the Partnership and the General Partner prior to the end of the aforesaid ten (10) Business Day First Preemptive Offer Period, setting forth the number of New Securities such Limited Partner elects to purchase. Each Limited Partner who timely subscribes for its entire then-current Percentage Interest of such New Securities offered in a First Preemptive Rights Notice shall be an "Accepting Limited Partner". If any Limited Partner fails to deliver a Preemptive Acceptance Notice within the Preemptive Rights Offer Period, then such Limited Partner will be deemed to have irrevocably waived its rights under this Section 3.10 with respect to the purchase of such New Securities in such Preemptive Offer, but such forfeiture shall not affect its rights with respect to any future issuances or sales of New Securities.

If all of the New Securities proposed to be issued or sold pursuant to a Preemptive (c) Right Sale have not been subscribed for within ten (10) Business Days following the Partnership's delivery of the First Preemptive Rights Notice to the Limited Partners, the Partnership shall offer to sell to each Accepting Limited Partner the remaining New Securities at the same price and on the same terms upon which the Partnership proposed to issue the same as set forth in the First Preemptive Rights Notice, which shall be specified by the Partnership in a second notice (the "Second Preemptive Rights Notice") delivered to each Accepting Limited Partner (the "Second Preemptive Offer"). Each Accepting Limited Partner shall have three (3)Business Days following the date of receipt of any Second Preemptive Rights Notice to irrevocably agree to purchase all or any portion of such remaining New Securities at the price and upon the terms specified in the Second Preemptive Rights Notice by giving written notice to the Partnership and the General Partner and stating therein the quantity of New Securities that such Accepting Limited Partner desires to purchase. In order for an Accepting Limited Partner to timely accept a Second Preemptive Offer, in whole or in part, such Accepting Limited Partner shall be required to deliver a Preemptive Acceptance Notice to the Partnership and the General Partner prior to the end of the aforesaid [three (3)] Business Day Second Preemptive Offer period, setting forth the New Securities such Accepting Limited Partner elects to purchase. If, as a result thereof, the Accepting Limited Partners elect to purchase more than the total number of remaining New Securities available for purchase, the number of remaining New Securities to be purchased by each Accepting Limited Partner shall be proportionately reduced based on such Accepting Limited Partner's Percentage Interest, or as otherwise agreed amongst such Accepting Limited Partners.

(d) Any New Securities subject to a Preemptive Rights Sale that are not purchased by Limited Partners pursuant to the terms of this <u>Section 3.10</u> may be issued and sold by the Partnership to the offerees thereof (at a per-Unit purchase price no less than the purchase price offered in the First Preemptive Rights Notice and on terms no more favorable to the purchasers thereof than the terms set forth in the First Preemptive Rights Notice) during the 180-day period immediately following the end of the Second Preemptive Offer Period (which 180-day period may be extended by the General Partner as necessary to allow for the expiration or termination of all waiting periods under antitrust laws applicable to such issuance and to obtain any applicable approvals or consents from Governmental Authorities.) Any New Securities not issued within such 180-day period (as extended, if applicable) will be subject to the provisions of this <u>Section 3.10</u> upon subsequent issuance.

(e) Notwithstanding anything to the contrary in this Section 3.10, the Partnership may elect to exempt any issuance of New Securities from the provisions of this Section 3.10, if the proposed Third Party Purchaser would be unwilling to purchase such New Securities with the dilution that would be caused by the exercise of preemptive rights by the Limited Partners under this Section 3.10. In addition, if the General Partner determines that the issuance of such New Securities is necessary or advisable to cure any urgent liquidity needs of the Partnership, the Partnership may close the issuance of New Securities without complying with the provisions of this Section 3.10 and then offer such New Securities to the Limited Partners by delivering a First Preemptive Rights Notice within thirty (30) days after the closing of such issuance and otherwise in compliance with this Section 3.10. In the event that the Partnership determines to deliver a First Preemptive Rights Notice after completion of the issuance of such New Securities, the Limited Partners desiring to exercise their preemptive rights under this Section 3.10 shall have the right to purchase their allocated share of the New Securities in any of the following manners determined by the General Partner: (A) in a secondary sale from the Person or Persons who initially acquired such New Securities, (B) from the Partnership and have the net proceeds of such sale used to repurchase New Securities from the Person or Persons who acquired New Securities prior to the delivery of the First Preemptive Rights Notice (which repurchase shall be made at cost), (C) in an incremental sale to the sale of such New Securities, or (D) use any one or more of clauses (A) through (C), provided, that the Partnership must structure such sale to the participating Limited Partners in such a manner that each participating Limited Partner has the opportunity to purchase up to the same aggregate number or principal amount of New Securities on the same terms as it could have purchased had (x) the First Preemptive Rights Notice been delivered prior to completion of such issuance of New Securities and (y) the same aggregate number or principal amount of New Securities been issued as the aggregate number or principal amount of New Securities that are issued prior to and after the delivery of the actual First Preemptive Rights Notice (net of any New Securities repurchased pursuant to clause (B) above).

ARTICLE IV.

ALLOCATIONS

Section 4.1 Allocation of Net Income and Net Loss.⁴ [For each fiscal year (or portion thereof), all items of income, profits, losses, credits, and deductions of the Partnership shall be allocated among the Partners in proportion to their Percentage Interests.]

Section 4.2 Allocations to Assigned Interests.⁵ [In the event of a transfer of Interests during any fiscal year made in compliance with the provisions of <u>Article VIII</u>, items of income, gain, loss, and deduction of the Partnership attributable to such Interests for such fiscal year shall be determined using the interim closing of the books method.]

ARTICLE V.

DISTRIBUTIONS

Section 5.1 **Distributions**.

(a) The Partnership shall make distributions of Available Cash to the Partners from time to time as determined by the General Partner, in the following order of priority:

(i) <u>First</u>, to pay interest and then principal of all outstanding loans (if any) made to the Partnership by the Partners (it being understood and agreed that: (A) all such loans shall be repaid in the order of priority in which they were made; and (B) all such loans having the same repayment priority shall be repaid to the Partners making such loan on a pro rata basis);

(ii) <u>Second</u>, to the Limited Partners, in the same proportion to each Limited Partner as such Limited Partner's unreturned Capital Contributions bear to the aggregate total unreturned Capital Contributions for all Limited Partners, until the amount of each Limited Partner's unreturned Capital Contributions has been reduced to zero; and

(iii) <u>Third</u>, to the Limited Partners, pro rata, in accordance with their respective Percentage Interests.

(b) All amounts withheld or required to be withheld under the Code or any provision of any state, local or foreign tax law regarding any payment, distribution, or allocation to the Partnership or any Partner and treated by the Code (whether or not withheld under the Code) or any such tax law as amounts payable by or about any Partner or any person owning an interest, directly or indirectly, in such Partner shall be treated as amounts distributed to the Partner regarding which such amount was withheld under this <u>Section 5.1(b)</u> and <u>Section 9.3</u> for all purposes under this Agreement. The General Partner is authorized to withhold from distributions, or regarding allocations, to the Partners and to pay over to any federal, state, local or foreign government any amounts required to be so withheld under the Code or any provisions

⁴ <u>NTD</u>: Tax provisions subject to review.

⁵ <u>NTD</u>: Tax provisions subject to review.

of any other federal, state, local, or foreign law and shall allocate any such amounts to the Partner regarding which such amount was withheld.⁶

Section 5.2 **Tax Distributions.** To the extent that distributions under <u>Section 5.1(a)</u> for the current tax year have been insufficient, and if the Partnership has adequate Available Cash, the Partnership shall, before April 15, June 15, September 15, and January 15 of each year (or such other dates by which installment tax contributions may be due), make a distribution in cash in the amount necessary to allow the Partners to pay the estimated or actual taxes due on each such date attributable to the income of the Partnership to be allocated to the Partners.⁷

Section 5.3 **Restrictions on Distributions**. The Partners shall look solely to the assets of the Partnership for any distributions, whether liquidating distributions or otherwise. If the assets of the Partnership remaining after the payment or discharge, or the provision for payment or discharge, of the debts, obligations, and other liabilities of the Partnership are insufficient to make any distributions, no Partner shall have any recourse against the separate assets of any other Partner.

ARTICLE VI.

MANAGEMENT

Section 6.1 **Control and Management.** The business and affairs of the Partnership shall be managed by the General Partner in accordance with this <u>Article VI</u>. Except as otherwise provided in this Agreement, including, without limitation, as provided in <u>Section 6.2</u>, the General Partner shall have sole and exclusive control over the Partnership, and the General Partner shall have the power and authority to take such action from time to time as the General Partner may deem to be necessary, appropriate, or convenient in connection with the management and conduct of the business and affairs of the Partnership.

Section 6.2 **Restrictions on Authority of General Partner**. Notwithstanding anything in <u>Section 6.1</u> to the contrary, in addition to any other rights granted to the Limited Partners in this Agreement, the General Partner shall not have authority to do or take any of the following actions (the "*Major Decisions*") without the affirmative vote of the Limited Partners holding more than 50% of the issued and outstanding Class A Units:

(a) The sale or exchange of any of the assets of the Partnership or any of its Subsidiaries, or the acquisition of any assets from outside the Partnership or any of its Subsidiaries, (in a single transaction or a series of related transactions), having a value over one million dollars (\$1,000,000) in US Dollars; or

(b) The consent or acquiescence to the filing of any involuntary petition for liquidation or reorganization of the Partnership (but specifically excluding any of its Subsidiaries with revenues in the twelve months prior to the date of such filing that are not material to the

⁶ <u>NTD</u>: Tax provisions subject to review.

⁷ <u>NTD</u>: Tax provisions subject to review.

Partnership (as determined by the General Partner)) under any federal or state law for the relief of debtors.

Section 6.3 Compensation of Partners and Partner Affiliates; Expenses of General Partner.

(a) Except for any fees payable to the General Partner that are (i) approved by the Limited Partners holding at least 66.66% of the issued and outstanding Class A Units, and (ii) on terms no more favorable to the General Partner than are typically paid to managing members or general partners of similar partnerships, no Partner or affiliate thereof shall be entitled to any fees or compensation from the Partnership. For purposes of this Section 6.3, an annual fee equal to or less than 1.25% of the estimated fair market value of all assets of the Partnership as of the date of determination shall be deemed to satisfy the requirement of Section 6.3(a)(ii).

(b) The General Partner shall be reimbursed by the Partnership for all reasonable and documented expenses incurred or made on behalf of the Partnership and its Subsidiaries.

(c) The General Partner shall have the right to engage any subcontractors to perform the obligations of the General Partner under this Agreement, and the reasonable cost of such subcontractors shall be an expense of the Partnership, subject to the condition that the General Partner waive any fees or compensation to which it may be entitled under Section 6.3(a).

Section 6.4 Meetings and Votes Without a Meeting. The General Partner may at any time call for a meeting of the Partners or for a vote without a meeting by written consent. Additionally, the General Partner shall call for a meeting or for a vote without a meeting by written consent (which written consent shall require no notice) following receipt of written request therefor from any Limited Partner holding more than 10% of the issued and outstanding Class A Units. Within five (5) days after the date on which the General Partner decides (or is obligated) to call a meeting of the Partners, the General Partner shall notify all Limited Partners of record of the time and place of the Partnership meeting, if called, and the general nature of the business to be transacted. Any action required or permitted to be taken by the Partners may be taken without a meeting if the Partners representing the minimum number of votes that would be necessary to authorize or take such action at such meeting have consented in writing or by electronic transmission (including via email) to the adoption of a resolution authorizing the action, with such written consent having the same force and effect as a vote of such Partners; provided, that the General Partner will deliver a copy of any action by written consent to all of the Limited Partners.

Section 6.5 **Other Activities.** Any Partner may engage in or have an interest in other business ventures of any nature or description, independently or with others, whether presently existing or hereafter created, including those in competition with the operations of the Partnership, and neither the Partnership nor any other Partner shall have any rights in or to such independent ventures or the income or profits derived therefrom resulting from being a Partner in the Partnership; *provided*, that the General Partner shall have the right to redact, withhold or otherwise limit access to information about the Partnership and its Subsidiaries and their respective activities to any Partner who is engaged in any such competitive business ventures.

Error! Unknown document property name.

Section 6.6 Indemnification of Partners.

(a) To the fullest extent permitted by applicable law, the Partnership agrees to indemnify each Indemnified Party, and to defend, protect, save, and hold each Indemnified Party harmless from all claims, fees, costs, losses, damages, judgments, fines, settlements and expenses (including reasonable attorneys' fees) (collectively, "Losses") incurred in connection with or resulting from any claims, actions, suits, proceedings or demands, whether civil, criminal, administrative or investigative, arising out of or in any way relating to the Partnership or any of its assets or properties, including amounts paid in settlement or compromise (if recommended by the Partnership's counsel) of any such claim, action, or demand and all fees, costs, and expenses (including reasonable attorneys' fees) in connection therewith. Notwithstanding the foregoing, indemnification shall not be available or paid to any Indemnified Party regarding any matter as to which there has been a final and non-appealable judgment entered by a court of competent jurisdiction determining that, in respect of the matter in question, the Indemnified Party's acts or omissions constituted bad faith, fraud or willful misconduct or, in the case of a criminal matter, the Indemnified Party acted with knowledge that his, her or its conduct was criminal.

(b) To the fullest extent permitted by applicable law, expenses (including legal fees) incurred by an Indemnified Party in defending any claim, action, suit, proceeding or demand shall, from time to time, be advanced by the Partnership prior to the disposition of such claim, action, suit, proceeding or demand upon receipt by the Partnership of an undertaking by or on behalf of the Indemnified Party to repay such amount if it shall be determined that the Indemnified Party is not entitled to be indemnified as authorized in Section 6.6(a).

(c) The indemnification provided under this <u>Section 6.6</u> shall be in addition to any other rights which an Indemnified Party may be entitled under any other agreement or vote of the Partners, as a matter of law or equity, or otherwise, and shall continue as to an Indemnified Party who is a Partner but who has ceased to serve in that capacity, and shall inure to the benefit of the heirs, successors, assigns, and administrators of the Indemnified Parties.

(d) The Partnership may maintain insurance, at its expense, to protect any Indemnified Party against any Losses described in this <u>Section 6.6</u> whether or not the Partnership would have the power to indemnify such Indemnified Party against such Losses under the provisions of this <u>Section 6.6</u>.

(e) Notwithstanding anything contained herein to the contrary (including in this <u>Section 6.6</u>) any indemnity by the Partnership relating to the matters covered in this <u>Section 6.6</u> shall be provided out of and to the extent of Partnership assets only, and no Partner (unless such Partner otherwise agrees in writing or is found in a final decision of a court of competent jurisdiction to have personal liability on account thereof) shall have personal liability on account thereof or be required to make additional Capital Contributions to help satisfy such indemnity of the Partnership.

(f) If this <u>Section 6.6</u> or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the Partnership shall nevertheless indemnify and hold harmless each Indemnified Person pursuant to this <u>Section 6.6</u> to the fullest extent permitted by

any applicable portion of this <u>Section 6.6</u> that shall not have been invalidated and to the fullest extent permitted by applicable law. The rights granted pursuant to this <u>Section 6.6</u> shall be deemed contract rights, and no amendment, modification or repeal of this <u>Section 6.6</u> shall have the effect of limiting or denying any such rights with respect to actions taken or omitted or any claim, action, suit, proceeding or demand arising prior to any amendment, modification or repeal.

The Partnership hereby acknowledges that certain of the Indemnified Parties may (g) have certain rights to indemnification, advancement of expenses or insurance (collectively, the "Other Indemnitors"). The Partnership agrees that: (i) to the extent legally permitted, (A) the Partnership is the indemnitor of first resort (i.e., its obligations to each Indemnified Party are primary and any obligation of the Other Indemnitors to advance expenses or to provide indemnification for the same expenses or liabilities incurred by any such Indemnified Party are secondary), and (B) the Partnership shall be required to advance the full amount of expenses incurred by an Indemnified Party and shall be liable for the full amount of all Losses, without regard to any rights that an Indemnified Party may have against the Other Indemnitors; and (ii) the Partnership irrevocably waives, relinquishes and releases the Investor Indemnitors from any and all claims for contribution, subrogation or any other recovery of any kind in respect of any of the matters described in clause (i) of this sentence for which any Indemnified Party has received indemnification or advancement from the Partnership. The Partnership further agrees that no advancement or payment by the Other Indemnitors on behalf of any Indemnified Party with respect to any claim for which an Indemnified Party has sought indemnification from the Partnership shall affect the foregoing and that the Other Indemnitors shall have a right of contribution and/or be subrogated to the extent of such advancement or payment to all of the rights of recovery of such Indemnified Party against the Partnership. The Partnership agrees that the Other Indemnitors are express third party beneficiaries of the terms of this Section 6.6(g).

ARTICLE VII.

ACCOUNTING; REPORTING; TAX MATTERS

Section 7.1 **Partnership Accounting Practices and Tax Elections**. The General Partner shall maintain or cause to be maintained at all times true and correct books, records, reports, and accounts in which shall be entered fully and accurately all transactions of the Partnership. The books and records of the Partnership shall be kept by the General Partner in accordance with GAAP, unless otherwise required under the Code. The accountants for the Partnership shall be selected by the General Partner. The General Partner may cause the Partnership to make such tax elections from time to time which it determines to be appropriate.

Section 7.2 Access to Records by Limited Partners. Each Limited Partner and its duly authorized representatives, attorneys, or attorney in fact of any of them, shall have the right, on reasonable request, to inspect and copy, during normal business hours, any and all Partnership records and documents relating to its interest in the Partnership.

Section 7.3 Income Tax Information. The General Partner shall cause the Partnership to provide each Partner (a) a Form K-1 prepared by the Partnership's accountant as and when required under applicable laws and (b) at such Partner's cost, on reasonable request, such other information as shall be reasonably required by such Partner to enable it to file any of

its tax returns or as such Partner may reasonably request for the purpose of enabling it to comply with any reporting, filing, or other requirements imposed by any statute, rule, regulation, or otherwise by any Governmental Authority or with its own internal rules, regulations, and policies generally applicable regarding investments of this nature.

Section 7.4 **Partnership Tax or Information Returns**. The General Partner shall cause the preparation and timely filing at the Partnership's expense, of all Partnership tax returns, and timely make all other filings required by any Governmental Entity having jurisdiction to require such filing, the cost of which, in each case, shall be borne by the Partnership. The General Partner shall, upon request, cause the Partnership to furnish copies of such returns to each Limited Partner.

Section 7.5 **Banking**. The General Partner shall open and thereafter maintain one or more separate bank accounts in the name of the Partnership with a federally insured reputable bank or other financial institution in which there shall be deposited funds of the Partnership. No other funds shall be deposited in said accounts. The funds in said accounts shall be used solely for the business of the Partnership, and all withdrawals therefrom are to be made only by the General Partner or such other persons as the General Partner may from time to time designate.

Section 7.6 **Financial Reports.** The General Partner shall be responsible, at the expense of the Partnership, for the preparation of financial reports of the Partnership and the coordination of financial matters of the Partnership with the Partnership's accountants. The Partnership shall make commercially reasonable efforts to deliver to each Limited Partner the following financial information:

(a) within 180 days after the end of each fiscal year of the Partnership, the consolidated unaudited balance sheet of the Partnership and its Subsidiaries as of the end of such fiscal year, together with related statements of operations, cash flow and partners' equity for such fiscal year prepared in accordance with GAAP; and

(b) within 60 days after the end of each fiscal quarter of the Partnership, the consolidated unaudited balance sheet of the Partnership and its Subsidiaries as of the end of such fiscal quarter, together with related statements of operations, cash flow and partners' equity for such fiscal quarter, prepared in accordance with GAAP (except that such financial statements may (x) be subject to normal year-end adjustments and (y) not contain all notes thereto that may be required in accordance with GAAP).

Section 7.7 **Fiscal Year**. The fiscal year of the Partnership shall be the calendar year, unless otherwise determined by the General Partner, or as otherwise required under the Code and the Treasury Regulations promulgated thereunder. As used in this Agreement, a fiscal year shall include any partial fiscal year at the beginning and end of the Partnership term.

Section 7.8 **Election of Adjusted Basis.** [In the event of a transfer of all or part of the Interest of a Limited Partner, the General Partner shall, on request by a Limited Partner, elect, on behalf of the Partnership, to adjust the basis of the assets of the Partnership under Section 754 of

the Code, if no material adverse income tax consequence shall result to the Partnership or any Partner.]⁸

Section 7.9 Tax Matters Partner. The General Partner shall be designated as the initial "partnership representative" within the meaning of Section 6223 of the Code or under any similar state or local law (the "Tax Matters Partner"). The General Partner shall have the authority to designate another person of its choosing as a replacement Tax Matters Partner. In the event the Partnership shall be the subject of an income tax audit by any federal, state or local authority, to the extent the Partnership is treated as an entity for purposes of such audit, including administrative settlement and judicial review, the Tax Matters Partner shall be authorized to act for, and its decision shall be final and binding upon, the Partnership and each Partner, and the Tax Matters Partner shall be indemnified and held harmless by the Partnership and each Partner for any action so taken by it in good faith. All expenses incurred in connection with any audit, investigation, settlement, or review by the IRS or any other Governmental Authority shall be borne by the Partnership. The General Partner is hereby authorized and shall take all actions necessary to qualify the Partnership as a partnership for federal income tax purposes in accordance with Regulation Section 301.7701-3 (and/or comparable provisions of state and local law). The Limited Partners agree that any and all actions taken by the Tax Matters Partner shall be binding on the Partnership and all of the Limited Partners and the Limited Partners shall reasonably cooperate with the Partnership and the General Partner, and undertake any action reasonably requested by the Partnership or the General Partner, in connection with any elections made by the Tax Matters Partner.⁹

ARTICLE VIII.

TRANSFERS

Section 8.1 **Prohibited Transfers; Transfers in General.**

(a) No Limited Partner shall sell, hypothecate, pledge, encumber, assign, or otherwise transfer (a "*Transfer*"), with or without consideration, all or any portion of its Interest in the Partnership to any other Person without the prior written consent of the General Partner (such consent not to be unreasonably withheld, conditioned or delayed), except for Transfers (i) to Permitted Transferees (which may be made upon prior written notice to the Partnership), (ii) in connection with a Drag-Along Transaction pursuant to <u>Section 8.3</u> or (iii) made by a Tag-Along Partner upon the exercise of its rights pursuant to <u>Section 8.4</u>. In no event shall any Transfer (other than pursuant to a Drag-Along Transaction) be to any competitor or potential competitor of the Partnership or any of its Subsidiaries (or any Affiliate of any such competitor) without the prior written consent of the General Partner (which may be withheld, conditioned or delayed for any reason).

(b) No Transfer of Interests to a person not already a Partner of the Partnership shall be deemed completed until the prospective Transferee is admitted as a Partner of the Partnership in accordance with <u>Section 8.7</u>.

⁸ <u>NTD</u>: Tax provisions subject to review.

⁹ <u>NTD</u>: Tax provisions subject to review.
(c) Except as otherwise provided in this Agreement, any Partner Transferring its Interest shall pay or reimburse all costs and expenses incurred by the Partnership in connection with such Transfer upon demand by the Partnership.

Section 8.2 Right of First Refusal.

(a) If a Limited Partner desires to Transfer all or any portion of its Units to any Person, such Limited Partner (the "*Transferring Limited Partner*") shall deliver a written notice (a "*ROFR Notice*") to the Partnership and each other Partner at least thirty (30) days prior to the proposed consummation of such Transfer; *provided*, that the right of first refusal set forth in this <u>Section 8.2</u> shall not apply to a Transfer of Units that is (i) made to a Permitted Transferee, (ii) proposed to be made by a Dragging Partner or required to be made by a Drag-Along Seller in a Drag-Along Transaction pursuant to <u>Section 8.3</u>, or (iii) made by a Tag-Along Partner upon the exercise of its rights pursuant to <u>Section 8.4</u>. The ROFR Notice shall set forth the terms and conditions of such offer, including the name of the prospective purchaser, the number of such Units to be transferred (the "*ROFR Units*"), the proposed purchase price per Unit (which shall be entirely in cash) of the ROFR Units (the "*Offer Price*"), payment terms, and all other material terms and conditions of the proposed Transfer. The ROFR Notice shall constitute an irrevocable offer by the Transferring Limited Partner to the Partnership and the other Partners to Transfer the ROFR Units at the Offer Price and on the terms and conditions contained in the ROFR Notice.

(b) The Partnership may elect to purchase any or all of the ROFR Units at the Offer Price and on the terms set forth in the ROFR Notice by delivering written notice to the Transferring Limited Partner (the "*Partnership Acceptance Notice*") stating the applicable number of ROFR Units that the Partnership elects to purchase within ten (10) Business Days after receipt of the ROFR Notice (the "*First Acceptance Period*"). The Partnership Acceptance Notice shall be binding and irrevocable upon delivery by the Partnership.

If the Partnership does not elect to purchase all of the ROFR Units, then for a (c) period of ten (10) Business Days commencing on the earlier of (i) receipt of a Partnership Acceptance Notice pursuant to which the Partnership has elected to purchase less than all of the ROFR Units, and (ii) the expiration of the First Acceptance Period (the "Second Acceptance Period" and together with the First Acceptance Period, each an "Acceptance Period"), the other Partners shall have the right to purchase such ROFR Units not so elected to be purchased by the Partnership, at the Offer Price and on the terms set forth in the ROFR Notice by delivering notice in writing to the Transferring Limited Partner (a "Partner Acceptance Notice") stating the applicable number of ROFR Units that such Partner elects to purchase. The specific number of such ROFR Units that each Partner shall be entitled to purchase after the Partnership has exercised, or declined to exercise, its rights under this Section 8.2 shall be determined on a pro rata basis in proportion to the respective number of Units owned beneficially by each such Partner as of the date of the ROFR Notice in relation to the total number of Units owned beneficially by all Partners. Each Partner shall also be entitled to indicate a desire to purchase all or a portion of any ROFR Units remaining after such pro rata allocation. Each Partner shall be allocated the maximum amount of ROFR Units set forth in its Partner Acceptance Notice, unless such allocation would result in the allocation of more securities in the aggregate than are available for purchase by the Partners, in which case such ROFR Units shall be allocated among the Partners pro rata in accordance with each such Partner's proportionate share as determined

above. Each Partner Acceptance Notice shall be binding and irrevocable upon delivery by such Partner.

(d) The failure of the Partnership or a Partner to deliver a Partnership Acceptance Notice or a Partner Acceptance Notice, respectively, by the end of the relevant Acceptance Period shall constitute a waiver of the Partnership's or such Partner's rights of first refusal under this <u>Section 8.2</u> with respect to the Transfer of the ROFR Units set forth in the ROFR Notice, but shall not affect its respective rights with respect to any future Transfers.

If the Partnership and/or the Partners shall have exercised their respective rights (e) to purchase all, and not less than all, of the ROFR Units, the Transfer of such Units shall be consummated as soon as practicable after the delivery of such Partnership Notice, but in any event within fifteen (15) Business Days after the expiration of the latest Acceptance Period (which may be extended by the General Partner to allow for the expiration or termination of all waiting periods under antitrust laws applicable to such Transfer and to obtain any applicable approvals or consents from Governmental Authorities). Each participant in the sale contemplated by this Section 8.2(e) shall take all actions, enter into agreements and deliver certificates and instruments and consents as may be reasonably necessary to consummate such sale. At the closing of any sale and purchase of the ROFR Units pursuant to this Section 8.2(e), the Transferring Limited Partner shall deliver to the Partnership and/or the participating Partners certificates (if any) representing the ROFR Units to be sold, free and clear of any liens or encumbrances (other than those contained in this Agreement), an instrument of Transfer containing customary representations and warranties (including with respect to ownership free and clear of any liens or encumbrances (other than those contained in this Agreement), due execution and authority and non-contravention) as reasonably requested by the General Partner, accompanied by evidence of transfer and all necessary transfer taxes paid and stamps affixed, if necessary, against receipt of the purchase price therefor from the Partnership and/or the participating Partners by wire transfer of immediately available funds.

(f) If the Partnership and the Partners fail to collectively elect to purchase all of the ROFR Units, then such Transferring Limited Partner may, during the sixty (60) day period immediately following the expiration of the latest Acceptance Period and subject to <u>Section 8.1</u> and <u>Section 8.5</u>, sell any unsold ROFR Units to any other Person (other than any competitor of the Partnership or any of its Subsidiaries, or to any Affiliate of any such competitor); *provided*, *however*, that the Transfer to such other Person shall be at the same price and on the same terms as set forth in the ROFR Notice. If such ROFR Units are not sold within such sixty (60) day period, then such Transferring Limited Partner must again comply with all of the terms and conditions of this <u>Section 8.2</u> with respect to any proposed Transfer of Units owned by the Transferring Limited Partner.

Section 8.3 Drag-Along Rights.

(a) If, at any time after the Effective Date, one or more Limited Partners individually or collectively holding 50% or more of the issued and outstanding Class A Units (the "*Dragging Limited Partners*") elect to consummate, or cause the Partnership to consummate, a transaction or series of related transactions involving the Transfer of 50% or more of the then issued and outstanding Class A Units to any Third Party Purchaser, in one or more bona fide arms-length transactions, whether structured as (i) a direct or indirect sale of Units, (ii) a merger of the Partnership or any of its Subsidiaries (other than one in which the Limited Partners immediately prior to such merger or consolidation own a majority, directly or indirectly, of the equity of the survivor or acquirer), or (iii) sale, lease, transfer, license or other disposition of assets of the Partnership and its Subsidiaries (taken as a whole) (each, a "*Drag-Along Transaction*"), the Dragging Limited Partners may give notice (a "*Drag-Along Transaction Notice*") to all of the other Limited Partners (the "*Drag-Along Partners*," and, together with the Dragging Limited Partners, the "*Drag-Along Sellers*") at least thirty (30) days prior to the proposed execution of definitive documentation in connection with the Drag-Along Transaction. (b)The Drag-Along Transaction Notice shall specify (i) the amount and form of consideration (which need not be paid exclusively in cash) to be received by the Drag-Along Sellers or the Partnership and any other consideration to be received directly or indirectly by any Drag-Along Seller or its Affiliates in connection with such Drag-Along Transaction regardless of whether such consideration relates to the Units to be Transferred in connection with such Drag-Along Transaction (which amount and form of consideration need not be paid exclusively in cash and shall be subject to the provisions of Section 8.3(d), be the same form and amount among each class of Units and which other terms and conditions shall be identical for each of the Drag-Along Sellers), (ii) in reasonable detail, any other material terms and conditions of the proposed Drag-Along Transaction, including a copy of the proposed definitive agreements (if available), (iii) the identity of the proposed Third Party Purchaser in the proposed Drag-Along Transaction, (iv) the date of anticipated completion of the proposed Drag-Along Transaction (which date shall be not less than sixty (60) days after the delivery of such Drag-Along Transaction Notice), and (v) the action or actions reasonably requested or required of each Drag-Along Limited Partner (without limiting those that may be reasonably requested or required in the future) in order to complete or facilitate such proposed Drag-Along Transaction (including (A) the Transfer of Units owned by the Drag-Along Limited Partner, (B) the voting by such Drag-Along Limited Partner in favor of the Drag-Along Transaction and the transactions contemplated thereby and the waiver of any related appraisal or dissenters' rights and (C) the execution and delivery of any merger, asset purchase, security purchase, recapitalization or other agreement, as applicable). Upon receipt of such Drag-Along Transaction Notice, each Drag-Along Limited Partner shall be obligated to take the actions referred to in subclause (v) above and any other actions reasonably requested by the Dragging Limited Partners in connection with the Drag-Along Transaction. The Drag-Along Limited Partners shall be permitted to sell their Units pursuant to any Drag-Along Transaction without complying with any other provisions of this Article VIII.

(c) If a Drag-Along Transaction contemplates the Transfer of more than 50% but less than all of the issued and outstanding Class A Units of the Partnership, then each Drag-Along Seller shall Transfer a number of Class A Units equal to the product of the following: (i) the number of Class A Units owned by such Drag-Along Seller multiplied by (ii) a fraction, the numerator of which is the aggregate number of Class A Units proposed to be Transferred in such Drag-Along Transaction and the denominator of which is the aggregate number of all issued and outstanding Class A Units as of immediately prior to entry into definitive documents with respect to the Drag-Along Transaction (including in each such case any Class A Units to be issued upon the conversion, exchange or exercise of any Securities in connection with such Drag-Along Transaction). (d) The obligations of the Limited Partners pursuant to this <u>Section 8.2</u> are subject to the following terms and conditions:

(i) Each Limited Partner will bear (a) its own costs of any Transfer of Units pursuant to a Drag-Along Transaction and (b) its pro rata share (based upon the relative amount of proceeds received in such Drag-Along Transaction) of the costs incurred in connection with such Drag-Along Transaction that are incurred for the benefit of all Limited Partners or the Partnership and are not otherwise paid by the Partnership or the Third Party Purchaser;

(ii) no Limited Partner shall be required to provide any representations, warranties or indemnities in connection with the Drag-Along Transaction, other than (A) customary several (and not joint) indemnities related to the Partnership, its Subsidiaries and their business for which the recourse is allocated pro rata based on the aggregate consideration received by the Limited Partners in the Drag-Along Transaction, and then only to the extent that each other Limited Partner provides similar indemnities and (B) customary (including with respect to qualifications) several (and not joint) representations, warranties and indemnities concerning: (1) such Limited Partner's valid title to and ownership of Units owned by such Limited Partner, free of all liens, claims and encumbrances (excluding those arising under applicable securities laws and this Agreement); (2) such Limited Partner's authority, power and right to enter into and consummate such Drag-Along Transaction; (3) the absence of any material violation, default or acceleration of any agreement to which such Limited Partner is subject or by which its assets are bound as a result of the Drag-Along Transaction; and (4) the absence of, or compliance with, any governmental consents, approvals, filings or notifications required to be obtained or made by such Limited Partner in connection with the Drag-Along Transaction (and then only to the extent that each other Limited Partner provides similar representations, warranties and indemnities with respect to the Units held by such Limited Partner) (clauses (1) through (4), a Limited Partner's "Individual Matters");

any Limited Partner's allocable holdback, escrow and indemnification (iii) obligations shall be determined on the same percentage basis as the percentage of proceeds payable to such Limited Partner relative to the total proceeds paid or payable to all Limited Partners in connection with the Drag-Along Transaction (i.e., no Limited Partner shall have a higher percentage of their total proceeds or potential proceeds held back, escrowed or be subject to indemnity payments than any other Limited Partner for any reason). Any such consideration placed in escrow, held back or subject to indemnity payments shall be allocated among Limited Partners such that, if the Third Party Purchaser is ultimately entitled to some or all of such escrow, holdback amounts or indemnity because of a breach of any particular Limited Partner's Individual Matters, the net ultimate proceeds received by all Limited Partners in connection with the Drag-Along Transaction (after giving effect to any disbursement out of the escrow or retention of any amounts held back on account of a breach of any Individual Matters) is consistent with the amounts they would have received as if the ultimate resolution of such escrow or holdback had been known at the closing of the Drag-Along Transaction and any liabilities relating thereto for Individual Matters had been borne solely by the Limited Partner responsible therefor; and

(iv) a Limited Partner's liability in connection with a Drag-Along Transaction shall in no event exceed the amount of consideration payable to such Limited Partner in connection with such Drag-Along Transaction.

(e) The Dragging Limited Partners shall have the right in connection with a prospective Drag-Along Transaction (or in connection with the investigation or consideration of any Drag-Along Transaction) to require the Partnership to reasonably cooperate with potential acquirers in such prospective Drag-Along Transaction by taking all customary and other actions reasonably requested by such potential acquirers, including making the Partnership's properties, books and records, and other assets reasonably available for inspection by such potential acquirers, establishing a physical or electronic data room including materials customarily made available to potential acquirers in connection with such processes and making its employees reasonably available for presentations, interviews and other diligence activities, in each case subject to each such potential acquirer and its representatives entering into a confidentiality agreement with terms at least as restrictive as those set forth in Section 11.10. In addition, the Dragging Limited Partners shall be entitled to take all steps reasonably necessary to carry out an auction of the Partnership, including selecting an investment bank, providing confidential information (pursuant to confidentiality agreements), selecting the winning bidder and negotiating the requisite documentation. The Partnership shall provide assistance with respect to these actions as reasonably requested. The Dragging Limited Partners shall keep the General Partner fully informed of the status of any Drag-Along Transaction.

(f) If other Securities in addition to Class A Units are to be transferred in such Drag-Along Transaction, notwithstanding anything herein to the contrary, the General Partner shall be permitted to determine (i) the amount of such Securities to transferred in connection with such Drag-Along Transaction, (ii) the form and amount of consideration received by the Drag-Along Sellers in such Drag-Along Transaction in respect of such Securities and (ii) the appropriate manner of treatment of all outstanding Securities in connection with such Drag-Along Transaction in accordance with the terms of such Securities.

(g) Each Limited Partner hereby constitutes and appoints each Dragging Limited Partner, the General Partner and the Partnership, with full power of substitution, as such Limited Partner's true and lawful representative and attorney-in-fact, in such Limited Partner's name, place and stead, to execute and deliver any and all agreements that (i) are required to be executed and delivered under and in accordance with this <u>Section 8.3</u> and (ii) have not been executed and delivered by such Limited Partner within five (5) Business Days following receipt thereof. The Partnership shall provide a copy of such agreements to such Limited Partner within five (5) Business Days after execution; *provided, however*, that failure to deliver such documents within such time period shall not impair or affect the validity of such agreements. The foregoing power of attorney is coupled with an interest and shall continue in full force and effect notwithstanding the subsequent death, incapacity, bankruptcy or dissolution of any Limited Partner.

Section 8.4 Tag-Along Rights.

(a) If, at any time, any Limited Partner (or group of Affiliated Limited Partners) (collectively, the "*Selling Limited Partner*") desire(s) to effect a Transfer (in one transaction or series of related transactions) of its Units representing more than 50% of the then issued and

outstanding Class A Units to any other Third-Party Purchaser (a "*Tag Sale*") and a Drag-Along Transaction Notice has not been delivered, if applicable, then at least thirty (30) days prior to the closing of such Tag Sale, the Selling Limited Partner shall make a written offer (the "*Participation Offer*") to each other Limited Partner (each, a "*Tag-Along Partner*") to include in the proposed Tag Sale a portion of Units held by such Limited Partner equal to the product of (1) the number of Units held by such Tag-Along Partner <u>multiplied</u> by (2) a fraction, the numerator of which is the number of Units proposed to be Transferred by the Selling Limited Partners in such Tag Sale and the denominator of which is the aggregate number of Units then held by the Selling Limited Partners. All Units included in a Tag Sale (including those of the Selling Limited Partner and the Tag-Along Partners electing to participate in such Tag Sale) shall be purchased upon the same terms and conditions and consideration (including form and amount) per Unit.

(b)The Participation Offer shall (i) include copies of all material documents proposed to be executed in connection with the Tag Sale, (ii) describe, in reasonable detail, the identity of the proposed purchaser, the terms and conditions of the proposed Tag Sale (including the number of Units proposed to be sold by the Selling Limited Partner, the form and amount of consideration offered per Unit, the proposed total enterprise value established by the Third Party Purchaser and a representation that the proposed Third Party Purchaser in such Tag Sale has been informed in writing of the rights of the Tag-Along Partners under this Section 8.3). The Tag Sale shall be conditioned upon (x) the consummation of the transactions contemplated in the Participation Offer with the proposed Third Party Purchaser named therein and (y) each selling Limited Partner's execution and delivery of all agreements and other documents as the Limited Partners are required to execute and deliver in connection with such Tag Sale. Each Tag-Along Partner may accept the Participation Offer by delivering written notice (the "Tag Exercise Notice") to the Selling Limited Partner within ten (10) Business Days after the date on which such Tag-Along Partner receives the Participation Offer. Each Tag-Along Partner who does not deliver a Tag Exercise Notice with respect to a Tag Sale within the ten (10) Business Day period set forth in the immediately preceding sentence shall be deemed to have waived all of such Tag-Along Partner's rights to participate in such Tag Sale.

(c) The Selling Limited Partner shall use reasonable efforts to cause the Third Party Purchaser in the Tag Sale to agree to acquire all Units of Tag-Along Partners who have exercised their rights to sell Units in accordance with such Participation Offer. If the Third Party Purchaser in the Tag Sale is unwilling or unable to acquire all such Units, then the Selling Limited Partner shall reduce, to the extent necessary, the number of Units it otherwise would have sold in the proposed Transfer so as to permit those Tag-Along Partners who have accepted the Participation Offer to sell the percentage of their Units that they are entitled to sell under this and the Selling Limited Partner and such other Tag-Along Partners shall Transfer the number of Units specified in the Participation Offer to the Third Party Purchaser in accordance with the terms of such Transfer as set forth in the Participation Offer. In the event the Selling Limited Partner or the Third Party Purchaser shall modify the terms of the proposed Tag Sale in any material way, the Selling Limited Partners shall promptly deliver an amended Participation Offer to each Tag-Along Partner setting forth, in reasonable detail, such modified terms. Each Tag-Along Partner shall, if it so desires to Transfer its Units pursuant to the Participation Offer, as so amended, deliver a new Tag Exercise Notice within five (5) Business Days after the date such amended Participation Offer is received by the Tag-Along Partner (and for the avoidance of doubt, the other provisions of this shall apply with respect to such amended Participation Offer). If the Selling Limited Partners have not consummated the applicable Tag Sale prior to the date that is 180 days (which 180-day period may be extended by the General Partner as necessary to allow for the expiration or termination of all waiting periods under antitrust laws applicable to such issuance and to obtain any applicable approvals or consents from Governmental Authorities) after the date that the Participation Offer (or amended Participation Offer, as applicable) is delivered, the applicable Participation Offer shall be null and void, and it shall be necessary for a new Participation Offer to be furnished, and the terms and provisions of this Section 8.3 separately complied with, in order to consummate any subsequent Tag Sale pursuant to this Section 8.3.

(d) Notwithstanding the foregoing, all of the provisions of <u>Section 8.2(d)</u> with respect to a Drag-Along Transaction shall apply, *mutatis mutandis*, with respect to any Tag Sale.

Section 8.5 Limited Partners Prohibition on Assignments Violating Securities Laws. No transfer or assignment of any Partner's Interest shall be made if counsel for the Partnership shall be of the opinion that such transfer or assignment may be in violation of any federal or state securities laws applicable to the Partnership.

Section 8.6 **Transfers in Violation of Restrictions Void.** Any assignment, sale, exchange, or other transfer in contravention of any of the provisions of this <u>Article VIII</u> shall be void and of no force or effect, and shall not bind or be recognized by the Partnership.

Section 8.7 Substituted Partners. Subject to the provisions of this <u>Article VIII</u>, any transferee of an Interest may become a "Substituted Partner" when all of the following conditions have been satisfied:

(a) A duly executed and acknowledged written instrument of assignment shall have been filed with the Partnership, which instrument shall specify the Interest being assigned and set out the intention of the assignor that the Transferee succeed to the assignor's Interest as a Substituted Partner;

(b) The Transferring Limited Partner and transferee shall have executed such other instrument(s) as the General Partner may deem necessary or desirable to effect such substitution, including the written acceptance; and

(c) The other provisions of this <u>Article VIII</u> shall have been complied with.

ARTICLE IX.

DISSOLUTION AND LIQUIDATION

Section 9.1 Dissolution and Winding Up.

(a) The Partnership shall be dissolved, and its affairs shall be wound up, on the occurrence of any of the following events:

(i) The approval of the General Partner and the Limited Partners holding at least 66.66% of the Class A Units.

(ii) All or substantially all of the Partnership's assets are sold or otherwise disposed of, and the only property of the Partnership consists of cash from such sale or disposition available for distribution to the Partners; *provided* that (i) the legal existence of the Partnership may be continued until the last day of its usual taxable year and (ii) if the Partnership receives deferred payments as consideration, the Partnership shall be continued until such date as determined by the General Partner.

- (iii) A decree of judicial dissolution is entered under the Act.
- (iv) Any other event causing dissolution of the Partnership under the Act.

(b) The Partnership shall not dissolve as the result of the death, retirement, resignation, expulsion, bankruptcy or dissolution of any Limited Partner, or upon the occurrence of any other event which terminates the continued partnership of a Limited Partner under this Agreement or under the Act. In such event, the Limited Partners shall take whatever steps may be required under the Act to continue the business of the Partnership.

Section 9.2 **Responsibility for Winding Up**. On the dissolution of the Partnership, the affairs of the Partnership shall be wound up by the General Partner or, if there is no General Partner remaining, the Partnership's affairs shall be wound up by the Limited Partners. If the Limited Partners wind up the Partnership's affairs, they shall be entitled to reasonable compensation on such terms and conditions no more favorable than permitted elsewhere in this Agreement to the General Partner.

Section 9.3 Liquidation and Distribution. Notwithstanding anything set out in this Agreement to the contrary, the person(s) responsible for winding up the affairs of the Partnership under <u>Section 9.2</u> shall take full account of the Partnership's assets and liabilities, shall liquidate the assets of the Partnership as promptly as is consistent with obtaining the fair market value thereof, and shall apply and distribute the proceeds in the following order of priority (with non-cash items being valued at fair market value, as reasonably determined by the persons responsible for the winding-up):

(a) <u>First</u>, to pay all outstanding debts and liabilities of the Partnership (to the extent that such debts and liabilities are then due), including any loans due to the Partners;

(b) <u>Second</u>, to fund a reasonable reserve for contingent liabilities of the Partnership (after passage of a reasonable time the balance, if any, in said reserve shall be distributed as set out below); and

(c) <u>Third</u>, all remaining proceeds shall be distributed in accordance with <u>Section</u> 5.1(a)(ii) to 5.1(a)(iii). Such distribution required by this <u>Section 9.3(c)</u> shall be made by the end of the fiscal year in which such dissolution occurs, or, if later, within ninety (90) days after the date of such dissolution.

[It is the intention of the Partners that the positive Capital Account balance of each Partner immediately before the receipt of any liquidating distributions by such Partner will be equal to the amount distributable to such Partner under Section 5.1(a)(ii). If a Partner's Capital Account is not equal to such amount, the General Partner shall cause the allocations of profit and loss, or items thereof (including items of gross income and deductions) for the tax year of the Partnership ending with the liquidation to be allocated in such a way so as to cause each Partner's Capital Account to equal the amount that such Partner is entitled to receive under Section 5.1(a)(ii).]¹⁰

Section 9.4 **Requirements On Liquidation**. [Notwithstanding anything set out in this Agreement to the contrary, if the Partnership is "liquidated" (or any Partner's interest in the Partnership is "liquidated") (as that term is defined in Regulation Section 1.704-1(b)(2)(ii)(g)) and any Partner's Capital Account (or, as the case may be, the Capital Account of the Partner whose interest is "liquidated") has a deficit balance, such Partner(s) shall have no obligation to restore such deficit balance or otherwise contribute to the capital of the Partnership and such deficit shall not be considered a debt owed to the Partnership or to any other Person for any purpose whatsoever.]¹¹

Section 9.5 No Recourse Against Partners for Return of Capital. The Limited Partners shall look solely to the assets of the Partnership for the return of their Capital Contributions, which shall be returned, if at all, from distributions, if any, made as provided in this Agreement, and they shall have no recourse against the General Partner or any other Limited Partner.

Section 9.6 **Cancellation of Certificate of Limited Partnership**. On the completion of the winding up of the Partnership's affairs, the Partners conducting the winding up of the Partnership's affairs shall execute and file in the office of the Secretary of State of the State of Delaware a certificate of cancellation.

ARTICLE X.

REPRESENTATIONS AND WARRANTIES

Section 10.1 **Representations, Warranties, and Acknowledgments by Partners**. Each Partner represents, warrants, and acknowledges, as to itself only, that:

(a) It is acquiring its Partnership Interest for its own account and not with a view to or for sale in connection with any distribution of the Units acquired.

(b) It is an "accredited investor" (as defined in Regulation D promulgated under the Securities Act of 1933, as amended (the "*Securities Act*")).

¹⁰ <u>NTD</u>: Tax provisions subject to review.

¹¹ <u>NTD</u>: Tax provisions subject to review.

(c) It: (i) has adequate net worth; (ii) has no need for liquidity in this investment; (iii) can bear the substantial economic risks of an investment in the Partnership for an indefinite period; and (iv) at the present time, can afford a complete loss of such investment.

(d) It acknowledges that the Units have not been registered under the Securities Act, or state securities laws, in reliance on exemptions therefrom for non-public offerings. It understands and agrees that the Units must be held indefinitely unless the sale or other transfer thereof is subsequently registered under the Securities Act and applicable state securities laws or an exemption from such registration is available. It further acknowledges that the Partnership is under no obligation to register the Units on its behalf or to assist it in complying with any exemption from registration.

(e) It acknowledges that it may not be able to sell or dispose of its Units as there will be no public market for them. In addition, it understands that its right to transfer the Interests will be subject to the conditions set out in this Agreement.

(f) Neither it nor any partner, Limited Partner, or stockholder of such Partner is, and no legal or beneficial interest in a partner, Limited Partner, or stockholder of such Partner is, or will be held, directly or indirectly by a person or entity that appears on a list of individuals and/or entities for which transactions are prohibited by the US Treasury Office of Foreign Assets Control or any similar list maintained by any other governmental authority, regarding which entering into transactions with such person or entity would violate the USA Patriot Act or regulations or any Presidential Executive Order or any other similar applicable law, ordinance, order, rule, or regulation.

ARTICLE XI.

MISCELLANEOUS

Section 11.1 Notices. Unless specifically stated otherwise in this Agreement, all notices, waivers, and demands required under this Agreement shall be in writing and delivered to all other parties at the addresses set out in the Preamble, by one of the following methods: (a) hand delivery, whereby delivery is deemed to have occurred at the time of delivery; (b) a nationally recognized overnight courier company, whereby delivery is deemed to have occurred the Business Day following deposit with the courier; (c) Registered United States Mail, signature required and postage-prepaid, whereby delivery is deemed to have occurred on the third Business Day following deposit with the United States Postal Service; or (d) electronic transmission if the transmission is completed no later than four p.m. EST on a Business Day and the original also is sent via overnight courier or United States Mail, whereby delivery is deemed to have occurred at the end of the Business Day on which electronic transmission is completed. Any party shall change its address for purposes of this <u>Section 11.1</u> by giving written notice as provided in this <u>Section 11.1</u>.

Section 11.2 Complete Agreement; Amendments and Modifications; Partial Invalidity; Waivers.

(a) This Agreement may be executed in counterparts, and when executed and delivered by all parties in person, by email pdf, shall become one (1) integrated agreement enforceable on its terms. This Agreement supersedes all prior agreements between or among the parties regarding the subject matter hereof and all discussions, understandings, offers, and negotiations with respect thereto, whether oral or written. All exhibits that are referenced in this Agreement or attached to it are incorporated herein and made a part hereof as if fully set out in the body of the document.

(b) The General Partner may amend or waive any provision of this Agreement with the approval of the Limited Partners holding at least 66.66% of the Percentage Interests; *provided, however*, that (i) no amendment or waiver that disproportionately and adversely affects the powers, rights, preferences or obligations of any Limited Partner relative to any other Limited Partner in existence immediately prior to such amendment or modification may be effected without the prior written consent of such disproportionately and adversely affected Limited Partner; (ii) no amendment that modifies the limited liability of a Limited Partner or a Limited Partner's obligations to make capital contributions may be effected without the prior written consent of such Limited Partner; and (iii) no amendment or waiver of any provision of this <u>Section 11.2(b)</u> shall be permitted without the consent of the percentage of Partners required to effect an amendment or waiver under such provision. The Ownership Ledger may be amended from time to time without the consent of the Limited Partners to reflect the admission of new Partners, withdrawals or resignations of Partners, additional issuances of Units to Partners, and the issuance and Transfer of Units, all in accordance with this Agreement.

(c) Any term or provision of this Agreement which is invalid or unenforceable in any jurisdiction will, as to that jurisdiction, be ineffective to the extent of such invalidity or unenforceability without rendering invalid or unenforceable the remaining terms and provisions of this Agreement or affecting the validity or enforceability of any of the terms or provisions of this Agreement in any other jurisdiction. If any provision of this Agreement is so broad as to be unenforceable, the provision will be interpreted to be only so broad as is enforceable.

(d) Any waiver of any provision or of any breach of this Agreement shall be in writing and signed by the party waiving said provision or breach. No waiver of any breach of any agreement or provision herein contained shall be deemed a waiver of any preceding or succeeding breach thereof or of any other agreement or provision herein contained. No extension of time for performance of any obligations or acts shall be deemed an extension of the time for performance of any other obligations or acts.

Section 11.3 Third-Party Beneficiary; Successors and Assigns.

(a) This Agreement is an agreement solely for the benefit of the Partners (and their permitted successors and/or assigns). Except as set forth in <u>Section 6.6</u>, no other Person shall have any rights hereunder nor shall any other person, party, or entity be entitled to rely on the terms, covenants, and provisions contained herein. The provisions of this <u>Section 11.3</u> shall survive the termination of this Agreement or dissolution of the Partnership.

(b) This Agreement and all its covenants, terms, and provisions, shall be binding on and inure to the benefit of each party and its permitted successors and assigns.

Section 11.4 Further Assurances. Each Partner agrees to do such things, perform such acts, and make, execute, acknowledge, and deliver such documents as may be reasonably necessary and customary to carry out the intent and purposes of this Agreement, if any of the foregoing do not materially increase any Partner's obligations hereunder or materially decrease any Partner's rights hereunder.

Section 11.5 Interpretation and Construction. The parties hereto acknowledge that each party and its counsel have had the opportunity to review this Agreement and that the normal rule of construction to the effect that any ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of this Agreement or any amendments or exhibits hereto. Any captions or headings used in this Agreement are for convenience only and do not define or limit the scope of this Agreement.

Section 11.6 Governing Law; Jurisdiction; Attorneys' Fees.

(a) This Agreement shall be enforced, governed, and construed in all respects in accordance with the internal laws of the State of Delaware, without giving effect to the choice of law or conflict of law rules or laws of such jurisdiction.

(b) EACH PARTY TO THIS AGREEMENT HEREBY SUBMITS TO THE EXCLUSIVE JURISDICTION OF THE DELAWARE COURT OF CHANCERY AND ANY STATE APPELLATE COURT THEREFROM WITHIN THE STATE OF DELAWARE (UNLESS THE DELAWARE COURT OF CHANCERY SHALL DECLINE TO ACCEPT JURISDICTION OVER A PARTICULAR MATTER, IN WHICH CASE, OF ANY DELAWARE STATE OR FEDERAL COURT WITHIN THE STATE OF DELAWARE), AND ANY JUDICIAL PROCEEDING BROUGHT AGAINST ANY PARTY HERETO WITH RESPECT TO ANY DISPUTE ARISING OUT OF THIS AGREEMENT OR ANY MATTER RELATED HERETO SHALL BE BROUGHT ONLY IN SUCH COURTS. Service of process, summons, notice, or other document by registered mail to the address set out in <u>Section 11.1</u> shall be effective service of process for any suit, action, or other proceeding brought in any such court.

Section 11.7 Waiver of Jury Trial. EACH PARTY TO THIS AGREEMENT HEREBY EXPRESSLY AND UNCONDITIONALLY WAIVES, IN CONNECTION WITH ANY SUIT, ACTION, OR PROCEEDING BROUGHT BY THE OTHER PARTY HERETO UNDER THIS AGREEMENT OR IN CONNECTION WITH ANY TRANSACTION CONTEMPLATED HEREBY, ANY AND EVERY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY.

Section 11.8 Equitable Remedies. Each Party hereto acknowledges that a breach or threatened breach by such party of any of its obligations under this Agreement would likely give rise to irreparable harm to the other parties, for which monetary damages would not be an adequate remedy, and hereby agrees that in the event of a breach or a threatened breach by such Party of any such obligations, each of the other Parties hereto shall, in addition to any and all other rights and remedies that may be available to them relating to such breach, be entitled to equitable relief, including a temporary restraining order, an injunction, specific performance, and

any other relief that may be available from a court of competent jurisdiction (without any requirement to post bond).

Section 11.9 **Ownership of Partnership Interest; Right of Partition**. The interest of each Partner in the Partnership shall be personal property for all purposes. No Partner shall have any right to partition the Property or any assets of the Partnership and each Partner hereby irrevocably waives any and all right to partition, or to maintain any action for partition, or to compel any sale regarding its interest in any assets or properties of the Partnership except as expressly provided in this Agreement.

Section 11.10 Confidentiality. No Partner shall, during or after the term of this Agreement, whether through an Affiliate or otherwise, use or disclose Confidential Information to any Person for any reason or purpose whatsoever, except (a) to authorized representatives and employees of such Partner on a confidential basis, (b) disclosure to a Governmental Authority as required to enforce such Partner's rights under this Agreement or any other written agreement between such Partner, on the one hand, and the Partnership or any of its Subsidiaries, on the other hand; (c) as part of such Partner's normal regulatory reporting or review procedures; (d) to any bona fide prospective purchaser of the equity or assets of such Partner or its Affiliates or the Units held by such Partner, or prospective merger partner of such Partner or its Affiliates, in each case to the extent not otherwise limited hereunder; provided that such purchaser or merger partner is advised of the confidential nature of such Confidential Information and agrees to be bound in writing to restrictions on use and disclosure of Confidential Information for a customary period and otherwise consistent with the terms hereof; provided, further, that in no event shall any Partner disclose any Confidential Information to a competitor or potential competitor of the business of the Partnership and its Subsidiaries without the prior written consent of the General Partner (which may be withheld in its sole discretion); and (e) as is required to be disclosed by order of a Governmental Authority, or by subpoena, summons or legal process, or by law, rule or regulation on the advice of counsel; provided that the Partner required to make such disclosure shall provide to the Partnership prompt written notice of any such disclosure to the extent permitted by applicable law to allow the Partnership reasonable time to seek a protective order preventing disclosure of such Confidential Information. Nothing in this Section 11.10 shall in any way limit or otherwise modify or limit any confidentiality covenants entered into by any Partners pursuant to any other agreement to which such Partner, on the one hand, and the Partnership or any of its Subsidiaries, on the other hand, are parties. Notwithstanding the foregoing, "Confidential Information" shall not include information that (i) is already in the possession of such Partner free of any obligation of confidentiality (to such Partner's knowledge) at the time that such information is first disclosed by or received from, through or on behalf of the Partnership or any of its Subsidiaries, (ii) is or becomes known to the general public through no fault or other action of such Partner or its representatives, (iii) is obtained lawfully from a third party who is not known by such Partner to be subject, directly or indirectly, to an obligation to the Partnership or any of its Subsidiaries to keep such information confidential or (iv) was or is independently developed by such Partner without the use of Confidential Information.

[Signature pages follow.]

IN WITNESS WHEREOF, the Parties hereto have executed this Agreement as of the date first written above.

PARTNERSHIP

ENERGOLD DRILLING ACQUISITION, L.P., a Delaware limited partnership

By: Energold Drilling Acquisition, G.P., LLC, its general partner

By:			
Name:			
Title:			

GENERAL PARTNER

ENERGOLD DRILLING ACQUISITION, G.P., LLC, a Delaware limited liability company

D	
Βv	•

Name: Title:

LIMITED PARTNERS

[TO COME]

By:		
Name:		
Title:		

EXHIBIT A

OWNERSHIP LEDGER

Name of Limited Partner	[Initial Capital Contribution]	Class A Units	Percentage Interest
[LIMITED PARTNER NAME]	\$[•]		[•]%
[LIMITED PARTNER NAME]	\$[•]		[•]%
[LIMITED PARTNER NAME]	\$[•]	-	[•]%
Total:	\$[•]		100%

SCHEDULE C

FORM OF MONITOR'S PLAN IMPLEMENTATION DATE CERTIFICATE

No. **S1910194** Vancouver Registry

IN THE SUPREME COURT OF BRITISH COLUMBIA

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND

IN THE MATTER OF THE BUSINESS CORPORATIONS ACT, S.B.C. 2002, C. 57

AND

IN THE MATTER OF ENERGOLD DRILLING CORP., CROS-MAN DIRECT UNDERGROUND LTD., EGD SERVICES LTD., BERTRAM DRILLING CORP. AND OMNITERRA INTERNATIONAL DRILLING INC.

PETITIONERS

MONITOR'S CERTIFICATE

(Plan Implementation)

All capitalized terms not otherwise defined herein shall have the meanings ascribed thereto in the Order of the Honourable \blacklozenge made in these proceedings on \blacklozenge (the "Sanction Order").

Pursuant to paragraph \blacklozenge of the Sanction Order, FTI Consulting Canada Inc. in its capacity as Court-appointed Monitor of the Petitioners (the "Monitor") delivers to the Petitioners this certificate and hereby certifies that it has been informed in writing by the Petitioners and the Agent that all of the conditions precedent set out in the Plan have been satisfied or waived, and that the Implementation Date has occurred and the Plan and the provisions of the Sanction Order which come into effect on the Implementation Date are effective in accordance with their respective terms. This Certificate will be filed with the Court and posted on the website maintained by the Monitor. DATED at the City of Vancouver, in the Province of British Columbia, this \blacklozenge day of \blacklozenge .

FTI CONSULTING CANADA INC., in its capacity as Court-appointed Monitor of the Petitioners and not in its personal or corporate capacity

R	1	٠
υ	y	٠

Name: Title: